STUDIES ON SINGULAR PERTURBATIONS OF OPTIMAL CONTROL SYSTEMS WITH APPLICATIONS TO NUCLEAR REACTOR CONTROL

KOICHI ASATANI

JUNE 1974

INSTITUTE OF ATOMIC ENERGY KYOTO UNIVERSITY

STUDIES ON SINGULAR PERTURBATIONS OF OPTIMAL CONTROL SYSTEMS WITH APPLICATIONS TO NUCLEAR REACTOR CONTROL

KOICHI ASATANI

JUNE 1974

INSTITUTE OF ATOMIC ENERGY KYOTO UNIVERSITY

DOC	
1974	
8	
電気系	

Preface

It is urged that effective techniques be developed to approximate large systems to systems with reduced model order. When more precise results are desired, one encouters the difficulty of increasing dimensionality, throughout the three steps of model making, analysis of the system and synthesis of the system. Moreover, the estimation of the error in the method adopted, is of great importance and in general more difficult than obtaining the approximate solution itself.

Systems can be classified into two classes, i.e., differential systems (dynamical systems) and algebraic systems. For both systems, several methods have been developed to reduce the dimension, in order to avoid the difficulty of increasing dimensionality. The singular perturbation theory, ε -coupling system, block diagonalization, sparse matrix theory, etc. are more or less approximation methods using subsystems which are generated from the original system and are more easily dealt with than the original. The singular perturbation theory is efficient in treating differential systems of a certain class.

The perturbation method in general is very useful in treating rather complex systems which are not very different from known systems. The conventional perturbation method treats the "regular system", where the perturbing term does not change the dimension of the system. The dynamical system involving a singularly perturbing parameter whose presence causes the dimension of the system to increase, cannot be dealt with by the regular perturbations. Such systems of singular perturbation type appear in many fields, e.g., network theory, fluid mechanics, magnethydrodynamics, etc.

In this thesis, devoted to the studies of the singular perturbation theory in connection with two-point boundary value problems arising in the optimal control systems, emphasis is made upon the error estimations of the derived approximate solution in the form of asymptotic expansions. The singular perturbation theory proves to be basic and widely

i

applicable to real physical systems of large size which usually happen to involve stiff differential equations. The manner of the description of this thesis is as follows; firstly, the method of constructing formal asymptotic solutions is presented, and then the basic theorems are offered establishing the validity of the method proposed.

Abstract

This thesis consists of six chapters. The historical survey of the singular perturbation theory is given in the first chapter. The outline and basic idea of the theory are also presented. Further, the well-known formula of Vasil'eva giving the initial conditions of the higher order recursive set of equations and the matching law of the method of matched asymptotic expansions are proved to be identical. The relation is first shown explicitly by the author.

In Chapter 2, a finite dimensional model is treated, which is described by a set of ordinary differential equations. Such models may offer a description of a broad class of realistic physical systems. Three problems occuring typically in optimization problems are considered in three sections. The results in these sections may cover many variations. In Section 2.2, we derive basic results on tracking problems. The results can be extended to the regulator problems, yielding theorems under hypotheses, different from those of Yackel and Kokotović [Y.73]. In Section 2.3, a fixed-end-point, fixed-endtime and minimum energy problem will be treated. The generalized Riccati transformation is introduced to treat the ill-conditioned two-point boundary value problems generated by the fixed-terminal problems. Section 2.4 is concerned with the system involving two singularly perturbing parameters of different orders of smallness. The results of this section can be easily applied to the systems involving more parameters. The last section is devoted to the study of the asymptotic behaviour of the performance index.

Chapter 3 deals with the nuclear reactor control as a lumped parameter system. Such a lumped parameter model is well known in the field of reactor engineering as a "point reactor". The results in the previous chapter are successfully applied to the analysis and synthesis of suboptimal control of the point reactor. This chapter consists of two sections. The early section is concerned with the

iii

error estimation of prompt jump approximation with the aid of the singular perturbation theory and some numerical examples are shown. The latter section treats the regulator problem of the reactor, also with numerical examples.

Chapter 4 is concerned with near-optimum control of systems involving spatially distributed parameters. Such systems are usually described by partial differential equations, integral equations or integro-differential equations. The system considered is represented by partial differential equations of parabolic type.

It is to be noted that the practical treatment of partial differential equations usually cannot be made without involving an approximation, spatial discritization, modal expansion, etc., since the equations can be regarded as of infinite dimensional. Hence one should realize the importance of the reduction of the number of independent variables if possible.

This chapter consists of two parts, the former one is devoted to the study of the singular perturbations of a certain class of a set of evolution equations of parabolic type. The latter deals with nearoptimum control problems on the basis of the results derived in the former one.

Chapter 5 is concerned with the nuclear reactor control as a distributed system. The importance of this treatment has been increasing, since the size of reactors is becoming larger and the spatial effects cannot be neglected. For instance, control of the neutron flux distribution, burnup control with consideration of hot-spotfactor, Xenon oscillation, etc. should be treated by considering distributed parameters.

The method proposed in this chapter is the singular perturbation theory connected with one of the modal expansion, the Helmholtz modal expansion. The Helmholtz mode gives the complete basis in the space L^2 . The theoretical foundation, upon which the validity of the method presented relies, has been given in Chapter 4.

In the last chapter, the results obtained are summarized.

iv



Table of Contents many and many and

「日本である」というのかい

●21. - 14.4 - 金 ²¹ 	:	Page
Preface	•••••	i
Abstract	• • • • • • • • • • • • • •	iii
Table of Contents	•••••	vi
Chapter 1 Singular Perturbation Theory	•••••	1
1.1 Introduction		1
1.2 Singular perturbation theory for ordinary	differential	
equations	· · · · · · · · · · · · · · ·	4
1.3 Singular perturbation theory for abstract	evolution	
equations	•••••	7
1.4 Asymptotic solutions and basic theorems		9
1.5 A related topic	•••••	15
Chapter 2 Near-Optimum Control of Lumped Parameter	er System	19
2.1 Introduction		19
2.2 Tracking problem	•••••	21
2.2.1 Problem statement		21
2.2.2 Asymptotic expansions	•••••	24
2.2.3 Basic theorems	•••••	28
2.3 Fixed-end-point minimum energy problem	• • • • • • • • • • • • •	32
2.3.1 Problem statement	• • • • • • • • • • • • • • •	32
2.3.2 Asymptotic expansions and approximate	solutions	35
2.3.3 Suboptimum trajectory	• • • • • • • • • • • • • • •	37
2.3.4 A simple example	•••••	39
2.3.5 Basic theorems		41
2.4 Systems with two small parameters		44
2.4.1 Problem statement		44
2.4.2 Asymptotic expansions		46
2.4.3 Basic theorems		51
2.5 Asymptotic behaviour of performance index		56

Chapter 3 N	uclear Reactor Control with Lumped Par	ameters	59
3.1 Err	or estimation of prompt jump approxima	ition	59
3.1.1	Introduction	•••••	59
3.1.2	Prompt jump approximation	•••••	60
3.1.3	Analysis via singular perturbation th	neory	62
3.1.4	Numerical examples	••••	66
3.2 Reg	ulator problem of a point reactor	••••	67
3.2.1	Formulation of the problem	•••••	67
3.2.2	Asymptotic expansions	••••	69
3.2.3	Boundary layer corrections	••••	72
3.2.4	Numerical example	•••••	74
Chapter 4 N	ear-Optimum Control of Distributed Par	ameter System	76
4.1 Int	roduction	• • • • • • • • • • • • • •	76
4.2 Evo	lution equations containing a small pa	arameter	78
4.2.1	Mathematical notations and preliminar	ies	78
4.2.2	Formal construction of asymptotic exp	ansion	81
4.2.3	Main theorems	•••••	86
4.2.4	Further generalization	• • • • • • • • • • • • • •	92
4.3 Tra	cking problem	•••••	96
4.3.1	Problem statement	• • • • • • • • • • • • • •	96
4.3.2	Eigenfunction expansions	• • • • • • • • • • • • • •	101
4.3.3	Asymptotic expansions via boundary la	yer method	105
4.3.4	Main theorem	••••	107
Chapter 5 N	uclear Reactor Control with Distribute	ed Parameters	111
5.1 Int	roduction	••••	111
5.2 For	mulation of the problem	•••••	112
5.2.1	Derivation of the state equation	•••••	112
5.2.2	Modal expansion	•••••	114
5.2.3	Consideration on the criticality cond	lition	118
5.2.4	Construction of asymptotic expansions	3	119
5.2.5	Numerical example	• • • • • • • • • • • • • •	123
5.3 Add	itional remarks	•••••	128

5.3.1	Spatially concentrated control	• • • • • • • • • • • • • •	128
5.3.2	Multi-group diffusion equation	• • • • • • • • • • • • • •	130
Chapter 6 Co	nclusions	• • • • • • • • • • • • •	131
Acknowledgmen	ts	• • • • • • • • • • • • • •	135
Appendices			
	a mi Dia i Statuta Décembri	Envetions	126

Appendix A	Theorems Related to Matrix Riccall Equations	100
Appendix B	A Matrix Inversion Formula	138
Appendix C	The Detailed Expression of Eq.(2.4.7)	139
Appendix D	Time-Invariant Regulator Problem [Y.73]	141
References		143

Chapter 1 Singular Perturbation Theory

In this chapter, an outline and survey consisting of four sections will be made. In Section 1.1, the aspects or situations of the singular perturbation theory are introduced by showing one of the simplest example. In Sections 1.2 and 1.3, a brief survey is made of the singular perturbation theory available at present to optimal control problems with lumped and distributed parameters.

The survey is not comprehensive if viewed from a purely mathematical point. The more complete review can be found in Vasil'eva's review paper [Vs.63.1], Wasow's laborious work [Ws.65], or in O'Malley [0.68.2]. The fourth section will be devoted to the description of preliminary results of the singular perturbations of ordinary differential equations. The proofs are omitted here because our main object lies only in that basic ideas and techniques of the singular perturbations are required to be demonstrated.

1.1 Introduction

Physicists or engineers often meet systems described by ordinary differential equations with parasitic parameters appearing as coefficients. Perturbation method is known well to be a good tool to investigate such systems. The method has its origin in the natural intuition, and in the experimental knowledge that there is a little difference between the trajectories of generating system and that of perturbed system. Analysis of the perturbed system is usually more intractable to be handled than the generating system. According to the "regular" perturbation method, the unknown quantities are expanded into power series with respect to a small parameter multiplying the perturbing term, and the coefficients of the expansions are determined stepwise by solving recursive equations, usually of the linear form. The method is powerful and practical in the fields of applied mathematics (Cole [C1.68]), fluid mechanics (Van Dyke [Vd.64]), nonlinear

oscillations (Hayashi [Hy.64], Nishikawa [N.64]), etc.

But among systems to be considered, there are peculiar ones including small parameters whose presence changes the order of the system. Such systems cannot be treated by the "regular" perturbation method. The "singular" perturbation theory is needed to investigate such singularly perturbed systems.

The question may arise how the "singular" perturbation theory differs from the "regular" perturbation theory. The following simplest example may show the singularity and complexity.

A scalar differential equation

$$\varepsilon \frac{d^2}{dt^2} x + \frac{d}{dt} x = 0,$$
 (1.1.1)

for instance, has the general solution

$$x = C_1 \exp(-t/\epsilon) + C_2,$$
 (1.1.2)

which is not continuous with respect to ε at $\varepsilon = 0$, unless t = 0. The quantities C_1 and C_2 are determined by subsidiary conditions of the problem. We consider here two-point boundary conditions, for example,

$$x(0) = \alpha, \quad x(1) = \beta.$$
 (1.1.3)

The exact solution is

$$x(t,\varepsilon) = \{1 - \exp(-\frac{1}{\varepsilon})\}^{-1}\{(\alpha-\beta)\exp(-\frac{t}{\varepsilon}) + \beta - \alpha\exp(-\frac{1}{\varepsilon})\} . (1.1.4)$$

Since for the positive ε , we have

$$\lim_{\epsilon \to 0} x(t,\epsilon) = \beta, \quad \text{for } t \in]0,1]. \quad (1.1.5)$$

The convergence is uniform in any closed t-interval $0 < \delta \leq t \leq 1$ but not in the whole interval $0 \leq t \leq 1$. The solution $x(t,0) = \beta$ is that of the reduced problem, obtained by letting $\varepsilon = 0$,

$$\frac{d}{dt} x = 0, \quad x(1) = \beta.$$
 (1.1.6)

In a narrow interval of width $O(\varepsilon)$ the solution (1.1.4) changes rapidly from $x(0,\varepsilon) = \alpha$ to a certain value neighboring $x(t,0) = \beta$ with difference of the order of ε . The outline of the situation is shown in Fig.1.1.1. Pearson [P.68] provides many examples of the boundary problems with illustrations.



Fig. 1.1.1 A schematic figure of a boundary layer.

Such an interval is called a "boundary layer", which is originally a terminology of fluid dynamics, because of a mathematical analogy with the boundary layers of fluid dynamics (see for example Van Dyke [Vd. 64]). For the negative ε we find that

$$\lim_{\epsilon \to 0} x(t,\epsilon) = \alpha, \qquad (1.1.7)$$

and the boundary layer occurs at t = 1.

In the preceding description, it is shown that the reduced solution does not generally satisfy the conditions of both ends. Also we can show in the same way that if initial value problems are considered, the reduced solution does not satisfy all the prescribed initial conditions. In other words, some of the prescribed conditions are lost in the reduced problem. Here we are confronted with the problem of what conditions are lost in the reduced problem, and how we can make corrections to the reduced solution. The singular perturbation theory provides us with right answers to these questions.

1.2 Singular perturbation theory for ordinary differential equations

Let us consider singular perturbations of a set of differential equations of the first order. In general, such systems can be written in the form

$$\frac{d}{dt} x = f(x, y, t, \varepsilon), \qquad x(0,\varepsilon) = \xi(\varepsilon), \qquad (1.2.1)$$

$$\varepsilon \frac{d}{dt} y = g(x, y, t, \varepsilon), \qquad y(0,\varepsilon) = \eta(\varepsilon),$$

where ϵ is a small positive parameter and x, f ϵE^n , y, g ϵE^m .

The reduced system (degenerate system) is obtained by setting $\epsilon = 0$ in (1.2.1) and by cancelling the initial condition for y.

$$\frac{d}{dt} x = f(x, y, t, \varepsilon), \qquad x(0,0) = \xi(0),$$

$$0 = g(x, y, t, \varepsilon).$$
(1.2.2)

The problem (1.2.1) is called the "full problem" and is said to degenerate regularly at $(x_0(t), y_0(t))$, the solution of the "reduced problem" (1.2.2) on [0,T], if the solution of (1.2.1) converges as $\varepsilon \rightarrow + 0$ to (x_0, y_0) uniformly on closed subsets of [0,T].

The singular perturbation theory has two aspects, the one is to investigate the regular degeneration of (1.2.1) and the other is to construct an asymptotic expansion of the solution of the full problem (1.2.1).

In both the aspects, the Jacobian matrix

$$\frac{\partial g}{\partial y}$$
 (x₀(t), y₀(t), t, 0) (1.2.3)

plays a principal role, the nonsingularity of which establishes the convergence and the solvability of the recursive equations in the outer region, which can be obtained through the outer problem:

$$\frac{d}{dt} x = f(x, y, t, \varepsilon), \qquad x(0,\varepsilon) = \xi(\varepsilon),$$

$$\varepsilon \frac{d}{dt} y = g(x, y, t, \varepsilon). \qquad (1.2.4)$$

A solution of (1.2.4), $x(t,\varepsilon)$, $y(t,\varepsilon)$, satisfies

$$x(t,0) = x_0(t), \quad y(t,0) = y_0(t).$$
 (1.2.5)

The inner problem

$$\frac{d}{d\tau} X = \varepsilon f(X, Y, \varepsilon\tau, \varepsilon), \qquad X(0, \varepsilon) = \xi(\varepsilon),$$

$$\frac{d}{d\tau} Y = g(X, Y, \varepsilon\tau, \varepsilon), \qquad Y(0, \varepsilon) = \eta(\varepsilon),$$
(1.2.6)

derived through the stretching transformation $\tau = t/\epsilon$, describes the solution of (1.2.1) for t near zero, or in the boundary layer.

The associated boundary layer problem

$$\frac{d}{d\tau} Y = g(x_0(t), Y, 0, 0), \qquad Y(0) = \eta(0), \qquad (1.2.7)$$

determines the asymptotic behavior of the solution of the inner problem (1.2.6).

Many papers on the regular degeneration of (1.2.1) require various stability properties of the system (1.2.7). Tikhonov [Tk.52] stated that if a certain solution of (1.2.7) is asymptotically stable as $\tau \rightarrow \infty$ then (1.2.1) degenerates regularly for each t restricted in a compact interval. His assumption was inadequate partly and was corrected by Hoppensteadt [Hp.67]. Levin and Levinson [Lv.54] gave the condition that all eigenvalues of (1.2.3) have negative real parts for t ε [0,T]. This condition is identical with the asymptotic stability of a solution

of (1.2.7). Hoppensteadt [Hp.66], [Hp.67] gave a proof of Tikhonov's theorem and extended the result to the case $T = \infty$.

Flatto and Levinson [F1.55] and Levin [Lv.56], [Lv.59], studied the regular degeneration of (1.2.1) under the conditional stability of the system (1.2.7). Levin showed that the full system (1.2.1) degenerates regularly if the initial data are given on the suitable manifold, called initial stable manifold. The conditional stability of (1.2.7) means that all the eigenvalues of (1.2.3) have nonzero real parts, or the system (1.2.3) possesses an exponential dichotomy (Coppel [Co.67]). Other studies of the conditionally stable case were made by Coppel and Chang [Cn.68], [Cn.69.1], [Cn.69.2]. Their results [Cn.69.1], [Cn.69.2] refine those of [F1.55], [Lv.56], [Lv.59], and [Cn.68] by removing certain smoothness conditions. Hoppensteadt [Hp. 71.1] also studied the conditional stability case, paying attention to the differentiability of the solution of (1.2.1) with respect to ε , while in [F1.55] - [Cn.69.2], the main object is with the continuity as $\varepsilon \rightarrow 0$ of the difference between the solution of (1.2.1) and that of (1.2.2).

As for the second aspect, i.e. constructing the asymptotic expansion solution, several researches have been made in various ways. From practical points of view, the second aspect is more important, which has been emphasized from many applications. Vasil'eva [Vs.63.1], Kaplun [Ka.61], Hoppensteadt [Hp.71.1], and O'Malley [0.71.1] have taken several approaches. Vasil'eva's method is similar to the popular method of inner and outer expansions. Kaplun [Ka.61] obtained expansion of the solution using the method of matched asymptotic expansions [Fk.69, Vd.64, Ku.67], which is directly applicable to partial differential equations (cf. Hendry and Bell [Hn.69, Hn.70, Hn.71]). In Hoppensteadt [Hp.71.1] and O'Malley [0.71.1] the solution of (1.2.1) was constructed by using boundary layer method with a little difference between them. All the expansions appearing in literatures [Vs. 63.1] - [0.71.1] can be shown to reduce to one another. Throughout these papers only the case where all eigenvalues of (1.2.3) have

negative real parts was studied. Tupčiev [Tp.62.1] extended Vasil'eva's result to the conditionally stable case.

Concerning two-point boundary value problems, several approaches have been made, e.g., by Harris [Hr.60], [Hr.62], Tupčiev [Tp.62.1], Vishik and Lyusternik [Vi.57], [Vi.58], Macki [Mc.67], O'Malley and Keller [0.68.1], O'Malley [0.69], [0.70.1], [0.70.2], Hadlock [Hd. 70.2], and Hoppensteadt [Hp.70], [Hp.71.1]. Compared with initial value problems, comprehensive methods have not yet been derived. Harris [Hr.60], [Hr.62] obtained some condition that ascertains regular degeneration of homogeneous two-point boundary value problem of the first order system. In O'Malley and Keller [0.68.1], certain cancellation law of boundary conditions was considered as to higher order scalar equations. The regular degeneration of a two-point boundary value problem arising in optimal control theory was also considered by Hadlock [Hd.70.2].

An asymptotic series solution was obtained in Tupćiev [Tp.62.1], Macki [Mc.67] for the nonlinear system of the first order equations. Vishik and Lyustenik [Vi.58], and O'Malley [0.70.1], [0.70.2] also considered asymptotic series solutions of a certain class of quasilinear system equations. Linear scalar equations were studied by Vishik and Lyusternik [Vi.57], O'Malley [0.69], etc. The comprehensive bibliography and survey on the subjects can be found in O'Malley [0. 68.2].

1.3 Singular perturbation theory for abstract evolution equations

The results of singular perturbations of ordinary differential equations have been extended to abstract evolution equations, including parabolic, hyperbolic, and integro-differential equations. (On the abstract evolution equation, parabolic equation etc, Dunford and Schwartz [Df.58] may be referred to.)

In Trenogin [Tr.63], the Cauchy problem in a real Banach space

$$\varepsilon \frac{d}{dt} y(t,\varepsilon) = F(y, t, \varepsilon), \qquad 0 < t \leq T,$$

$$y(0,\varepsilon) = 0,$$
 (1.3.1)

is considered, where both the regular degeneration and asymptotic solution of (1.3.1) are studied, by expanding the nonlinear operator $F(y, t, \varepsilon)$ into a Taylor series in an appropriate sense with respect to y and ε . Krein [Kr.67, Ch.4] investigated the asymptotic method on the Cauchy problem

$$\varepsilon \frac{d}{dt} y(t,\varepsilon) = A(t,\varepsilon)y(t,\varepsilon) + f(t), \qquad (1.3.2)$$

The more general Cauchy problem was also considered by Hoppensteadt [Hp.69.1], [Hp.69.2] of the form,

$$\varepsilon \frac{d}{dt} y(t,\varepsilon) + A(t,\varepsilon)y(t,\varepsilon) = f(t, y, \varepsilon), \qquad (1.3.3)$$

where - A (t, ε) is the generator of an analytic semigroup of bounded operators in E (parabolic case) [Hp.69.1]. In Hoppensteadt [Hp.69.2], both parabolic and hyperbolic cases were stated without proof.

In these studies [Tr.63] - [Hp.69.2], each problem is regarded as a Cauchy problem of an infinite dimensional ordinary differential equation in a Banach space, which is tractable on the basis of singular perturbations developed for ordinary differential equations reviewed in the preceding section.

Another approach to singular perturbations of partial differential equations, taken by Hoppensteadt [Hp.71.2] for the initial boundary value problem, appears as

$$\varepsilon \frac{\partial u}{\partial t} - \sum_{i,j=1}^{n} a_{ij}(x, t, u, \nabla u, \varepsilon) \frac{\partial^2 u}{\partial x_i \partial x_j} = f(x, t, u, \nabla u, \varepsilon) \quad (1.3.4)$$

$$u(x, t, \varepsilon) = 0 \qquad x \in bdry \Omega, \quad 0 \leq t \leq T,$$

$$u(x, 0, \varepsilon) = \vartheta(x, \varepsilon) \qquad x \in \overline{\Omega}$$

where the matrix (a_{ij}) is symmetric and positive definite uniformly in its arguments, and $\Omega \in E^n$. He suggested that the results obtained in [Hp.71.2] can be extended to the more general problems, e.g., systems of parabolic type, higher order parabolic equations with more general boundary conditions, and parabolic equations coupled with integro-differential equations.

The method of matched asymptotic expansion can be applied to partial differential equations (see, e.g., Hendry [Hn.69], [Hn.70], [Hn.71]) but a rigorous mathematical proof has not yet been derived for partial differential equations. However a heuristic theoretical consideration and many numerical experiments on the subject support the method [Dm.69].

The more general problems, however,

$$\frac{d}{dt} x(t,\varepsilon) = F(x, y, t, \varepsilon),$$

$$\varepsilon \frac{d}{dt} y(t,\varepsilon) = G(x, y, t, \varepsilon),$$
(1.3.5)

have rarely been treated. A certain class of such systems will be treated in the following Chapter 4.

1.4 Asymptotic solutions and basic theorems

Consider the initial value problem

$$\frac{d}{dt} x = f(x, y, t, \varepsilon), \qquad x(0,\varepsilon) = \xi(\varepsilon),$$

$$\varepsilon \frac{d}{dt} y = g(x, y, t, \varepsilon), \qquad y(0,\varepsilon) = \eta(\varepsilon),$$
(1.4.1)

where x, f εE^{n} , and y, g εE^{m} .

In order to construct asymptotic expansions of solutions, we expand x and y into Taylor series in ε :

$$x = x_0 + \varepsilon x_1 + \frac{\varepsilon^2}{2} x_2 + \dots,$$

$$y = y_0 + \varepsilon y_1 + \frac{\varepsilon^2}{2} y_2 + \dots,$$
(1.4.2)

Substituting these series into Eq.(1.4.1), and equating the coefficients of like powers of ε , we derive the recursive equation in the outer region, i.e., on t ε [0 + δ , T], δ = O(ε).

$$\frac{d}{dt} x_{0} = f(x_{0}, y_{0}, t, 0),$$

$$0 = g(x_{0}, y_{0}, t, 0),$$

$$\frac{d}{dt} x_{r} = f_{x}x_{r} + f_{y} y_{r} + p_{r-1},$$

$$\frac{d}{dt} y_{r-1} = g_{x}x_{r} + g_{y} y_{r} + q_{r-1},$$
(1.4.3)_r

where the remainders p_{r-1} and q_{r-1} depend only on t, $x_0(t)$, $y_0(t)$,, $x_{r-1}(t)$, $y_{r-1}(t)$.

The asymptotic solution (1.4.2) is valid in the interval except for the boundary layer $0 \le t \le 0(\varepsilon)$. The boundary layer correction is required here in order to make the solution valid uniformly in the whole interval considered. The basic idea depends upon Vasil'eva's method (see Section 3.1) but here we will show simpler results, following Hoppensteadt [Hp.71.1]. These results are equivalent to those derived by 0'Malley [0.71.1].

We shall seek a solution of the form

$$x = x(t,\varepsilon) + X(\tau,\varepsilon),$$

$$y = y(t,\varepsilon) + Y(\tau,\varepsilon),$$
(1.4.4)

where $x(t,\varepsilon)$ and $y(t,\varepsilon)$ have an asymptotic expansion (1.4.2) and correspond to the outer solution which represents "slow mode". And $X(\tau,\varepsilon)$ and $Y(\tau,\varepsilon)$ have also an asymptotic expansion

$$X(\tau, \varepsilon) = X_0(\tau) + \varepsilon X_1(\tau) + \frac{\varepsilon^2}{2} X_2(\tau) + ...,$$

$$Y(\tau, \varepsilon) = Y_0(\tau) + \varepsilon Y_1(\tau) + \frac{\varepsilon^2}{2} Y_2(\tau) + ...,$$
(1.4.5)

which is valid in the boundary layer and is required to tend to zero as $\tau \rightarrow \infty$. The quantities $X(\tau, \varepsilon)$ and $Y(\tau, \varepsilon)$ are called boundary layer correction terms and represent "fast mode". Substituting the expansions (1.4.2) and (1.4.5) into Eq.(1.4.4), we have an asymptotic solution uniformly valid

$$x = \sum_{\substack{r=0\\ m \in 0}}^{\infty} [x_{r}(t) + X_{r}(\tau)](r!)^{-1} \varepsilon^{r},$$

$$y = \sum_{r=0}^{\infty} [y_{r}(t) + Y_{r}(\tau)](r!)^{-1} \varepsilon^{r}.$$
(1.4.6)

In Vasil'eva's method, we must construct series solutions of three kinds, outer, inner, and intermediate solution. The boundary layer correction term is equivalent to the difference between the inner and intermediate solution. Then we have recursive equations for $X_i(\tau)$ and $Y_i(\tau)$

$$\begin{aligned} \frac{d}{d\tau} X_0 &= 0, \\ \frac{d}{d\tau} Y_0 &= g(x_0(0) + X_0, y_0(0) + Y_0, 0, 0), \\ \frac{d}{d\tau} X_r &= P_r(\tau), \\ \frac{d}{d\tau} Y_r &= g_x(x_0(0) + X_0, y_0(0) + Y_0, 0, 0) X_r \\ &+ g_y(x_0(0) + X_0, y_0(0) + Y_0, 0, 0) Y_r + Q_r(\tau). \end{aligned}$$
(1.4.7)

Here the remainders P_r and Q_r are polynomials consisting only of t, $x_0(\tau)$, $y_0(\tau)$, ..., $x_{r-1}(\tau)$, $y_{r-1}(\tau)$, X_0 , Y_0 , ..., X_{r-1} , Y_{r-1} . Initial conditions for (1.4.6) are determined by the prescribed conditions (1.4.1), i.e.,

$$x_r(0) + X_r(0) = \xi_r, \quad y_r(0) + Y_r(0) = \eta_r,$$
 (1.4.8)

where ξ_r and η_r are the coefficients of ϵ^r of Taylor series of ξ and η respectively. It is required that

$$\lim_{\tau \to \infty} X_{r}(\tau) = 0, \qquad \lim_{\tau \to \infty} Y_{r}(\tau) = 0. \qquad (1.4.9)$$

Now we can solve Eq. $(1.4.3)_0$ under the initial condition

$$x_0(0) = \xi_0,$$
 (1.4.10)

then from Eq. $(1.4.7)_0$ and (1.4.8), we have

$$X_{0}(\tau) = 0.$$
 (1.4.11)

Thus we must solve

$$\frac{d}{d\tau} Y_0 = g(x_0(0), y_0(0) + Y_0, 0, 0), \quad Y_0(0) = n_0 - y_0(0). \quad (1.4.12)$$

This problem can be written as

$$\frac{d}{d\tau} Y_0 = g_y Y_0 + q_0(Y_0)$$
(1.4.13)

where

$$q_0(Y_0) = o(|Y_0|), \quad \text{as} \quad |Y_0| \to 0.$$

If the initial data is given in the domain of influence of the asymptotically stable root $y = \phi(x)$ of

$$g(x, \phi(x), t, 0) = 0,$$
 (1.4.14)

then we can determine the leading terms X_0 , Y_0 , from (1.4.7)₀, (1.4.8), and (1.4.9).

 $X_1(\tau)$ can be determined from (1.4.7) as

$$X_{1}(\tau) = X_{1}(0) + \int_{0}^{\tau} P_{1}(\sigma) d\sigma.$$
 (1.4.15)

Assertion (1.4.9) requires $X_1(0)$ to satisfy

$$X_1(0) = -\int_0^{\infty} P_1(\sigma) d\sigma.$$
 (1.4.16)

This infinite integral exists since the integrand is dominated by functions of boundary layer type which decay exponentially. Then we obtain the initial data for the second coefficient of the outer expansion

$$x_1(0) = \xi_1 - X_1(0).$$
 (1.4.17)

Thus we can solve the first correction equation $(1.4.3)_1$ in the outer region. Then $y_1(0)$ can be determined from $(1.4.3)_1$, and we can solve the boundary layer correction equation $(1.4.7)_1$ for $Y_1(\tau)$. In this way, recursive equations (1.4.3) and (1.4.7) are solved stepwise.

Several conditions are needed to establish main theorems. First we state the prerequisite conditions, and then basic results are offered. These conditions are denoted by C1, C2, ..., as follows:

C1. The reduced problem $(1.4.3)_0$ has a continuous solution $x = x_0(t)$, $y = y_0(t)$, in some interval t ε [0,T].

C2. The functions f, g have continuous derivatives up to the order n + 2 with respect to their arguments (x, y, t) in some neighborhood of the points (x₀(t), y₀(t), t), t ε [0,T], and $\varepsilon \varepsilon$ [0, ε_0].

C3. The initial data $\xi(\epsilon)$, $\eta(\epsilon)$ are smooth functions of ϵ for $\epsilon \in [0, \epsilon_0]$.

The most important condition is from the assertion (1.4.9).

C4. The Jacobian matrix of (1.4.12) satisfies the condition

$$\frac{\partial g}{\partial y}(x_0(t), y_0(t), t, 0) \leq -\kappa < 0$$
 (1.4.18)

for each t $\in [0,T]$.

We can replace C4 by the following equivalent condition.

C4. All the eigenvalues of the Jacobian matrix $g_y(t)$ have negative real parts for each $t \in [0,T]$.

Lemma 1-1 If conditions C1 - C4 hold, then the reduced solution is uniquely determined.

Note that C4 implies that the solution of (1.4.12) is asymptotically stable with respect to the root $y = \phi(t_0,\xi)$ of

$$g(x_0, \phi(t_0, \xi), 0, 0) = 0.$$
 (1.4.19)

C5. The initial data exist in the domain of influence of the root $y = \phi(t_0,\xi)$.

The following theorem is due to Tikhonov [Tk.52]

<u>Theorem 1-1</u> If conditions C1 - C5 hold, then the following convergence relations between the reduced solution and the full solution are satisfied:

$$\lim_{\varepsilon \to 0} x(t,\varepsilon) = x_0(t), \quad \text{for } t \in [0,T], \quad (1.4.20)$$

$$\lim_{\varepsilon \to 0} y(t,\varepsilon) = y_0(t), \quad \text{for } t \in]0,T]. \quad (1.4.21)$$

Following Vasil'eva [Vs.63.1] we have an error estimation theorem:

<u>Theorem 1-2</u> If conditions C1 - C5 hold, then there exist $\varepsilon_0 > 0$ and functions $R^n(t,\varepsilon)$ and $S^n(t,\varepsilon)$ uniformly bounded in the interval considered, such that

$$x(t,\varepsilon) = \sum_{\substack{r=0\\n}}^{n} [x_{r}(t) + X_{r}(\tau)](r!)^{-1}\varepsilon^{r} + R^{n}(t,\varepsilon)\varepsilon^{n+1}, (1.4.22)$$

$$y(t,\varepsilon) = \sum_{r=0}^{n} [y_{r}(t) + Y_{r}(\tau)](r!)^{-1}\varepsilon^{r} + S^{n}(t,\varepsilon)\varepsilon^{n+1}, (1.4.23)$$

for $\varepsilon \in [0, \varepsilon_0]$.

If we restrict our attention to the outer expansion, we have the following theorem as a corollary of Theorem 1-2.

<u>Theorem 1-3</u> If conditions C1 - C5 hold, then there exist $\varepsilon_0 > 0$ and bounded functions R(t, ε) and S(t, ε) such that

$$x(t,\varepsilon) = \sum_{r=0}^{n} x_r(t)(r!)^{-1}\varepsilon^r + R^n(t,\varepsilon)\varepsilon^{n+1}, \text{ for } t \in [\delta,T], (1.4.24)$$
$$y(t,\varepsilon) = \sum_{r=0}^{n} y_r(t)(r!)^{-1}\varepsilon^r + S^n(t,\varepsilon)\varepsilon^{n+1}, \text{ for } t \in [\delta,T], (1.4.25)$$

where

$$\delta$$
 = -Celoge; C is independent of ϵ .

In Hoppensteadt [Hp.71.1] condition C4 or C4' is replaced by the more general condition, as follows:

C6. The matrix $g_y(t)$ has k eigenvalues, $1 \leq k \leq n$, with negative real parts satisfying $\operatorname{Re}(\lambda(t)) \leq -\mu$, and n-k eigenvalues with positive real parts satisfying $\operatorname{Re}(\lambda(t)) \geq \mu$ for $t \in [0,T]$.

1.5 A related topic

In this section, a relation between the Vasil'eva's method and the method of matched asymptotic expansions is considered with regard to the determination of the initial conditions of the recursive set of equations.

For simplicity, we consider the following set of equations:

$$\frac{d}{dt} x = f(x, y, t, \varepsilon),$$

$$\varepsilon \frac{d}{dt} y = g(x, y, t, \varepsilon),$$
(1.5.1)

where x and y are scalars. The initial condition is given as follows:

$$x(0) = \xi, \quad y(0) = \eta.$$
 (1.5.2)

The reduced system is obtained by setting $\varepsilon = 0$ in Eq.(1.5.1), with the initial condition for x. The first order correction of the outer expansion can be represented as a solution of

$$\frac{d}{dt} x_{1} = f_{x}x_{1} + f_{y}y_{1} + p_{1}(x_{0}, y_{0}),$$

$$\frac{d}{dt} y_{0} = g_{x}x_{1} + g_{y}y_{1} + q_{1}(x_{0}, y_{0}),$$
(1.5.3)

where f_x denotes the partial derivative of the function f with respect to x at (x_0, y_0) , similar notations are adopted for the others, and x_0 and y_0 are solutions determined by the reduced system.

Following Vasil'eva [Vs.63.1], the initial condition for the first order correction equation (1.5.3) is obtained as follows:

$$x_1(0) = \int_0^\infty [f(x_0(0), \overline{y}(\tau) - f(x_0(0), y_0(0))] d\tau,$$
 (1.5.4)

where $\overline{y}(\tau)$ is a solution of the boundary layer system associated with the full system (1.5.1) with (1.5.2),

$$\frac{d}{d\tau} \bar{y}(\tau) = g(x_0(0), \bar{y}(\tau), 0, 0) \qquad (1.5.5)$$

with the initial condition

$$y(0) = \eta.$$
 (1.5.6)

The conditions needed are assumed to hold so that we can apply the singular perturbation theory to the system (1.5.1) with (1.5.2). Then the boundary layer system (1.5.5) is asymptotically stable, i.e., the condition C4 in Section 1.4 holds

$$\frac{\partial g}{\partial y}(x_0(t), y_0(t), t, 0) \leq -\kappa < 0$$
(1.5.7)

for each $t \in [0,T]$.

Since Lemma 1-1 gives the solvability of the reduced problem, we may represent formally y_0 in the form of a function of x_0 ,

$$y_0 = \phi(x_0)$$
 (1.5.8)

which is a solution of

$$g(x_0(t), y_0(t), t, 0) = 0.$$
 (1.5.9)

If the condition (1.5.7) holds, then we obtain

$$\lim_{\tau \to \infty} \frac{d\overline{y}}{d\tau} = \lim_{\tau \to \infty} g(x_0(0), \overline{y}(\tau), 0, 0) = 0.$$
(1.5.10)

Considering Eqs.(1.5.9) and (1.5.10) collectively, we have the following equality relation:

$$\lim_{\tau \to \infty} y(\tau) = y_0(0), \qquad (1.5.11)$$

if the function g: $x \rightarrow y$ is one-valued. In the case where g is a multivalued function, Eq.(1.5.11) also holds by taking an appropriate branch $\phi(x)$ satisfying Eq.(1.5.9).

The matching principle of the method of matched asymptotic expansions is stated as below [Vn.64]:

lim [the k-th order solution of the inner expansion]
T→∞
= lim [the k-th order solution of the outer expansion], (1.5.12)
t→0

which is equivalent to Eq.(1.5.11) for k = 0. For the higher order, the similar argument is carried out. Extensions to the case where x and y are multi-dimensional vectors are easily obtained.

The outline of the above situation is shown in Fig.1.5.1



Fig. 1.5.1 A schematic figure of limit relations.

Chapter 2 Near-Optimum Control of Lumped Parameter System

2.1 Introduction

In this chapter, a finite dimensional model is treated, which is described by a set of ordinary differential equations. Such models may offer a description of a broad class of realistic physical systems.

In the optimization problem of such a system of the high order, it is urged from practical points of view to develop an effective synthesis method based on an approximate model, "low order model", obtainable by reducing the system order. Designers usually make the model order to be lower by neglecting some small parasitic parameters, e.g., time-constants, masses, moments of inertia, inductance, capacitance, and so forth, whose presence causes the model order higher than acceptable for practical treatment. For instance the numerical calculation of optimal control needs increased endeavor with the increase of the order of the control system, because the generated canonical system equation is "2n-dimensional" when the original system is "ndimensional".

Such a conventional "low-order design" is made on the basis of the designer's intuition or his practical experiences. The "low-order design" has some drawbacks, e.g., that it may destroy original stable properties of more accurate "high-order design" (see for example [Ko. 68]), or may not maintain controllability and observability of the original system. Moreover no means are available to improve the loworder approximation because of the absence of systematic and theoretical basis.

The singular perturbation theory offers very useful and powerful tools to the problems. Kokotović and Sannuti [Ko.68] considered first the singular perturbations of optimal control problems of certain class. They analyzed the regular degeneration of control and state vector,

and constructed the higher correction term in the outer region by using Vasil'eva's theory [Vs.63.1]. In singular perturbations, the stability of boundary layer system plays an important role, mentioned on occasion in the preceding chapter. In the study [Ko.68] of nonlinear regulator problem it is needed that the boundary layer system be asymptotically stable as $\tau \rightarrow \infty$. The result was improved in Sannuti and Kokotović [Sa.69.1], by removing the assumption on continuity and different-iability of control with respect to a singularly perturbing parameter (see also [Sa.69.1]).

In Sannuti and Kokotović [Sa.69.2], another approach is taken by utilizing the Riccati transformation. They studied linear regulator problems under the condition that the state matrix of the boundary layer system should be stable and the full system should satisfy a special condition. Kokotović and Yackel [Ko.72] also studied linear regulator problems by introducing new concepts of boundary layer controllablility and boundary layer observability; the theorem obtained in [So.69.2] was refined and generalized by using the new concepts, and was derived as a corollary of the main theorem in [Ko.72].

A quite different approach to the problem was made in O'Malley [0. 72.1], Hadlock et al. [Hd.70.1], [Hd.70.2], and Sannuti [Sa.71]. In these studies, the so-called boundary layer method proposed in O'Malley [0.71.1] or in Hoppensteadt [Hp.71.1] was adopted in order to obtain a temporally uniform approximate solution of the asymptotic expansion form. The results of O'Malley [0.72.1] and Hadlock [Hd.70.2] are more general than those of Sannuti [Sa.71] in that their results correspond to the conditionally stable case in Section 1.1., which all the eigenvalues of boundary layer system of the state equation concerned, have negative real part.

Boundary layer method is also applied to matrix Riccati differential equations in Yackel and Kokotović [S.73.1].

In this chapter, the author shall show the most generalized results of the singular perturbation theory of optimal control problems derived by him. The chapter treats three problems occuring typically in optimi-

zation problems, and covers many variations. In the following section, we derive basic results on tracking problems. The results can be easily extended to the linear regulator problem, yielding theorem under hypotheses different from those of Yackel and Kokotović [Y.73]. In Section 2.3, fixed-end-point, fixed-end-time, minimum energy problems will be treated. The generalized Riccati transformation is introduced to treat the ill-conditioned two-point boundary value problems generated by the fixed-terminal problems. Section 2.4 deals with the system involving two singularly perturbing parameters of different orders of smallness. The results of this section can be easily applied to systems involving more parameters. The last section is devoted to the study of the asymptotic behaviour of the performance index.

2.2 Tracking problem

2.2.1 Problem statement

Let the state equation be

$$\frac{d}{dt} x(t,\varepsilon) = A(t,\varepsilon)x(t,\varepsilon) + B(t,\varepsilon)u(t,\varepsilon), \qquad (2.2.1)$$

$$A = \begin{bmatrix} A_1 & A_2 \\ \varepsilon^{-1}A_3 & \varepsilon^{-1}A_4 \end{bmatrix}, B = \begin{bmatrix} B_1 \\ \varepsilon^{-1}B_2 \end{bmatrix}, x = \begin{bmatrix} x_1 \\ x_2 \end{bmatrix},$$

where x_1 is an n-dimensional state vector, x_2 m-dimensional, control vector u r-dimensional, and A_i 's, B_i 's consistent dimensional matrices respectvely. A small positive parameter is expressed by ε .

The tracking problem is to minimize the performance index

$$J = \frac{1}{2} e(t_f)' Fe(t_f) + \frac{1}{2} \int_{t_0}^{t_f} [e(t)'Qe(t) + u(t)'Ru(t)]dt, \qquad (2.2.2)$$

where

$$\mathbf{F} = \begin{bmatrix} \mathbf{F}_1 & \varepsilon \mathbf{F}_2 \\ \varepsilon \mathbf{F}_2' & \varepsilon \mathbf{F}_3 \end{bmatrix}, \quad \mathbf{Q} = \begin{bmatrix} \mathbf{Q}_1 & \mathbf{Q}_2 \\ \mathbf{Q}_2' & \mathbf{Q}_3 \end{bmatrix}, \quad (2.2.3)$$

under the prescribed initial conditions

$$x(t_0) = \xi,$$
 (2.2.4)

or in the partitioned form

$$\begin{bmatrix} \mathbf{x}_1(\mathbf{t}_0) \\ \mathbf{x}_2(\mathbf{t}_0) \end{bmatrix} = \begin{bmatrix} \xi_1 \\ \xi_2 \end{bmatrix}.$$

The partitioned form of F shall prove to be logical in the sequel. The vector e(t) is an error vector of 1-dimensional

$$e(t) = z(t) - y(t)$$
 (2.2.5)

where z(t) is a desired vector of 1-dimensional and y(t) is an output vector of 1-dimensional

y(t) =
$$Cx(t) = C_1 x_1(t) + C_2 x_2(t),$$

C = $[C_1, C_2].$
(2.2.6)

As well-known, the optimal control u* is given, through the constructing Hamiltonian, as below (Athan and Falb [At.66]):

$$u^* = -R^{-1}B'p. (2.2.7)$$

We have the canonical equation by eliminating costate vector p,

$$\begin{pmatrix} \mathbf{x} \\ \mathbf{p} \\ \mathbf{p} \end{pmatrix} = \begin{pmatrix} \mathbf{A} & -\mathbf{B}\mathbf{R}^{-1}\mathbf{B}^{*} \\ -\mathbf{Q} & -\mathbf{A}^{*} \end{pmatrix} \begin{bmatrix} \mathbf{x} \\ \mathbf{p} \end{bmatrix}$$
(2.2.8)

The equation (2.2.8) is to be solved under the boundary conditions

$$x(t_0) = \xi,$$
 (2.2.9a)

$$p(t_f) = Fe(t_f).$$
 (2.2.9b)

In the course of deriving the canonical equation (2.2.8), we can easily see that the costate vector p_2 corresponds to the quantity of x_2/ϵ . Then the form of terminal condition (2.2.9b) supports the partitioned form of F mentioned previously.

Several methods are proposed to handle such two-point boundary value problems, e.g., sweep method (Gelfand and Fomin [G1.63]) shooting method (Keller [K1.68]), invariant imbedding (Bailey and Wing [B1.65]), etc.

The Riccati transformation is known well in the field of optimal control as one of initial value methods which transform the original two-point boundary value problem into a single terminal value problem. The Riccati transformation is rigorously based upon the linear space theory. The validity of the transformation is proved in many articles (see for instance Meyer [My.73]).

The transformation is defined by

$$p(t) = K(t)x(t) + g(t).$$
 (2.2.10)

By eliminating p(t) in Eq.(2.2.8) by using Eq.(2.2.10), the following differential equation of matrix Riccati type is obtained:

$$\dot{K} = -KA - A'K + KBR^{-1}B'K - C'QC,$$
 (2.2.11)

with the associated differenatial equation

$$\dot{g} = -[A - BR^{-1}B'K]'g.$$
 (2.2.12)

The terminal conditions for K(t), and g(t) are given as follows from Eqs.(2.2.9b) and (2.2.10):

$$K(t_f) = C'(t_f)FC(t_f),$$
 (2.2.13)

$$g(t_f) = -C'(t_f)F_Z(t_f).$$
 (2.2.14)

The optimum trajectory is a solution of

$$x = [A - BR^{-1}B'K]x + BR^{-1}B'g, \qquad (2.2.15)$$

under the initial value condition (2.2.9a)

It is to be noted that the partitioned form of F suggests that the effect of the state vector $x_2(t)$ upon the terminal cost should be little. In other words, if the order of each element of F is roughly equal to each other, the elements F_2 and F_3 should take a greater value than F_1 .

2.2.2 Asymptotic expansions

The asymptotic property of the original system (2.2.1) as $\epsilon \to 0$, indicates that K has the partitioned form

$$K = \begin{bmatrix} K_1 & \varepsilon K_2 \\ \varepsilon K_2' & \varepsilon K_3 \end{bmatrix}, \qquad (2.2.16)$$

and for g

$$g = \begin{bmatrix} g_1 \\ \varepsilon g_2 \end{bmatrix}, \qquad (2.2.17)$$

where K_1 is an nxn matrix, K_2 nxm, K_3 mxm, and g_1 is an n-dimensional vector, and g_2 m-dimensional. The partitioning is seen reasonable by simple manipulations of direct substitutions. Moreover the partitioned forms of K and g, and the terminal conditions (2.2.12), (2.2.13) for K and g also support the partitioned form of F.

Equation (2.2.11) can be written for each element K_i as

$$\dot{\mathbf{K}}_{1} = -\mathbf{K}_{1}\mathbf{A}_{1} - \mathbf{A}_{1}\mathbf{K}_{1} - \mathbf{K}_{2}\mathbf{A}_{3} - \mathbf{A}_{3}\mathbf{K}_{2} + \mathbf{K}_{1}\mathbf{E}_{1}\mathbf{K}_{1} + \mathbf{K}_{2}\mathbf{E}_{2}\mathbf{K}_{1} + \mathbf{K}_{1}\mathbf{E}_{2}\mathbf{K}_{2}^{'} + \mathbf{K}_{2}\mathbf{E}_{3}\mathbf{K}_{2}^{'} - \mathbf{C}_{1}\mathbf{Q}_{1}\mathbf{C}_{1}, \qquad (2.2.18a)$$

$$\varepsilon\dot{\mathbf{K}}_{2} = -\mathbf{K}_{1}\mathbf{A}_{2} - \mathbf{K}_{2}\mathbf{A}_{4} - \mathbf{A}_{3}\mathbf{K}_{3} + \mathbf{K}_{1}\mathbf{E}_{2}\mathbf{K}_{3} + \mathbf{K}_{2}\mathbf{E}_{3}\mathbf{K}_{3} - \mathbf{C}_{1}\mathbf{Q}_{2}\mathbf{C}_{2} + \varepsilon(-\mathbf{A}_{1}\mathbf{K}_{2} + \mathbf{K}_{1}\mathbf{E}_{1}\mathbf{K}_{2} + \mathbf{K}_{2}\mathbf{E}_{2}\mathbf{K}_{2}), \qquad (2.2.18b)$$

 $\varepsilon \dot{K}_{3} = -K_{3}A_{4} - A_{4}K_{3} + K_{3}E_{3}K_{3} - C_{2}Q_{3}C_{2} + \varepsilon (-K_{2}A_{2} - A_{2}K_{2})$

+
$$K_3 E_2' K_2 + K_2' E_2 K_3$$
) + $\varepsilon^2 K_2' E_1 K_2$. (2.2.18c)

And for g_i's, we have

$$\dot{g}_{1} = -(A_{1} - E_{1}K_{1} - E_{2}K_{2}')'g_{1} - (A_{3} - E_{2}'K_{1} - E_{3}K_{2}')'g_{2} - C_{1}Q_{1}z_{1}$$

$$-C_{2}Q_{2}z_{2}, \qquad (2.2.19a)$$

$$\varepsilon \dot{g}_{2} = -(A_{2} - \varepsilon E_{1}K_{2} - E_{2}K_{3})'g_{2} - (A_{4} - \varepsilon E_{2}'K_{2} - E_{3}K_{3})'g_{2} - C_{1}Q_{2}'z_{1}$$

$$-C_{2}Q_{3}z_{2}, \qquad (2.2.19b)$$

where

$$E_1 = B_1 R^{-1} B_1', \quad E_2 = B_1 R^{-1} B_2', \quad E_3 = B_2 R^{-1} B_2'.$$

The reduced system for K_i^0 's, is obtained by setting $\varepsilon = 0$

$$\dot{K}_{1}^{0} = -K_{1}^{0}A_{1} - A_{1}'K_{1}^{0} - K_{2}^{0}A_{3} - A_{3}'K_{2}^{0} + K_{1}^{0}E_{1}K_{1}^{0} + K_{2}^{0}E_{2}'K_{1}^{0} + K_{1}^{0}E_{2}K_{2}^{0'} + K_{2}^{0}E_{3}K_{2}^{0'} + C_{1}'Q_{1}C_{1}, \qquad (2.2.20a)$$

$$0 = -K_1^0 A_2 - K_2^0 A_4 - A_3' K_3^0 + K_1^0 E_2 K_3^0 + C_1' Q_1 C_2, \qquad (2.2.20b)$$

$$0 = -K_{3}^{0}A_{4} - A_{4}'K_{3}^{0} + K_{3}^{0}E_{3}K_{3}^{0} + C_{2}'Q_{3}C_{2}.$$
 (2.2.20c)

For g_i^0 's the reduced system is

$$\dot{g}_{1}^{0} = -(A_{1} - E_{1}K_{1}^{0} - E_{2}K_{2}^{0}')'g_{1}^{0} - (A_{3} - E_{2}'K_{1}^{0} - E_{3}K_{2}^{0}')'g_{2}^{0}$$

$$- C_{1}Q_{1}z_{1} - C_{2}Q_{2}z_{2}, \qquad (2.2.21a)$$

$$0 = -(A_{2} - E_{2}'K_{3}^{0})'g_{1}^{0} - (A_{4} - E_{3}K_{3}^{0})'g_{2}^{0} - C_{1}Q_{2}'z_{1} - C_{2}Q_{3}z_{2}. \qquad (2.2.21b)$$

The terminal conditions for these systems are given by, using the cancellation law of O'Malley et al. [0.68]

$$K_1^0(t_f) = C_1'(t_f)F_1C_1(t_f),$$
 (2.2.22)

and

$$g_1^0(t_f) = -C_1'(t_f)F_1z_1(t_f),$$
 (2.2.23)

where z_1 corresponds to the vector $C_1 X_1$.

The first correction system can be derived by differentiating Eq.(2.2.17) with respect to ε and setting $\varepsilon = 0$, as follows:

$$\dot{\mathbf{K}}_{1}^{1} = -\mathbf{K}_{1}^{1}\mathbf{A}_{1} - \mathbf{A}_{1}'\mathbf{K}_{1}^{1} - \mathbf{K}_{2}^{1}\mathbf{A}_{3} - \mathbf{A}_{3}'\mathbf{K}_{2}^{1} + \mathbf{K}_{1}^{1}\mathbf{E}_{1}\mathbf{K}_{1}^{0} + \mathbf{K}_{1}^{0}\mathbf{E}_{1}\mathbf{K}_{1}^{1} + \mathbf{K}_{2}^{1}\mathbf{E}_{2}'\mathbf{K}_{1}^{0} + \mathbf{K}_{2}^{0}\mathbf{E}_{2}'\mathbf{K}_{1}^{1} + \mathbf{K}_{1}^{1}\mathbf{E}_{2}\mathbf{K}_{2}^{0'} + \mathbf{K}_{1}^{0}\mathbf{E}_{2}\mathbf{K}_{2}^{1'} + \mathbf{K}_{2}^{1}\mathbf{E}_{3}\mathbf{K}_{2}^{0'} + \mathbf{K}_{2}^{0}\mathbf{E}_{3}\mathbf{K}_{2}^{1'}, \qquad (2.2.24a)$$
$$0 = -\mathbf{K}_{1}^{1}\mathbf{A}_{2} - \mathbf{K}_{2}^{1}\mathbf{A}_{4} - \mathbf{A}_{1}'\mathbf{K}_{2}^{0} - \mathbf{A}_{3}'\mathbf{K}_{3}^{1} + \mathbf{K}_{1}^{0}\mathbf{E}_{1}\mathbf{K}_{2}^{0} + \mathbf{K}_{2}^{0}\mathbf{E}_{2}'\mathbf{K}_{2}^{0} + \mathbf{K}_{1}^{0}\mathbf{E}_{2}\mathbf{K}_{3}^{1} + \mathbf{K}_{1}^{1}\mathbf{E}_{2}\mathbf{K}_{3}^{0} + \mathbf{K}_{1}^{0}\mathbf{E}_{3}\mathbf{K}_{2}^{0} - \dot{\mathbf{K}}_{2}^{0}, \qquad (2.2.24b)$$
$$0 = -\mathbf{K}_{3}^{1}\mathbf{A}_{4} - \mathbf{A}_{4}'\mathbf{K}_{3}^{1} + \mathbf{K}_{3}^{1}\mathbf{E}_{3}\mathbf{K}_{3}^{0} + \mathbf{K}_{3}^{0}\mathbf{E}_{3}\mathbf{K}_{3}^{1} - \mathbf{K}_{2}^{0'}\mathbf{A}_{2} - \mathbf{A}_{2}'\mathbf{K}_{2}^{0} + \mathbf{K}_{3}^{0}\mathbf{E}_{2}'\mathbf{K}_{3}^{0} + \mathbf{K}_{2}^{0}\mathbf{E}_{2}\mathbf{K}_{3}^{0} - \dot{\mathbf{K}}_{3}^{0}. \qquad (2.2.24c)$$

For
$$g_1^j$$
, we have
 $g_1^1 = -(A_1 - E_1K_1^0 - E_2K_2^0)'g_1^1 - (A_3 - E_2'K_1^0 - E_3K_2^0')'g_2$
 $+ (E_1K_1^1 + E_2K_2^1')g_1^0 + (E_2'K_1^1 + E_3K_2^1')'g_2^0,$ (2.2.25a)
 $0 = -(A_2 - E_2K_3^0)'g_1^1 + (A_4 - E_3K_3^0)'g_2^1 + (E_1K_2^0 + E_2K_3^1)'g_1^0$
 $+ (E_2'K_2^0 + E_3K_3^1)'g_2^0 - g_2^0,$ (2.2.25b)

where it is assumed that A_i 's, B_i 's, C_i 's, Q_i 's, etc. are constant with respect to ε in order to avoid introducing superfluous notations, but this assumption is not essential. The first correction system (2.2.24) - (2.2.25) can also be obtained in the following way.

Expanding K_i's and g_i's into Taylor series in $\epsilon,$ we get

$$K_{i} = K_{i}^{0} + \varepsilon K_{i}^{1} + \frac{\varepsilon^{2}}{2} K_{i}^{2} + \dots, \quad i = 1, 2, 3, (2.2.26)$$

$$g_i = g_i^0 + \epsilon g_i^1 + \frac{\epsilon^2}{2} g_i^2 + \dots, \quad i = 1, 2.$$
 (2.2.27)
Substituting (2.2.26) and (2.2.27) into Eqs.(2.2.17) and (2.2.18b), and equating the coefficients of like powers of ε , we have the reduced system as a coefficient of ε^{q} , and the first correction system as a coefficient of ε . The higher correction systems can be derived by using the same procedures.

The terminal conditions under which Eqs.(2.2.24) and (2.2.25) are solved are not generally set to be zero. This situation is peculiar to the singular perturbation theory. Following Vasil'eva's integral formula [Vs.63.1], we obtain the relations between the terminal conditions and the associate boundary layer systems.

The boundary layer system plays a crucial role in the singular perturbation theory; in this case, for K it results in

$$-\frac{d}{d\tau} \overline{K}_{2} = -K_{1}^{0}A_{2} - \overline{K}_{2}A_{4} - A_{3}'\overline{K}_{3} + K_{1}^{0}E_{2}\overline{K}_{3} + \overline{K}_{2}E_{3}\overline{K}_{3} - C_{1}'Q_{2}C_{2},$$

$$(2.2.28)$$

$$-\frac{d}{d\tau} \overline{K}_{3} = -\overline{K}_{3}A_{4} - A_{4}'\overline{K}_{3} + \overline{K}_{3}E_{3}\overline{K}_{3} - C_{2}'Q_{3}C_{2}.$$

$$(2.2.29)$$

 $-\frac{d}{d\tau}\overline{g}_{2} = -(A_{2} - E_{2}\overline{K}_{3})'g_{1}^{0} - (A_{4} - E_{3}\overline{K}_{3})'\overline{g}_{2} - C_{1}Q_{2}'z_{1} - C_{2}Q_{3}z_{2},$ (2.2.30)

where the independent variable is τ , and t is regarded as a fixed parameter. The above system is considered as a kind of stretched system derived by the "stretching transformation" $\tau = [(t_f - t)/\epsilon], \tau \in [0,\infty]$. The terminal conditions for these equations are

$$\overline{K}_{2}(t_{f}) = C_{1}'(t_{f})F_{2}C_{2}(t_{f}),$$
 (2.2.31)

$$\overline{K}_{3}(t_{f}) = C_{2}'(t_{f})F_{3}C_{2}(t_{f}),$$
 (2.2.32)

and

For g, we get

$$\overline{g}_{2}(t_{f}) = -C_{2}'(t_{f})F_{3}z_{2}(t_{f}).$$
 (2.2.33)

Then the terminal conditions for the first correction equations are given as follows:

$$K_{1}(t_{f}) = \int_{0}^{\infty} [(-\overline{K}_{2}A_{3} - A_{3}'\overline{K}_{2} + \overline{K}_{2}E_{2}'K_{1}^{0} + K_{1}^{0}E_{2}\overline{K}_{2}' + \overline{K}_{2}E_{3}\overline{K}_{2}') - (-K_{2}^{0}A_{3} - A_{3}'K_{2}^{0} + K_{2}^{0}E_{2}'K_{1}^{0} + K_{1}^{0}E_{2}K_{2}^{0} + K_{2}^{0}E_{3}K_{2}^{0}')]d\tau,$$

$$(2.2.34)$$

$$(2.2.34)$$

$$g_{1}(t_{f}) = \int_{0}^{[E_{2}(K_{2}' - K_{2}^{0}')g_{1}^{0} - (A_{3} - E_{2}'K_{1}^{0} - E_{3}K_{2}')g_{2}} - (A_{3} - E_{2}'K_{1}^{0} - E_{3}K_{2}^{0}')g_{2}^{0}]d\tau, \qquad (2.2.35)$$

where \overline{K}_2 , \overline{K}_3 and \overline{g}_2 are solutions of the boundary layer systems (2.2.28) - (2.2.30) with the boundary conditions (2.2.31) - (2.2.33).

A similar procedure is applied to solve the recursive equation for the higher order, and it is possible to construct an approximate solution whose accuracy is a desired one.

2.2.3 Basic theorems

We have derived a representation of the approximate solution in the form of asymptotic expansions. Here the basic theorems are given upon which the validity of the approximation depends.

In the regular optimization problem, the following conditions are usually assumed to hold for t $\in [t_0, t_f]$, and $\varepsilon \in [0, \varepsilon_0]$:

C1. A_i 's and B_i 's are holomorphic with respect to t and ε .

C2. R and Q are positive definite and holomorphic with respect to t and $\boldsymbol{\epsilon}.$

C3. F is time invariant, positive semidefinite and continuous with respect to ε .

C4. z is bounded, and holomorphic with respect to ε .

We state below the conditions C5 and C6, essential to the singular perturbation theory.

C5. The system (2.2.1) is boundary layer controllable.

In other words,

rank
$$[B_2, A_4 B_2, A_4^2 B_2, \dots, A_4^{m-1} B_2] = m.$$

C6. A_4 is a stable matrix.

We first state a main theorem which gives asymptotic accuracy.

<u>Theorem 2-1</u> If conditions C1 - C6 hold, then there exist functions $U_i^k(t,\epsilon)$ and $V_i^k(t,\epsilon)$ bounded in the outer region such that

$$\begin{split} & K_{1}(t) = \sum_{\substack{j=0 \\ k}}^{k} K_{1}^{j}(t) \varepsilon^{j}(j!)^{-1} + U_{1}(t,\varepsilon) \varepsilon^{k+1}, \text{ for } t \in [t_{0}, t_{f}^{-\delta}], \\ & (2.2.36a) \\ & K_{i}(t) = \sum_{\substack{j=0 \\ k}}^{k} K_{i}^{j}(t) \varepsilon^{j}(j!)^{-1} + U_{i}(t,\varepsilon) \varepsilon^{k+1}, \text{ for } t \in [t_{0}, t_{f}^{-\delta}]; \\ & i = 2,3, (2.2.36b,c) \\ & g_{1}(t) = \sum_{\substack{j=0 \\ k}}^{k} g_{1}^{j}(t) \varepsilon^{j}(j!)^{-1} + V_{1}(t,\varepsilon) \varepsilon^{k+1}, \text{ for } t \in [t_{0}, t_{f}^{-\delta}], \\ & (2.2.37a) \\ & g_{2}(t) = \sum_{\substack{j=0 \\ j=0}}^{k} g_{2}^{j}(t) \varepsilon^{j}(j!)^{-1} + V_{2}(t,\varepsilon) \varepsilon^{k+1}, \text{ for } t \in [t_{0}, t_{f}^{-\delta}], \\ & (2.2.37b) \end{split}$$

where

$$\delta = -C \epsilon \log \epsilon$$
; C is independent of ϵ .

The proof of this theorem is carried out by proving the following three lemmas. In Lemma 2-1 it is shown that the solution $K_3(\tau)$ of the boundary layer system (2.2.29) is asymptotically stable with respect to the positive definite root $K_3(t)$ of Eq.(2.2.20c) for $t = t_f$. The similar result for $K_2(\tau)$ is shown in Lemma 2-2. Lemma 2-3 establishes the existence and uniqueness of the solution of the reduced system (2.2.20). Lemma 2-2 corresponds to Lemma 1-1 in Chapter 1. These facts make Theorem 1-3 applicable to this problem.

Lemma 2-1 If conditions C1- C6 hold, then the boundary layer equation (2.2.29) has an isolated and asymptotically stable solution as $\tau \rightarrow \infty$, which is positive definite and is obtained by equating the right hand side of Eq.(2.2.29) to zero. Moreover, the matrix

 $-(-A_4 + K_3Q_3)$ is stable.

<u>proof</u> The lemma is proved from the results obtained in proper theorems in Reid [R.63], [R.65] (see Appeendix A) dealing with two point boundary value problems associated with the state regulator problem. In the boundary layer system (2.2.24), t is considered as a fixed parameter, so that every coefficient matrix appearing in Eq.(2. 2.29) is time invariant. Conditions C1 - C6 may involve every assumption in the Theorems A-2 and A-3 in Appendix A. Q.E.D.

<u>Lemma 2-2</u> If conditions C1 - C6 hold, then the boundary layer equations (2.2.29) and (2.2.30) have, respectively, an isolated and asymptotically stable solution, which is given as in Lemma 2-1.

proof The Jacobian matrix of Eq.(2.2.28) is given by

$$-(-A_4 + K_3Q_3)$$

which is a stable matrix as $\tau \to \infty$ by Lemma 2-1. Hence the solution $K_2(t_f)$ of Eq.(2.2.28) is asymptotically stable with respect to the equilibrium $K_2(t_f)$ obtained in the same manner as in Lemma 2-1. O.E.D.

<u>Lemma 2-3</u> ([Ko.72]) If conditions Cl - C6 hold, then the solution of the reduced system (2.2.20) exists and is unique on the interval $[t_0, t_f]$.

<u>proof</u> We can derive the following matrix Riccati equation by manipulating Eq.(2.2.20).

$$\dot{K}_{1}^{0} = -K_{1}^{0}A_{0} - A_{0}'K_{1}^{0} + K_{1}^{0}B_{0}R^{-1}B_{0}K_{1}^{0} - Q_{0},$$

$$K_{1}(t_{f}) = C_{1}'(t_{f})F_{1}C_{1}(t_{f}),$$
(2.2.38)

where

$$A_0 = A_1 + H_1A_3 + E_2H_2' + H_1E_1H_2',$$

$$B_{0} = B_{1} + H_{1}B_{2},$$

$$Q_{0} = -H_{2}A_{3} - A_{3}'H_{2}' - H_{2}E_{3}H_{2}' + Q_{1},$$

$$H_{1} = (E_{2}K_{3} - A_{2})(A_{4} - E_{3}K_{3})^{-1},$$

$$H_{2} = (A_{3}'K_{3} + Q_{2})(A_{4} - E_{3}K_{3})^{-1},$$

and K_3 is a positive definite solution of Eq.(2.2.20c). The well-known theory of optimal control [At.66] ascertains that if R is positive definite, Q and F are positive semidefinite, then the solution K_1 exists and is unique in the interval $[t_0, t_f]$. In order to show the positive definiteness of Q_0 , we derive

$$v'Q_{0}v = w'Q_{3}w + v'Q_{2}w + w'Q_{2}'v + v'Q_{1}v$$

= [v', w'] $\begin{bmatrix} Q_{1} & Q_{2} \\ Q_{2}' & Q_{3} \end{bmatrix} \begin{bmatrix} v \\ w \end{bmatrix} > 0,$ (2.2.39)

where v is an arbitrary vector and w is given by

$$w = -(A_4K_3 + Q_3)'^{-1}(Q_2 + A_3'K_3)'v. \qquad (2.2.40)$$

Hence Q_0 is positive definite, which completes the proof. Q.E.D.

Lemma 2-1 - 2-3 establish the conditions needed under which Theorem 1-3 holds. Therefore the proof of Theorem 2-1 is completed. It is remarked that the result of this section has been deduced under the conditions different from that Kokotović and Yackel [Ko.72] who adopted the following condition C7 in place of our condition C6.

C7. The system is boundary layer observable, i.e.,

rank[
$$C_2'$$
, $A_4'C_2'$, ..., $(A_4')^{m-1}C_2'$] = m.

The boundary layer correction should be added to the outer expansion in order to obtain the uniformly valid solution. But it is not considered here, and shall be investigated in Section 3.2. The result of Section 3.2.3 can be easily applied to this problem. If we obtain the uniformly valid expansion, the similar theorem to Theorem 1-2 holds.

2.3 Fixed-end-point, minimum energy problem

This section deals with fixed-end-point, fixed-end-time, minimum energy problems, which lead to the so-called ill-conditioned two-point boundary value problems.

Here is introduced the generalized Riccati transformation, and the situations peculiar to the fixed-end-point problems are shown. For the free-end-point problem treated in Section 2.2, it is needed that the state matrix of the boundary layer system should be stable, but for the fixed-end-point problem in this section our analysis shows that the assumption required becomes that it is either positively or negatively stable.

2.3.1 Problem statement

The state equation having the same form as shown in Section 2.2, is as follows:

$$\frac{d}{dt} x(t,\varepsilon) = A(t,\varepsilon)x(t,\varepsilon) + B(t,\varepsilon)u(t,\varepsilon), \qquad (2.3.1)$$

$$A = \begin{bmatrix} A_1 & A_2 \\ \varepsilon^{-1}A_3 & \varepsilon^{-1}A_4 \end{bmatrix}, \quad B = \begin{bmatrix} B_1 \\ \varepsilon^{-1}B_2 \end{bmatrix}, \quad x = \begin{bmatrix} x_1 \\ x_2 \end{bmatrix}.$$

The prescribed boundary conditions are

$$x(t_0,\varepsilon) = \xi(\varepsilon), \quad x(t_f,\varepsilon) = \eta(\varepsilon),$$
 (2.3.2)

or in the partitioned form

$$\begin{bmatrix} x_1(t_0,\varepsilon) \\ x_2(t_0,\varepsilon) \end{bmatrix} = \begin{bmatrix} \xi_1(\varepsilon) \\ \xi_2(\varepsilon) \end{bmatrix}, \begin{bmatrix} x_1(t_f,\varepsilon) \\ x_2(t_f,\varepsilon) \end{bmatrix} = \begin{bmatrix} \eta_1(\varepsilon) \\ \eta_2(\varepsilon) \end{bmatrix}.$$

The problem is to minimize the performance index

$$J = \frac{1}{2} \int_{t_0}^{t_f} [x(t)'Qx(t) + u(t)'Ru(t)]dt. \qquad (2.3.3)$$

This problem, as easily seen, leads to the canonical equation

$$\begin{bmatrix} \mathbf{x} \\ \mathbf{p} \\ \mathbf{p} \end{bmatrix} = \begin{bmatrix} \mathbf{A} & -\mathbf{B}\mathbf{R}^{-1}\mathbf{B'} \\ -\mathbf{Q} & -\mathbf{A'} \end{bmatrix} \begin{bmatrix} \mathbf{x} \\ \mathbf{p} \end{bmatrix}, \qquad (2.3.4)$$

where p is a costate vector of x. The canonical system (2.3.4) under the boundary conditions (2.3.2) is ill-conditioned in the sense of Mufti et al. [Mf.69], i.e., in our case the boundary condition of the costate vector p is not specified.

Now we define a Riccati transformation [Mf.69]

$$x(t) = M(t)p(t) + d(t),$$
 (2.3.5)

where M(t) is an $(n + m) \times (n + m)$ -matrix and d(t) is an (n + m)-dimendional vector. Notice that (2.3.5) is different from the conventional Riccati transformation (2.2.9) in control theory. By simple manipulation, we obtain the differential equation of Riccati type for M(t) and the associated differential equation for d(t) as below:

$$\dot{M} = AM + MA' + MQM - BR^{-1}B,$$
 (2.3.6)

$$A = (A + MQ)d.$$
 (2.3.7)

Equations (2.3.6) and (2.3.7) are solved under the initial conditions

$$M(t_0) = 0,$$
 (2.3.8)

and

$$d(t_0) = \xi,$$
 (2.3.9)

or under the final conditions

$$M(t_f) = 0,$$
 (2.3.8')

$$d(t_f) = \eta.$$
 (2.3.9')

Partitioning M(t) and d(t) into the forms

$$M = \begin{bmatrix} M_1 & M_2 \\ M_2' & \varepsilon^{-1}M_3 \end{bmatrix}, \quad d = \begin{bmatrix} d_1 \\ d_2 \end{bmatrix}, \quad (2.3.10)$$

we get the equations for M_i , i = 1, 2, 3;

$$\begin{split} \dot{M}_{1} &= A_{1}M_{1} + M_{1}A_{1} + A_{2}M_{2}' + M_{2}A_{2}' + M_{1}Q_{1}M_{1} + M_{2}Q_{2}M_{1} + M_{1}Q_{2}M_{2} \\ &+ M_{2}Q_{2}M_{2}' - B_{1}R^{-1}B_{1}', \\ \epsilon \dot{M}_{2} &= M_{2}A_{4}' + M_{2}Q_{3}M_{3} + M_{1}A_{3} + M_{1}Q_{2}M_{3} - B_{1}R^{-1}B_{2}' \\ &+ \epsilon (A_{1}M_{2} + A_{2}M_{3} + M_{1}Q_{1}M_{2} + M_{2}Q_{2}M_{2}), \\ \epsilon \dot{M}_{3} &= A_{4}M_{3} + M_{3}A_{4}' + M_{3}Q_{3}M_{3} - B_{2}R^{-1}B_{2} + \epsilon (A_{3}M_{2} + M_{2}'A_{3}' \\ &+ M_{3}Q_{2}'M_{2} + M_{2}'Q_{2}M_{3}) + \epsilon^{2}M_{2}'Q_{1}M_{2}, \end{split}$$

and for d_{i} , i = 1, 2;

$$\dot{d}_{1} = (A_{1} + M_{1}Q_{1} + M_{2}Q_{2})d_{1} + (A_{2} + M_{1}Q_{2} + M_{2}Q_{2})d_{2},$$

$$\varepsilon \dot{d}_{2} = (A_{3} + M_{3}Q_{2}' + \varepsilon M_{2}'Q_{1})d_{1} + (A_{4} + M_{3}Q_{3} + \varepsilon M_{2}'Q_{2})d_{2},$$
(2.3.12)

where Q_i 's are the elements of

$$\mathbf{Q} = \begin{bmatrix} \mathbf{Q}_1 & \mathbf{Q}_2 \\ \mathbf{Q}_2' & \mathbf{Q}_3 \end{bmatrix}.$$

We call the system (2.3.11), (2.3.12) with (2.3.8), (2.3.9) the "forward full system" and the system (2.3.11), (2.3.12) with (2.3.8'), (2.3.9') the "backward full system".

After M and d are determined under the terminal conditions,

we get the optimal control u*

$$u^* = -R^{-1}B'M(t)^{-1}(x(t) - d(t)). \qquad (2.3.13)$$

The optimum trajectory is then given by

$$\dot{\mathbf{x}} = (\mathbf{A} - \mathbf{BR}^{-1}\mathbf{B'M}^{-1})\mathbf{x} + \mathbf{BR}^{-1}\mathbf{B'M}^{-1}\mathbf{d}, \qquad (2.3.14)$$

where it is assumed that M is invertible, so that Eqs.(2.3.13) and (2.3.14) hold for $t \in [t_0, t_f]$ for the forward system, or for $t \in [t_0, t_f]$ for the backward system.

2.3.2 Asymptotic expansions and approximate solutions

The reduced equations for M_i 's and d_i 's are obtained by letting $\varepsilon = 0$ in Eqs.(2.3.11) and (2.3.12),

$$\dot{M}_{1}^{0} = A_{1}M_{1}^{0} + M_{1}^{0}A_{1}' + A_{2}M_{2}^{0}' + M_{2}^{0}A_{2}' + M_{1}^{0}Q_{1}M_{1}^{0} + M_{2}^{0}Q_{2}M_{1}^{0} + M_{1}^{0}Q_{2}M_{2}^{0}' + M_{2}^{0}Q_{2}M_{2}^{0}' - B_{1}R^{-1}B_{1},$$

$$0 = M_{2}^{0}A_{4}' + M_{2}^{0}Q_{3}M_{3}^{0} + M_{1}^{0}A_{3} + M_{1}^{0}Q_{2}M_{3}^{0} - B_{1}R^{-1}B_{2}',$$

$$0 = A_{4}M_{3}^{0} + M_{3}^{0}A_{4}' + M_{3}^{0}Q_{3}M_{3}^{0} - B_{2}R^{-1}B_{2}',$$
(2.3.15)

and

$$\dot{d}_{1}^{0} = (A_{1} + M_{1}^{0}Q_{1} + M_{2}^{0}Q_{2})d_{1}^{0} + (A_{2} + M_{1}^{0}Q_{2} + M_{2}^{0}Q_{3})d_{2}^{0},$$

$$0 = (A_{3} + M_{3}^{0}Q_{2}')d_{1}^{0} + (A_{4} + M_{3}^{0}Q_{3})d_{2}^{0},$$
(2.3.16)

with the initial conditions for the forward reduced system

$$M_1^0(t_0) = 0,$$
 (2.3.17)

and

$$d_1^0(t_0) = \xi_1, \qquad (2.3.18)$$

or with the final conditions for the backward reduced system

$$M_1^0(t_0) = 0,$$
 (2.3.17')

and

$$d_1^0(t_0) = \eta_1.$$
 (2.3.18')

The superscript zero means that the corresponding variables are of the reduced system.

Now expanding M_i 's and d_i 's into Taylor series in ε , substituting them into Eqs.(2.3.11) and (2.3.12), and comparing the coefficients of like powers of ε thereof, we have the first correction system, the second, and so forth. The deduction is easy to derive as in the preceding section.

In this case, for the forward system the boundary layer system is

$$\frac{d}{d\tau} \overline{M}_{2}(\tau) = \overline{M}_{2}(\tau)A_{4}' + \overline{M}_{2}(\tau)Q_{3}\overline{M}_{2}(\tau) + M_{1}^{0}(t)A_{3} + M_{1}^{0}(t)Q_{2}\overline{M}_{3}(\tau),$$

$$\frac{d}{d\tau} \overline{M}_{3}(\tau) = A_{4}\overline{M}_{3}(\tau) + \overline{M}_{3}(\tau)A_{4}' + \overline{M}_{3}(\tau)Q_{3}\overline{M}_{3}(\tau) - B_{2}R^{-1}B_{2}',$$

$$\frac{d}{d\tau} \overline{d}_{2}(\tau) = (A_{3} + \overline{M}_{3}(\tau)Q_{2}')d_{1}^{0}(t) + (A_{4} + \overline{M}_{3}(\tau)Q_{3})\overline{d}_{2}(\tau),$$

$$(2.3.21)$$

where the independent variable is τ , and t is considered as a fixed parameter. This system can be obtained by the "left stretching transformation" $\tau = [(t-t_0)/\epsilon], \epsilon \in [0, \epsilon_0]$. The terminal conditions for these equations are

$$\bar{f}_{2}(\tau = 0) = 0,$$
 (2.3.22)

$$\overline{M}_{3}(\tau = 0) = 0,$$
 (2.3.23)

$$\overline{d}_2(\tau = 0) = \xi_2.$$
 (2.3.24)

For the backward system, we can derive easily the backward boundary layer system in place of Eqs.(2.3.19) - (2.3.21) by introducing

the "right stretching transformation" $\tau' = [(t_f - t)/\epsilon], \epsilon \in [0,\epsilon_0],$ with terminal conditions,

$$M_{2}(\tau' = 0) = 0,$$
 (2.3.22')

$$\overline{M}_{3}(\tau^{t}=0)=0,$$
 (2.3.23')

$$\overline{d}_2(\tau^* = 0) = n_2$$
. (2.3.24')

Then the terminal conditions for the first correction system are given as follows for the forward system:

$$M_{1}^{1}(t_{0}) = \int_{0}^{\infty} \left[(A_{2}\overline{M}_{2}'(\tau) + \overline{M}_{2}(\tau)A_{2}' + \overline{M}_{2}(\tau)Q_{2}M_{1} + M_{1}^{0}Q_{2}\overline{M}_{2}'(\tau)) - (A_{2}M_{2}^{0} + M_{2}^{0}A_{2}' + M_{2}^{0}Q_{2}M_{1}^{0} + M_{1}^{0}Q_{2}M_{2}^{0}') \right] d\tau, \qquad (2.3.25)$$

$$d_{1}^{1}(t_{0}) = \int_{0}^{\infty} \left[(\overline{M}_{2}(\tau) - M_{2}^{0})Q_{2}d_{1}^{0} + (A_{2} + M_{1}^{0}Q_{2})(\overline{d}_{2}(\tau) - d_{2}^{0}) + \overline{M}_{2}(\tau)Q_{3}\overline{d}_{2}(\tau) - M_{2}^{0}Q_{3}d_{2}^{0} \right] d\tau, \qquad (2.3.26)$$

where \overline{M}_2 , \overline{M}_3 , and \overline{d}_2 are solutions of the boundary layer system (2.3.19) - (2.3.21) with the boundary conditions (2.3.22) - (2.3.24). For the backward system, the variable τ in Eqs.(2.3.25) and (2.3.26) should be replaced by τ' , and instead of the initial conditions (2.3.22) - (2.3.24), we adopt the final conditions (2.3.22') -(2.3.24') for the backward boundary layer system, then obtaining $M_i^{1}(t_f)$ and $d_i^{1}(t_f)$.

A similar procedure is applied to solve the recursive equation for the higher order. As discussed later, whether the forward or backward system should be selected depends upon the property of the boundary layer system.

2.3.3 Suboptimum trajectory

Between the variables M, d in Eq.(2.3.5) and K, g in Eq.(2.2.9), there are the following relations:

$$K(t) = M(t)^{-1}$$
, (2.3.27)

and

$$g(t) = -M(t)^{-1}d(t)$$
. (2.3.28)

Expanding M(t)⁻¹ into Taylor series in ε , we have(see Appendix B)

$$K(t) = M(t)^{-1}$$

$$= \begin{bmatrix} M_{1}^{0} & 0 \\ 0 & 0 \end{bmatrix} + \begin{bmatrix} M_{1}^{1} & -1 + M_{1}^{-2}M_{2}M_{3}^{-1}M_{2}' & -M_{1}M_{2}M_{3}^{-1} \\ -M_{3}^{-1}M_{2}'M_{1}^{-1} & M_{3}^{-1} \end{bmatrix} + 0(\varepsilon^{2})$$

$$= K^{0} + \varepsilon K^{1} + 0(\varepsilon^{2})$$

$$= \begin{bmatrix} K_{0} & 0 \\ 0 & 0 \end{bmatrix} + \varepsilon \begin{bmatrix} K_{1} & K_{2} \\ K_{2}' & K_{3} \end{bmatrix} + 0(\varepsilon^{2}), \qquad (2.3.29)$$

where the superscript zero of M_{i} is omitted for simplicity.

The optimum trajectory is given by Eq.(2.3.14). Expanding each variable into Taylor series in ε and comparing coefficients of like powers of ε thereof, we have for the reduced system

with

$$x_1(t_0) = \xi_1$$
, or $x_1(t_f) = \eta_1$,

where

$$E_1 = B_1 R^{-1} B_1', \quad E_2 = B_1 R^{-1} B_2', \quad E_3 = B_2 R^{-1} B_2'.$$

For the first correction system

$$\begin{aligned} \mathbf{x}_{1}^{1} &= (\mathbf{A}_{1} - \mathbf{E}_{1}\mathbf{K}_{0})\mathbf{x}_{1}^{1} + \mathbf{A}_{2}\mathbf{x}_{2}^{1} + (\mathbf{E}_{1}\mathbf{K}_{0} + \mathbf{E}_{2}\mathbf{K}_{2}')\mathbf{d}_{1}^{1} + \mathbf{E}_{2}\mathbf{K}_{3}'\mathbf{d}_{2}^{1} \\ &- \mathbf{E}_{1}\mathbf{K}_{2}\mathbf{x}_{1}^{0} + (\mathbf{E}_{1}\mathbf{K}_{2}' + \mathbf{E}_{1}\mathbf{K}_{1})\mathbf{d}_{1}^{0}, \end{aligned}$$

$$\begin{aligned} \mathbf{0} &= (\mathbf{A}_{3} - \mathbf{E}_{2}\mathbf{K}_{0})\mathbf{x}_{1}^{1} + (\mathbf{A}_{4} - \mathbf{E}_{3}\mathbf{K}_{3})\mathbf{x}_{2}^{1} + \mathbf{E}_{2}\mathbf{K}_{0}\mathbf{d}_{1}^{1} + \mathbf{E}_{3}\mathbf{K}_{3}'\mathbf{d}_{2}^{1} - \mathbf{E}_{3}\mathbf{K}_{2}'\mathbf{x}_{1}^{0} \end{aligned}$$

$$- E_{2}'K_{2}x_{2}^{0} + (E_{3}K_{2}' + E_{2}K_{1})d_{1}^{0} + E_{2}'K_{2}d_{2}^{0}, \qquad (2.3.31b)$$

with the terminal condition for the forward system

$$x_1^1(t_0) = \int_0^\infty [A_2(\overline{x}_2(\tau) - x_2^0) + E_2K_2'(\overline{d}_2(\tau) - d_2^0)]d\tau$$
, (2.3.32)

where $\overline{x}_{2}(\tau)$ in the integrand is a solution of the boundary layer system

$$\frac{d}{d\tau} \overline{x}_{2}(\tau) = (A_{3} - E_{2}K_{0})x_{1}^{0} + (A_{4} - E_{3}K_{3})\overline{x}_{2}(\tau) + E_{2}'K_{0}d_{1}^{0} + E_{3}K_{2}'\overline{d}_{2}(\tau),$$
(2.3.33)

with

$$\overline{x}_2(\tau = 0) = \xi_2.$$
 (2.3.34)

For the backward system, similar conditions are easily obtained by the same procedure as in the preceding description. It is remarked that we need to solve the forward system for x_i 's when we solve the forward system for M_i 's and d_i 's and vice versa. This situation is different from the case using the conventional Riccati transformation (2.2.10).

After determining M 's, d 's, and x 's, the suboptimal control u_{sub} is given as follows:

$$u_{sub} = u^0 + \varepsilon u^1,$$
 (2.3.35)

where

$$u^{0} = -R^{-1}[(B_{1}'K_{0} + B_{2}'K_{2}')(x_{1}^{0} - d_{1}^{0}) + B_{2}'K_{3}(x_{2}^{0} - d_{2}^{0})],$$

$$u^{1} = -R^{-1}[(B_{1}'K_{0} + B_{2}'K_{2}')(x_{1}^{1} - d_{1}^{1}) + B_{2}'K_{3}(x_{2}^{1} - d_{2}^{1}) + B_{1}'K_{1}(x_{1}^{0} - d_{1}^{0}) + B_{1}'K_{2}(x_{2}^{0} - d_{2}^{0})].$$

2.3.4 A simple example

We shall show the outline of constructing the terminal values

of higher order system for

$$\frac{d}{dt} x_{1} = x_{1} - u,$$
(2.3.36)
$$\varepsilon \frac{d}{dt} x_{2} = x_{1} - x_{2} + u,$$

where x_1 , x_2 and u are scalars with $x_i(t_0) = \xi_i$ and $x_i(t_f) = 0$ prescribed. The performance index to be minimized is given as follows:

$$J = \frac{1}{2} \int_{t_0}^{t_f} (x'x + ru^2) dt. \qquad (2.3.37)$$

The Riccati transformation x(t) = M(t)p(t) + d(t) yields

$$\frac{d}{dt} m_1 = m_1^2 + 2m_2 + m_2^2 - r^{-1}, \qquad (2.3.38a)$$

$$\varepsilon \frac{d}{dt} m_2 = m_1 - m_2 + m_3 + m_2 m_3 - r^{-1} + \varepsilon m_1 m_2,$$
 (2.3.38b)

$$\epsilon \frac{d}{dt} m_3 = -2m_3 + m_3^2 - r^{-1} + 2\epsilon m_2 + \epsilon^2 m_2^2,$$
 (2.3.38c)

for each element of M = $\begin{bmatrix} m_1 & m_2 \\ m_2 & \varepsilon^{-1}m_3 \end{bmatrix}$, where m_1 , m_2 and m_3 are scalars.

The boundary layer system of Eq.(2.3.38) is

$$\frac{d}{d\tau} \,\overline{m}_2 = m_1(\tau_0) - \overline{m}_2 + \overline{m}_3 + \overline{m}_2 \overline{m}_3 - r^{-1}, \qquad (2.3.39a)$$

and

$$\frac{d}{d\tau} \bar{m}_3 = -2\bar{m}_3 + \bar{m}_3^2 - r^{-1}. \qquad (2.3.39b)$$

Now we have an asymptotically stable point as $\tau \rightarrow \infty$ for \overline{m}_3 ,

$$\overline{m}_{3}(\infty) = 1 - (1 + r^{-1})^{1/2} < 0,$$
 (2.3.40)

by equating the right hand side of Eq.(2.3.39b) to zero, where

$$\bar{m}_3 = 1 + (1 + r^{-1})^{1/2} > 0$$

is excluded because of Lemma 2-4 in Section 2.3.5. For \overline{m}_2 we get from Eqs.(2.3.39a) and (2.3.40)

$$\overline{m}_{2}(\infty) = -1 - (1 + r^{-1})^{1/2}$$
 (2.3.41)

The initial value of $m_1^1(t_0)$ is given from Eq.(2.3.25):

$$m_{1}(t_{0}) = \int_{0}^{\infty} [2\overline{m}_{2}(\tau) + \overline{m}_{2}^{2}(\tau) - 2m_{2}^{0}(t_{0}) - m_{2}^{02}(t_{0})]d\tau. \quad (2.3.42)$$

Eliminating $m_2(\tau)$ in Eq.(2.3.42) by using Eqs.(2.3.39a), (2.3.39b), (2.3.40) and (2.3.41), we get

$$m(t_{0}) = \left| -\frac{1}{2(1+r^{-1})^{1/2}} \overline{m}_{2}(\tau) - \frac{1+(1+r^{-1})^{1/2}}{1+r^{-1}} \overline{m}_{2}(\tau) \right|_{0}^{\infty}$$
$$= \frac{-2+(r^{-1}-2)(1+r^{-1})^{1/2}}{2(1+r^{-1})^{1/2}}, \qquad (2.3.43)$$

which can be obtained by a simple manipulation under the asymptotic stability of Eqs.(2.3.39a) and (2.3.39b), considering $\overline{m}_2(0) = 0$. Now we can compute the reduced system for $m_1^0(t)$ and the first correction system for $m_1^l(t)$, and the similar procedure can be applied to higher order terms, $m_1^j(t)$, for j > 1 and to the recursive equations for $d_1^j(t)$, and $x_1^j(t)$.

2.3.5 Basic theorems

Some conditions are needed to establish main theorems and lemmas. We first state the prerequisite conditions and then the basic theorems are offered.

As in the preceding section, the following conditions are usually assumed to hold for $t \in [t_0, t_f]$ and $\epsilon \in [0, \epsilon_0]$:

C1. A_i 's and B_i 's are holomorphic with respect to t and ε . C2. R and Q are positive definite and holomorphic with

respect to t and ϵ .

C3. ξ and η are continuous with respect to ϵ .

The conditions C1 - C3 are equivalent to C1 - C3 in Section 2.2.

We introduce an important condition in an analogous form to the observability, as follows:

C4. rank $[Q_3, A_4Q_3, A_4^2Q_3, \dots, A_4^{m-1}Q_3] = m.$

We add the following two conditions which are essential to the singular perturbation theory.

C5a. A_4 is a stable matrix. C5b. $-A_4$ is a stable matrix.

These conditions C5a. and C5b. are exclusive and main theorems need either one.

<u>Lemma 2-4</u> If conditions Cl - C5a (Cl - C4 and C5b) hold, then the forward (backward) boundary layer equation (2.3.20) has an isolated and asymptotically stable solution as $\tau \rightarrow \infty$ ($\tau' \rightarrow \infty$), which is negative definite (positive definite) and is given by equating the right hand side of Eq.(2.3.20) to zero. Moreover a matrix $A_4 + M_3Q_3$ (- $A_4 - M_3Q_3$) is stable.

Lemma 2-4 plays a decisive role in the sequel and directly leads to Lemma 2-5.

Lemma 2-5 If conditions Cl - C5a (Cl - C4 and C5b) hold, then the forward (backward) boundary layer equations (2.3.19) and (2.3.21) have respectively an isolated and asymptotically stable solution, which is given as in Lemma 2-4.

Lemmas 2-4 and 2-5 lead to the main theorems.

<u>Theorem 2-2</u> If conditions Cl - C5a (Cl - C4 and C5b) hold for the forward system (backward system), then the following convergence relations between the reduced solutions and the full solutions are satisfied,

Following Vasil'eva [Vs.63.1], we have Theorem 2-3 which gives error estimations.

<u>Theorem 2-3</u> If conditions C1 - C5a (C1 - C4 and C5b) hold, then exist bounded functions $U_i(t,\epsilon)$ and $V_i(t,\epsilon)$ such that

$$\begin{split} M_{1}(t) &= \sum_{j=0}^{k} M_{1}^{j}(t)(j!)^{-1} \varepsilon^{j} + U_{1}(t,\varepsilon)\varepsilon^{k+1}, \\ M_{i}(t) &= \sum_{j=0}^{k} M_{i}^{j}(t)(j!)^{-1}\varepsilon^{j} + U_{i}(t,\varepsilon)\varepsilon^{k+1}, \\ &; \text{ for } t\varepsilon [t_{0}+\delta,t_{f}](t\varepsilon [t_{0},t_{f}-\delta]); i = 2,3, \\ d_{1}(t) &= \sum_{j=0}^{k} d_{1}^{j}(t)(j!)^{-1}\varepsilon^{j} + V_{1}(t,\varepsilon)\varepsilon^{k+1}, \\ d_{2}(t) &= \sum_{j=0}^{k} d_{2}^{j}(t)(j!)^{-1}\varepsilon^{j} + V_{2}(t,\varepsilon)\varepsilon^{k+1}, \\ &; \text{ for } t\varepsilon [t_{0}+\delta,t_{f}](t\varepsilon [t_{0},t_{f}-\delta]), \end{split}$$

where

$$\delta = -C \epsilon \log \epsilon$$
; C is independent of ϵ .

Now we have similar theorems for x and u, and these are obtained by simple manipulation directly from Theorems 2-2 and 2-3, so that the detail is omitted. The proofs of these theorems and lemmas are shown directly or by simple modification from relevant theorems in Refs. [W.65, Vs.63.1, R63, R65, Lv.54]. Lemma 2-4 is proved from Theorems A-2 and A-3 in Reid [R.63, R65] (see Appendix A). Note that in the boundary layer system, t is regarded as a parameter and therefore the matrices in the matriceare constant, so that all assumptions of Theorem A in Reid are satisfied. Each Jacobian matrix of the right hand sides of the boundary layer systems (2.3.19) - (2.3.21) with respect to M_2 , M_3 and d_2 respectively is $A_4 + M_3Q_3$, hence the proof of Lemma 2-5 is deduced straight from Lemma 2-4.

Theorem 2-2 is a modified one of the popular theorem [Lv.54] in the two-point boundary value problem of singular perturbation type. In this regard, Tikhonov's convergence theorem [Tk.52] in the initial value problem should be referred to. Theorem 2-3 is an extended one of Theorem 1-3 of Chapter 1.

2.4 Systems with two small parameters

The singular perturbations of differential equations with several parameters were studied in a few papers by Hoppensteadt [Hp.69.3], O'Malley [0.71.2], Vasil'eva [Vs.63.2], etc. In Hoppensteadt, the regular degeneration was considered as for the many parameter case. O'Malley [0.71.2], and Vasil'eva [Vs.63.2] obtained asymptotic expansion of solutions as to the two parameter case.

But no studies can be found on the subject of singular perturbation approach to optimal control problems of the system involving several parameters. This section treats state regulator problems of systems containing two parameters of different orders of smallness. The results of two parameter cases can be easily extended to many parameter cases with slight modifications.

2.4.1 Problem statement

Let the state equation be given as below:

$$\frac{d}{dt} x(t,\varepsilon) = A(t,\varepsilon)x(t,\varepsilon) + B(t,\varepsilon)u(t,\varepsilon), \qquad (2.4.1)$$

$$A = \begin{bmatrix} A_{11} & A_{12} & A_{13} \\ \varepsilon_1^{-1}A_{21} & \varepsilon_1^{-1}A_{22} & \varepsilon_1^{-1}A_{23} \\ \varepsilon_2^{-1}A_{31} & \varepsilon_2^{-1}A_{32} & \varepsilon_2^{-1}A_{33} \end{bmatrix}; B = \begin{bmatrix} B_1 \\ \varepsilon_1^{-1}B_2 \\ \varepsilon_2^{-1}B_3 \end{bmatrix}, x = \begin{bmatrix} x_1 \\ x_2 \\ x_3 \end{bmatrix},$$

where \mathbf{x}_1 is an n-dimensional state vector, \mathbf{x}_2 m_1-dimensional, \mathbf{x}_3 m_2-dimensional, and

$$0 << \epsilon_2 << \epsilon_1 << 1,$$
 (2.4.2)

or

$$\lim_{\varepsilon \to 0} \varepsilon_2 / \varepsilon_1 = 0. \tag{2.4.3}$$

Here we consider the state regulator problem to minimize the performance index

$$J = \frac{1}{2} x'(t_f) Fx(t_f) + \frac{1}{2} \int_{t_0}^{t_f} [x'(t)Qx(t) + u'(t)Ru(t)]dt, (2.4.4)$$

where

$$\mathbf{F} = \begin{bmatrix} \mathbf{F}_{11} & \varepsilon_{1}\mathbf{F}_{12} & \varepsilon_{2}\mathbf{F}_{13} \\ \varepsilon_{1}\mathbf{F}_{12}^{'} & \varepsilon_{1}\mathbf{F}_{22} & \varepsilon_{2}\mathbf{F}_{23} \\ \varepsilon_{2}\mathbf{F}_{13}^{'} & \varepsilon_{2}\mathbf{F}_{23}^{'} & \varepsilon_{2}\mathbf{F}_{33} \end{bmatrix}, \quad \mathbf{Q} = \begin{bmatrix} \mathbf{Q}_{11} & \mathbf{Q}_{12} & \mathbf{Q}_{13} \\ \mathbf{Q}_{12}^{'} & \mathbf{Q}_{22} & \mathbf{Q}_{23} \\ \mathbf{Q}_{13}^{'} & \mathbf{Q}_{23}^{'} & \mathbf{Q}_{33} \end{bmatrix}.$$

The elements of a matrix F are expressed as in the above for the sake of convenience in our formulation. We assume $Q_{ij} = 0$ when i = j, for simplicity, though this is not essential. We can find feedback control u*, under the usual hypothesis of regulator problem:

$$u^* = -R^{-1}B'K(t)x(t),$$
 (2.4.5)

where K(t) satisfies the following matrix differential equation of

Riccati type:

$$\dot{K} = -KA - A'K + KBR^{-1}B'K - Q, K(t_f) = F,$$
 (2.4.6)

Now partitioning K into

$$K = \begin{bmatrix} K_{11} & \varepsilon_1 K_{12} & \varepsilon_2 K_{13} \\ \varepsilon_1 K'_{12} & \varepsilon_1 K_{22} & \varepsilon_2 K_{23} \\ \varepsilon_2 K'_{13} & \varepsilon_2 K'_{23} & \varepsilon_2 K_{33} \end{bmatrix},$$

we get the differential equations for each array

$$\frac{d}{dt} K_{11} = -K_{11}A_{11} - A_{11}K_{11} + K_{11}E_{11}K_{11} - Q_{11} + f_{1}(K_{ij}); (i,j)\neq(1,1), (2.4.7a)$$

$$\mu_{2} \frac{d}{dt} K_{12} = K_{12}(-A_{22} + E_{22}K_{22} + E_{23}K_{23}) + f_{2}(K_{ij}); (i,j)\neq(1,2), (2.4.7b)$$

$$\mu_{2} \frac{d}{dt} K_{22} = K_{22}(-A_{22} + E_{23}K_{23}) + (-A_{22} + K_{23}E_{23})K_{22} + K_{22}E_{22}K_{22} - Q_{22} - f_{3}(K_{23}), (2.4.7c)$$

$$\mu_{1}\mu_{2} \frac{d}{dt} K_{13} = K_{13}(-A_{33} + E_{33}K_{33}) + f_{4}(K_{ij}); (i,j)\neq(3,3), \quad (2.4.7d)$$

$$^{\mu}1^{\mu}2 \frac{d}{dt} K_{23} = K_{23}(-A_{33} + E_{33}K_{33}) + f_5(K_{22}, K_{33}),$$
 (2.4.7e)

$$^{\mu}1^{\mu}2 \frac{d}{dt} K_{33} = -K_{33}A_{33} - A_{33}K_{33} + K_{33}E_{33}K_{33} - Q_{33}, \qquad (2.4.7f)$$

where $E_{ij} = B_i R^{-1} B_j'$, $\mu_1 = \epsilon_2/\epsilon_1$, and $\mu_2 = \epsilon_1$, and f_i 's are polynomials of the first order of its argument with respect to the order of μ_i^0 . For details, Appendix C may be referred to. The terminal conditions for the full system are

$$K_{ij}(t_f) = F_{ij}; i, j = 1, 2, 3.$$
 (2.4.8)

2.4.2 Asymptotic expansions

We develop a method simpler than those presented by O'Malley

[0.71.2] or by Vasil'eva [Vs.63.2] in different ways. This method is an extended and modified one of Hoppensteadt's method [Hp.71.1] dealing with single parameter case.

For simplicity of notations, imbedding each element of K_{ij} into appropriate dimensional vectors u, v, and w, where u consists of elements of K_{13} , K_{23} , and K_{33} , we have

$$\frac{d}{dt} u = -f(u, v, w; \mu, t), \quad u(t_f) = u^0, \quad (2.4.9a)$$

$$^{\mu}2 \frac{d}{dt} v = -\phi(u, v, w; \mu, t), \quad v(t_{f}) = v^{0}, \quad (2.4.9b)$$

$$^{\mu}1^{\mu}2 \frac{d}{dt}w = -\psi(u, v, w; \mu, t), w(t_{f}) = w^{0},$$
 (2.4.9c)

where $\mu = (\mu_1, \mu_2)'$, and minus signs of the right hand side of Eq.(2.4.9) is for the convenience's sake in the arguments below. It is to be noted that the system (2.4.9) has a hierarchy of boundary layers, here doubly structured.

Now introducing new independent variables by

$$\tau_{1} = (t_{f} - t)/\mu_{1}\mu_{2}, \quad \tau_{2} = (t_{f} - t)/\mu_{2}, \quad (2.4.10)$$

$$\tau_{1} \in [0, \infty[, \quad \tau_{2} \in [0, \infty[; \text{ as } |\mu| \to 0,$$

we have stretched systems of two kinds, the fundamental boundary layer system

$$\frac{d}{d\tau_1} u = \mu_1 \mu_2 f(u, v, w; \mu, \mu_1 \mu_2 \tau_1), \qquad (2.4.11a)$$

$$\frac{d}{d\tau_{1}}v = \mu_{1}\phi(u, v, w; \mu, \mu_{1}\mu_{2}\tau_{1}), \qquad (2.4.11b)$$

$$\frac{d}{d\tau_{1}}w = \psi(u, v, w; \mu, \mu_{1}\mu_{2}\tau_{1}), \qquad (2.4.11c)$$

and the secondary boundary layer system

$$\frac{d}{d\tau_2} u = \mu_2 f(u, v, w; \mu, \mu_2 \tau_2), \qquad (2.4.12a)$$

$$\frac{d}{d\tau_2} v = \phi(u, v, w; \mu, \mu_2 \tau_2), \qquad (2.4.12b)$$

$$\mu_1 \frac{d}{d\tau_2} w = \psi(u, v, w; \mu, \mu_2 \tau_2). \qquad (2.4.12c)$$

We seek solutions of Eq.(2.4.9) of the form

 ∞

$$u = \sum_{r,s=0}^{n} [u_{rs}(t) + U_{rs}(\tau_1) + \overline{U}_{rs}(\tau_2)](r!s!)^{-1} \mu_1^{r} \mu_2^{s}, (2.4.13a)$$

$$\lim_{\tau_1 \to \infty} U_{rs}(\tau_1) = 0, \quad \lim_{\tau_2 \to \infty} \overline{U}_{rs}(\tau_2) = 0, \quad (2.4.14a)$$

and other variables v and w are also assumed to have the same form and referred to Eqs.(2.4.13b), (2.4.13c), (2.4.14b) and (2.4.14c).

In the outer region, $\Sigma u_{rs}(t)$, $\Sigma v_{rs}(t)$ and $\Sigma w_{rs}(t)$ satisfy Eq.(2.4.9), then we have a recursive system in the outer region by substituting the outer expansions into Eq.(2.4.9) and comparing the coefficent of μ_i^r , r = 0, 1, 2, ..., thereof,

$$\frac{d}{dt} u_{00} = -f(u_{00}, v_{00}, w_{00}; 0, t),$$

$$0 = -\phi(u_{00}, v_{00}, w_{00}; 0, t),$$

$$0 = -\psi(u_{00}, v_{00}, w_{00}; 0, t),$$
(2.4.15)₀₀

and

$$\frac{d}{dt} u_{rs} = -f_{u}u_{rs} - f_{v}v_{rs} - f_{w}w_{rs} + p_{rs}(t),$$

$$\frac{d}{dt} v_{r,s-1} = -\phi_{u}u_{rs} - \phi_{v}v_{rs} - \phi_{w}w_{rs} + q_{rs}(t), \qquad (2.4.15)_{rs}$$

$$\frac{d}{dt} w_{r-1,s-1} = -\psi_{u}u_{rs} - \psi_{v}v_{rs} - \psi_{w}w_{rs} + r_{rs}(t).$$

For the fundamental boundary layer system, we have the recursive set of equations by substituting Eq.(2.4.13) into Eq.(2.4.11) as follows:

$$\frac{\mathrm{d}}{\mathrm{d}\tau} \mathrm{U}_{00} = 0,$$

$$\frac{d}{d\tau_{1}} V_{00} = 0, \qquad (2.4.16)_{00}$$

$$\frac{d}{d\tau_{1}} W_{00} = \Psi(u_{00}(t_{f}) + U_{00} + \overline{U}_{00}(0), v_{00}(t_{f}) + V_{00} + \overline{V}_{00}(0), w_{00}(t_{f}) + W_{00} + \overline{W}_{00}(0); 0, 0),$$

and

$$\frac{d}{d\tau_{1}} U_{rs} = P_{rs}(\tau_{1}),$$

$$\frac{d}{d\tau_{1}} V_{r,s-1} = Q_{rs}(\tau_{1}),$$

$$\frac{d}{d\tau_{1}} W_{r-1,s-1} = \psi_{u} U_{rs} + \psi_{v} V_{rs} + \psi_{w} W_{rs} + R_{rs}(\tau_{1}).$$
(2.4.16)_{rs}

For the secondary boundary layer system, we obtain

and

$$\frac{d}{d\tau_{2}}\overline{\overline{U}}_{rs} = \overline{P}_{rs},$$

$$\frac{d}{d\tau_{2}}\overline{\overline{V}}_{rs} = \phi_{u}\overline{\overline{U}}_{rs} + \phi_{v}\overline{\overline{V}}_{rs} + \phi_{w}\overline{\overline{W}}_{rs} + \overline{Q}_{rs}(\tau_{2}), \qquad (2.4.17)_{rs}$$

$$\frac{d}{d\tau_{2}}\overline{\overline{W}}_{r,s-1} = \psi_{u}\overline{\overline{U}}_{rs} + \psi_{v}\overline{\overline{V}}_{rs} + \psi_{w}\overline{\overline{W}}_{rs} + \overline{R}_{rs}(\tau_{2}),$$

where the remainders p_{rs} , q_{rs} , r_{rs} , P_{rs} , Q_{rs} , R_{rs} , etc. are polynomials involving only the terms known successively in the preceding steps.

The terminal conditions for these systems can be derived in the initial conditions for the terms with subscript (r,s) = (1,0). From the secondary boundary layer system (2.4.12), we get

$$U_{10}(\tau_{1}) = U_{10}(0) + \int_{0}^{\tau_{1}} P_{10}(\sigma) d\sigma,$$

$$V_{10}(\tau_{1}) = V_{10}(0) + \int_{0}^{\tau_{1}} Q_{10}(\sigma) d\sigma.$$
(2.4.18)

The restriction on the boundary layer correction terms (2.4.14) makes us choose $U_{10}(0)$ and $V_{10}(0)$ as

$$U_{10}(0) = - \int_{0}^{\infty} P_{10}(\sigma) d\sigma, \qquad (2.4.19)$$

$$V_{10}(0) = -\int_{0}^{\infty} Q_{10}(\sigma) d\sigma.$$
 (2.4.20)

As for the fundamental boundary layer system (2.4.11), we can obtain

$$\overline{U}_{10}(0) = - \int_{0}^{\infty} \overline{P}_{10}(\sigma) d\sigma. \qquad (2.4.21)$$

The integrals (2.4.19) - (2.4.21) may exist if the conditions

$$\phi_{\mathbf{v}} \leq -\kappa < 0, \tag{2.4.22}$$

$$\Psi_{\rm W} \leq -\kappa < 0, \qquad (2.4.23)$$

hold. Then we can determine the terminal condition for the outer system of u_{10} by using Eqs.(2.4.13), (2.4.19) - (2.4.21),

$$u_{10}(t_f) = u_{10}^0 - U_{10}(0) - \overline{U}_{10}(0).$$
 (2.4.24)

Under the condition (2.4.24) we can solve $u_{10}(t)$, $v_{10}(t)$, and $w_{10}(t)$, and we get $v_{10}(t_f)$ and $w_{10}(t_f)$.

From Eq.(2.4.20), $V_{10}(0)$ is known then $\overline{V}_{10}(0)$ is derived as follows:

$$\overline{V}_{10}(0) = v_{10}^0 - v_{10}(t_f) - V_{10}(0).$$
 (2.4.25)

The secondary recursive system $(2.4.17)_{10}$ is solved with conditions (2.4.21) and (2.4.25). Thus we have $W_{10}(0)$, then $W_{10}(0)$ as follows:

$$W_{10}(0) = w_{10}^0 - w_{10}(t_f) - \overline{W}_{10}(0).$$
 (2.4.26)

Using Eq.(2.4.26), the fundamental recursive system (2.4.16)₁₀ is worked out. In this way, every system can be solved thoroughly.

A similar procedure may treat each higher order system satisfactorily. This method has the advantage that we need not solve any differential equation in order to get initial conditions (2.4.25) and (2.4.26).

2.4.3 Basic theorems

We first state prerequisite conditions for clarity, though they may be obvious from conditions stated in the preceding sections.

Cl. A_i 's, and B_i 's are holomorphic with respect to t and ε_i . C2. R and Q are positive definite and holomorphic with respect to t and ε_i .

C3. F is time invariant, positive semidefinite and continuous in $\boldsymbol{\varepsilon}_{1},$

The following conditions are proper to the several parameter case.

C4. The system (2.4.1) is fundamental boundary layer controllable, which is defined by

rank $[B_3, A_{33}B_3, \dots, A_{33}^m 2^{-1}B_3] = m_2,$

for fixed t ε [t₀, t_f].

C5. A₃₃ is a stable matrix.

C6. The system (2.4.1) is secondary boundary layer controllable, which is defined by

rank
$$[B_2, A_2B_2, \dots, A_2^m 1^{-1}B_2] = m_1,$$

for fixed t ε [t₀, t_f], where A₂ is given by

$$A_2 = A_{22} - E_{23}K_{23}.$$
 (2.4.27)

C7. A₂ is a stable matrix.

As discussed before, conditions C5 and C7 can be replaced by the following C5' and C7' respectively.

C5'. The system (2.4.1) is fundamental boundary layer observable, which is defined as,

rank $[C_3', A_{33}'C_3', \ldots, (A_{33}')^m 2^{-1}C_3'] = m_2$, for fixed t ε $[t_0, t_f]$, where C^{3} is a solution of

$$C_{3}C_{3} = Q_{33}$$
 (2.4.28)

C7'. The system (2.4.1) is secondary boundary layer observable, which is defined as,

rank
$$[C'_2, A'_2C'_2, \ldots, (A'_2)^m 1^{-1}C'_2] = m_1$$

for fixed t ϵ [t_0, t_f], where C2 is a solution of

$$C_2'C_2 = Q_2,$$
 (2.4.29)

where

$$Q_2 = Q_{22} + K_{23}A_{23} + A_{32}K_{23} - K_{23}E_{33}K_{23}$$
 (2.4.30)

Under these conditions, we now give main theorems.

<u>Theorem 2-4</u> If conditions C1 - C7 hold, then the following convergence relations between the reduced solutions and the full solutions are satisfied,

$$\lim_{\mu \to 0} K_{11}(t) = K_{11}^{00}(t), \text{ for } t \in [t_0, t_f],$$
(2.4.31)
$$\lim_{\mu \to 0} K_{ij}(t) = K_{ij}^{00}(t), \text{ for } t \in [t_0, t_f[; i, j=1, 2, 3, i\neq j. (2.4.32)]$$

As easily seen this theorem gives regular degeneration of the several parameter case. The succeeding theorem assures the asymptotic correctness. <u>Theorem 2-5</u> If conditions C1 - C7 hold, then there exist functions $S_{ij}(t,\mu)$ uniformly bounded in the interval considered, such that

$$K_{ij}(t) = \sum_{r+s \le N} [K_{ij}^{rs}(t) + M_{ij}^{rs}(\tau_{1}) + N_{ij}^{rs}(\tau_{2})](r!s!)^{-1} \mu_{1}^{rs} + \lambda^{N+1} S_{ij}(t,\mu), \qquad (2.4.33)$$

where $\lambda = \max(\mu_1, \mu_2)$, and $M_{ij}^{rs}(\tau_1)$ and $N_{ij}^{rs}(\tau_2)$ are respectively fundamental and secondary boundary layer correction terms corresponding to coefficients of the outer expansion $K_{ij}^{rs}(t)$. These matrices can be obtained through rearranging the elements of corresponding $U_{rs}(\tau_1), U_{rs}(\tau_2), V_{rs}(\tau_1)$, etc.

As in the single parameter case, if we restrict our attention to the outer expansion, it is easy to find the following theorem.

<u>Theorem 2-6</u> If conditions C1 - C7 hold, then there exist functions $S_{ii}(t,\mu)$ uniformly bounded, such that

$$K_{11}(t) = \sum_{\substack{r+s \leq N \\ ij}} K_{11}(t) (r!s!)^{-1} \mu_{1}^{r} \mu_{2}^{s} + \lambda^{N+1} S_{11}(t,\mu), \text{ for } t \in [t_{0}, t_{f}-\delta], (2.4.34)$$

$$K_{ij}(t) = \sum_{\substack{r+s \leq N \\ ij}} K_{ij}^{rs}(t) (r!s!)^{-1} \mu_{1}^{r} \mu_{2}^{s} + \lambda^{N+1} S_{ij}(t,\mu), i,j=1,2,3; i \neq 1, for \\ t \in [t_{0}, t_{f}-\delta], (2.4.35)$$

where

 $\delta = -C\lambda \log \lambda$; C is independent of λ .

Proofs of these theorems can be derived by inferences analogous to those of Section 2.2, and the outline is given here on the proof of Theorem 2-5. Theorem 2-4, and 2-6, will be proved as corollaries of Theorem 2-5.

Under the conditions C4 and C5, it is proved that the fundamental boundary layer equation for K_{33}

$$\frac{d}{d\tau_1} \overline{K}_{33} = \overline{K}_{33} A_{33} + A_{33}' \overline{K}_{33} - \overline{K}_{33} E_{33}' \overline{K}_{33} + Q_{33}$$
(2.4.36)

is asymptotically stable as $\tau_1 \rightarrow \infty$, which is equivalent to the system (2.2.29). Thus the following lemma is obtained in the same form as Lemma 2-1 of the single parameter systems.

<u>Lemma 2-6</u> If conditions C1 - C5 hold, then the fundamental boundary layer equation (2.4.36) has an isolated and asymptotically stable solution as $\tau_1 \rightarrow \infty$, which is positive definite and is given by equating the right hand side of Eq.(2.4.36) to zero. Furthermore, $(A_{33} - E_{33}K_{33})$ is a stable matrix.

This lemma as usual plays a crucial role in the singular perturbations of matrix Riccati equations, and is directly followed by Lemma 2-7.

<u>Lemma 2-7</u> If conditions Cl - C5 hold, then the fundamental boundary layer equations for K_{13} and K_{23} have respectively an isolated and asymptotically stable solution, which is derived as in Lemma 2-6.

The proof of this lemma is obtained easily, since each Jacobian matrix of the right hand side of the boundary layer equations for K_{13} and K_{23} with respect to K_{13} and K_{23} respectively is $A_{33} - E_{33}K_{33}$, and asymptotic behaviors of K_{13} and K_{23} are dominated by that of K_{33} .

The above argument is not seen peculiar to the several parameter case, if we refer to the results of Section 2.2. But the following discussions are characteristic of this case.

As for the secondary boundary layer system, we can have the following lemma.

<u>Lemma 2-8</u> If conditions C1 - C3, C6, and C7 hold, then the secondary boundary layer equation for K_{22}

$$\frac{d}{d\tau_2}K_{22} = K_{22}A_2 + A_2K_{22} - K_{22}E_{22}K_{22} + Q_2$$
(2.4.37)

has an isolated and asymptotically stable solution as $\tau_2 \rightarrow \infty$, which is positive definite and is given as in Lemma 2-6. Furthermore, the matrix $(A_{22} - E_{23}K_{23}' - E_{22}K_{22})$ is a stable matrix.

Lemma 2-8 leads to a lemma similar to Lemma 2-7, which assures the asymptotic stablity of the boundary layer equation for K_{12} , and is ommited in order to avoid repeating the same number over again.

Lemmas 2-6 - 2-8 show that every boundary layer equation associated with the original full system (2.4.7) has an isolated and asymptotically stable solution respectively, and therefore the conditions that assure the regular degeneration stated in Theorem 2-4 are all satisfied, the proof of which can be obtained by using a theorem of Hoppensteadt [Hp.69.3].

As for Theorem 2-5, we can prove it following O'Malley's result [0.71.2], with simple modifications and manipulations.

The results derived above are easily extended to the several parameters system of the form

$$\mu_{1}\mu_{2}\cdots\mu_{m}\frac{d}{dt}y_{m} = A_{m0}x + \sum_{i=1}^{m}A_{mi}y_{i} + B_{m}u,$$

where μ_1 , μ_2 , ..., μ_m are positive small parameters. It is to be noted that the system (2.4.38) has a hierarchy of m-fold structured boundary layers.

It can easily be seen that there is a close relation between the several parameter case and the time invariant case with $t_f = \infty$ of the single parameter case (see Appendix D).

2.5 Asymptotic behaviour of performance index

It is natural to investigate the asymptotic behaviour of the performance index, once we have obtained the asymptotic expansion of the control and the trajectory.

In this section, we are concerned with the performance index of the tracking problem treated in Section 2.2, since the problem involves the regulator problem as a special case. And it is easy to extend the result to the other optimization problems, with performance index of non-quadratic type.

Concerning the terminal free, state regulator problem, the following theorem was proved in case of excluding the terminal cost by Werner and Cruz [Wr.68].

<u>Theorem 2-7</u> It is assumed that the performance index J has a quadratic form of

$$J = \frac{1}{2} e'(t_f) Fe(t_f) + \frac{1}{2} \int_{t_0}^{t_f} [e'(t)Qe(t) + u'(t)Ru(t)]dt, (2.5.1)$$

Let the exact optimal control be denoted by $u_e(t)$ and the approximate solution of the control represented in the form of the truncated series up to the n-th term be denoted by $u_a^n(t)$ as

$$u_a^n(t) = \sum_{r=0}^n u^r \varepsilon^r (r!)^{-1}.$$
 (2.5.2)

Expanding the performance index J into Taylor series in

$$J(u) = \sum_{r=0}^{\infty} \frac{d^{r}}{d\varepsilon^{r}} J \bigg|_{\varepsilon=0}^{\varepsilon^{r}} (r!)^{-1}, \qquad (2.5.3)$$

we obtain the following approximate relations

$$\frac{d^{k}}{d\varepsilon^{k}} J(u_{e}(t)) \Big|_{\varepsilon=0} = \frac{d^{k}}{d\varepsilon^{k}} J(u_{a}^{n}(t)) \Big|_{\varepsilon=0} , \qquad (2.5.4)$$

for $k = 0, 1, 2, \ldots, 2n+1$.

The proof of this theorem was originally made for the case that the terminal cost is not included in the performance index, and the system is of the regular perturbation type by Werner and Cruz [Wr.68]. Sannuti and Kokotović mentioned first the theorem above as to the singular perturbations of the regulator problem [Sa.69.1].

It is easy to see that, if we obtain the approximate solutions temporally uniform, there is little difference between the singular perturbations and the regular perturbations. The proof can be made in line with Werner and Cruz, and is omitted here.

As to the case that the performance index is of nonquadratic type, as for instance

$$J = \int_{t_0}^{t_f} (|x| + r|u|) dt, \qquad (2.5.5)$$

we can easily obtain the following theorem instead of Theorem 2-7. If the problem is restricted to the fixed terminal systems, then the following theorem also holds.

Theorem 2-8 If we obtain the approximate solution u_a^n as given in Theorem 2-7, and expand the performance index into Taylor series in ε , then we have the following approximate relations

$$\frac{d^{k}}{d\varepsilon^{k}} J(u_{e}(t)) \Big|_{\varepsilon=0} = \frac{d^{k}}{d\varepsilon^{k}} J(u_{a}^{n}(t)) \Big|_{\varepsilon=0}, \qquad (2.5.6)$$

for $k = 0, 1, 2, \ldots, n$.

This theorem is intuitive and it may be proved when the performance index has no singularity with respect to its arguments. Such a condition holds usually in the ordinary optimization problems.

It is to be noted that if we restrict our attention to the outer expansion, which is rational from the practical motivation, only the

slight changes are needed in the theorems described above as to the interval in which the validity of the asymptotic accuracy holds. The effect of the boundary layer upon the performance index is the same order of ε width of the boundary layer, unless the size of the boundary layer jump is the same order of ε^{-1} .

O'Malley [0.72.1] also stated the subject in the same form as Theorem 2-8 in case of the state regulator problem with quadratic cost.

A remarkable result is Theorem 2-7 particularly when the quadratic cost function is considered. The extension to the several parameter case is easy to be handled.

Chapter 3 Nuclear Reactor Control with Lumped Parameters

In this chapter, the nuclear reactor control as a lumped parameter system is considered. Such a lumped parameter model is well known in the field of reactor engineering as a "point reactor" or "one point approximation". Small reactors can be treated by the one point approximation with relatively small errors, if we restrict ourself to study the case in which the neutron flux changes slowly. Studies of dynamics and control of a point reactor have been made by many authors (see for instance Weaver[We.68], or Mohler and Shen [Mh. 70]).

The results in the previous chapters are successfully applied to the analysis and syntheses of suboptimal control of the nuclear reactor as a lumped parameter system.

This chapter consists of two sections. The early section is concerned with the error estimation of prompt jump approximation with the aid of singular perturbation theory. The prompt jump approximation is intuitively done by engineers and the singular perturbation theory can establish the validity of the approximation. The latter section treats the regulator problem of the reactor. Both sections have some numerical examples which enable us to understand the importance and validity of the theory presented.

3.1 Error estimation of prompt jump approximation

3.1.1 Introduction The second successive and the second se

In reactor dynamics, many ideas have been developed to give an error estimation of prompt jump approximation and to improve its accuracy. Goldstein and Shotkin [Gs.69] proposed the method of using integration by parts after transforming a system equation into an integral equation of Volterra type. Their method is effective in the outer region; but is not successful in giving uniformly valid solu-

tions in the interval considered. The singular perturbation theory, an outline of which is described in Chapter 1, has been applied to this aim with satisfactory results.

In this section, we obtain uniformly valid solutions with use made of Vasil'eva's theory. The basic idea of the theory is different from that of boundary layer method given in Section 1.4. But the resulting series are equivalent to each other. And the singular perturbation theory has a principal innovation in the treatment of initial conditions of the generated recursive equations, quite defferent from that of the classical perturbation theory, as is mentioned on occasions.

Our object in this section is to give a refined error estimation of prompt jump approximation providing a method of improving its accuracy in preparation for the synthesis of suboptimal control of a point reactor system.

3.1.2 Prompt jump approximation

In the development of the point reactor model, spatial independence and one energy grouping are assumed. The kinetic equation of the model without source term and temperature - dependent reactivity feedback is given by

$$\frac{d}{dt} n = \frac{\delta k - \beta}{\ell} n + \lambda C, \qquad (3.1.1)$$

$$\frac{\mathrm{d}}{\mathrm{d}t} C = \frac{\beta}{\ell} n - \lambda C, \qquad (3.1.2)$$

where n(t) and C(t) are respectively neutron density and precursor concentration. Normalization of n(t) and C(t) with respect to their equilibrium values n_0 , C_0 makes the system (3.1.1), (3.1.2) into the form of

$$\frac{\ell}{\beta} \frac{d}{dt} n(t) = (\rho(t) - 1)n(t) + C(t), \qquad (3.1.3)$$

$$\frac{d}{dt} C(t) = \lambda n(t) - \lambda C(t), \qquad (3.1.4).$$

where $\rho(t)$ is the reactivity in dollar units, i.e.,

$$\rho(t) \neq k(t)/\beta.$$
 (3.1.5)

The initial condition is given by

$$n(0) = 1, C(0) = 1.$$
 (3.1.6)

The derivation of the above model can be found in many of the texts, for example in Weaver [Wv.68]

If the relative rate of the change of the reactor power in the mean prompt generation time is sufficiently small so that

$$\left|\frac{\pounds}{\beta}\frac{\dot{n}}{n}\right| << |1 - \rho(t)| \qquad (3.1.7)$$

holds, then the term $(l/\beta)n(t)$ in the kinetic equation (3.1.3), (3.1.4) can be neglected as compared to $(\rho(t) - 1)n(t)$. The resulting approximation is written as follows:

$$0 = (\rho(t) - 1)n(t) + C(t),$$

$$\frac{d}{dt} C = \lambda n(t) - \lambda C(t).$$
(3.1.8)

This approximation is called the prompt jump approximation. Figure 3. 1.1 is a sketch of the exact kinetic equation and the prompt jump approximation to it in case of step reactivity insertion where the prompt jump approximation is shown by the dashed line.



Fig. 3.1.1 Prompt jump approximation and the exact solution. It is clear that the prompt jump approximation fails near promptcritical, as seen in the condition (3.1.7), in which the right hand side of Eq.(3.1.7), $1 - \rho(t) = 0$. The condition that gives the validity of the prompt jump approximation is given in various forms (see for instance Akcasu, Lellouche, and Shotkin [Ac.71]).

3.1.3 Analysis via singular perturbation theory

In Eq.(3.1.1), if we set $\varepsilon = \ell/\beta$, then ε has a relatively small value, for example $\varepsilon = 0(10^{-2})$ for the thermal reactor and $\varepsilon = 0(10^{-5})$ for the fast reactor, therefore equations (3.1.1) and (3.1.2) are of singular perturbation type.

Let solutions in the outer region be in the form of asymptotic expansions

$$n(t,\varepsilon) = \sum_{\substack{r=0\\ \infty}}^{\infty} n_r(t)\varepsilon^r(r!)^{-1},$$

$$C(t,\varepsilon) = \sum_{\substack{r=0\\ r=0}}^{\infty} C_r(t)\varepsilon^r(r!)^{-1}.$$
(3.1.9)

Substituting the above into Eqs.(3.1.1) and (3.1.2) and comparing coefficients of like powers of ε thereof, we obtain a recursion set of differential equations

$$\epsilon^{0}: \qquad 0 = (\rho(t) - 1)n_{0}(t) + C_{0}(t), \qquad (3.1.10)$$

$$\dot{C}_{0}(t) = \lambda(n_{0}(t) - C_{0}(t)), \qquad (3.1.10)$$

$$\epsilon^{r}: \qquad 0 = (\rho(t) - 1)n_{r}(t) + C_{r}(t) - \dot{n}_{r-1}(t), \qquad (3.1.11)$$

$$\dot{C}_{r}(t) = \lambda(n_{r}(t) - C_{r}(t)). \qquad (3.1.11)$$

Equation (3.1.8) for the leading terms $n_0(t)$ and $C_0(t)$ is equivalent to the prompt jump approximation (PJA).

The classical perturbation theory cannot be applied to this system, which may allow initial conditions of the form
$$C_0(0) = C(0), \quad C_r(0) = 0; \quad r > 0.$$
 (3.1.12)

and leads to a wrong result. Therefore we must treat initial conditions by the singular perturbation theory.

Adopting the current procedure, we first obtain solutions in the inner region (the boundary layer) and the outer region, and then we use the singular perturbation theory by Vasil'eva to connect these solutions [Vs.63.1].

In order to obtain a solution in the boundary layer, we consider a stretched system based on the stretching transformation($\tau = t/\epsilon$),

$$\frac{d}{d\tau} \overline{n}(\tau) = (\rho(\tau) - 1)\overline{n}(\tau) + \overline{C}(\tau),$$

$$\frac{d}{d\tau} \overline{C}(\tau) = \varepsilon \lambda(\overline{n}(\tau) - \overline{C}(\tau)),$$
(3.1.13)

which admits given initial conditions (3.1.6). The solutions of this problem have convergent expansions in powers of ε ,

$$\overline{\mathbf{n}}(\tau,\varepsilon) = \sum_{\substack{\mathbf{r}=0\\ \infty}}^{\infty} \overline{\mathbf{n}}_{\mathbf{r}}(\tau)\varepsilon^{\mathbf{r}}(\mathbf{r}!)^{-1},$$

$$\overline{\mathbf{C}}(\tau,\varepsilon) = \sum_{\mathbf{r}=0}^{\infty} \overline{\mathbf{C}}_{\mathbf{r}}(\tau)\varepsilon^{\mathbf{r}}(\mathbf{r}!)^{-1}.$$
(3.1.14)

As the stretched system (3.1.13) has no singularity at $\varepsilon = 0$, through the usual procedure of the regular perturbation method, we have a recursive set of differential equations

$$\varepsilon^{0}: \qquad \frac{\mathrm{d}}{\mathrm{d}\tau} \overline{n}_{0}(\tau) = (\rho(\tau) - 1)\overline{n}_{0}(\tau) + \overline{C}_{0}(\tau),$$

$$\frac{\mathrm{d}}{\mathrm{d}\tau} \overline{C}_{0}(\tau) = 0,$$
(3.1.15)

$$\epsilon^{\mathbf{r}}: \qquad \frac{d}{d\tau} \,\overline{\mathbf{n}}_{\mathbf{r}}(\tau) = (\rho(\tau) - 1)\overline{\mathbf{n}}_{\mathbf{r}}(\tau) + \overline{C}_{\mathbf{r}}(\tau),$$

$$\frac{d}{d\tau} \,\overline{C}_{\mathbf{r}}(\tau) = \lambda(\overline{\mathbf{n}}_{\mathbf{r}-1}(\tau) - \overline{C}_{\mathbf{r}-1}(\tau)),$$
(3.1.16)

with initial conditions

$$\overline{n}_{0}(0) = n(0), \quad \overline{C}_{0}(0) = C(0),$$

$$\overline{n}_{r}(0) = 0, \quad \overline{C}_{r}(0) = 0; \quad r > 0.$$
(3.1.17)

While many authors have proposed various methods in order to relate the two types of expansions, (3.1.9) and (3.1.14), we expand the coefficients $n_r(t)$ and $C_r(t)$ of series (3.1.9) into formal Taylor series involving τ and ε ; possibly divergent,

$$n_{r}(t) = \sum_{s=0}^{\infty} n_{rs} t^{s} = \sum_{s=0}^{\infty} n_{rs} \tau^{s} \varepsilon^{s},$$

$$C_{r}(t) = \sum_{s=0}^{\infty} C_{rs} t^{s} = \sum_{s=0}^{\infty} C_{rs} \tau^{s} \varepsilon^{s}.$$
(3.1.18)

Substituting Eq.(3.1.18) into Eq.(3.1.9) and rearranging the summations according to powers of ε , we have the coupling series in the form

$$n(t,\varepsilon) = \sum_{r=0}^{\infty} \hat{n}_{r}(\tau)\varepsilon^{r}, \quad \hat{n}_{r}(\tau) = \sum_{s=0}^{r} n_{r-s,s}\tau^{s},$$

$$C(t,\varepsilon) = \sum_{r=0}^{\infty} \hat{C}_{r}(\tau)\varepsilon^{r}, \quad \hat{C}_{r}(\tau) = \sum_{s=0}^{r} C_{r-s,s}\tau^{s}.$$
(3.1.19)

The series (3.1.19) are also formal solutions of the stretched system (3.1.13), but the difference between these two series lies in the initial values at $\tau = 0$.

Thus $\hat{n}_{_{\bf r}}(\tau)$ and $\hat{C}_{_{\bf r}}(\tau)$ satisfy the differential equations in the form

$$\hat{n}_{0}(\tau) = \frac{1}{1 - \rho(\tau)} C_{0}(0), \qquad (3.1.20)$$

$$\hat{c}_{0}(\tau) = C(0), \qquad (3.1.20)$$

and

$$\frac{d}{d\tau} \hat{n}_{r}(\tau) = (\rho(\tau) - 1)\hat{n}_{r}(\tau) + \hat{c}_{r}(\tau),$$

$$\frac{d}{d\tau} \hat{c}_{r}(\tau) = \lambda(\hat{n}_{r-1}(\tau) - \hat{c}_{r-1}(\tau)).$$
(3.1.21)

Several papers on the related subjects have been published [Am.69, Am.71], which have been based on the method of matched asymptotic expansions [Ku.67, Vd.64, Cl.68], but one of these [Am.71] leaves some doubt on the author's choice of initial conditions of the outer region.*

Here, we make use of Vasil'eva's method to determine initial conditions of outer region equations in such a manner that solutions of inner region equations (3.1.15) and (3.1.16) and those of Eqs.(3.1.20) and (3.1.21) coincide at $\tau = \infty$. Thus we have the initial conditions of outer region equations (3.1.10) and (3.1.11) and those of Eqs.(3.1.20) and (3.1.21) as follows:

$$C_{0}(0) = C_{0}(0) = C(0), \qquad (3.1.22)$$

$$C_{r}(0) = \hat{C}_{r}(0) = \int_{0}^{\infty} [\lambda \overline{n}_{r-1}(\sigma) - \lambda n_{r-1}(0)] d\sigma; r > 0,$$

where $n_{r-1}(\sigma)$ is a solution of the boundary layer equation belonging to the full system (3.1.1) and (3.1.2):

$$\frac{d}{d\sigma} \bar{n}_{r-1}(\sigma) = (\rho(0) - 1)\bar{n}_{r-1}(\sigma) + \bar{c}_{r-1}(0), \qquad (3.1.23)$$

with the initial conditions

$$\bar{n}_0(0) = n(0), \ \bar{n}_r(0) = 0; \ r > 0.$$
 (3.1.24)

* The method adopts the initial conditions $n_0(0) = n(0)$, $n_1 = 0$. But it is not logical in the light of the theory of matched asymptotic expansion, which requires the limit process to decide initial values for outer expansion.

It is to be noted that in Eq.(3.1.23) \overline{C}_{r-1} is regarded as a fixed paremeter, while in Eq.(3.1.15) \overline{C}_{r-1} is a variable.

After determining successively n_r , \overline{n}_r , \hat{n}_r , C_r , \overline{C}_r , and \hat{C}_r , we define the functions $R(t,\varepsilon)$ and $S(t,\varepsilon)$ by

$$n(t,\varepsilon) = \sum_{r=0}^{m} [n_r(t) + \overline{n_r}(\tau) - \hat{n_r}(\tau)]\varepsilon^r(r!)^{-1} + R(t,\varepsilon)\varepsilon^{m+1}, \quad (3.1.25)$$

$$C(t,\varepsilon) = \sum_{r=0}^{m} [C_r(t) + \overline{C}_r(\tau) - \hat{C}_r(\tau)]\varepsilon^r(r!)^{-1} + S(t,\varepsilon)\varepsilon^{m+1}, \quad (3.1.26)$$

where $\tau = t/\epsilon$. Assuming that $\rho(t) - 1 < 0$, the functions $R(t,\epsilon)$ and $S(t,\epsilon)$ are bounded for $0 \le t \le \infty$; $0 \le \epsilon << 1$. This means that an error of truncated series up to the (m + 1)-th term is of the order of ϵ^{m+1} .

3.1.4 Numerical examples

For simplicity we restrict ourselves to a step input of reactivity $(\rho(t) = \rho_0, time independent).$

In Fig.3.1.2 we can see PJA has a boundary layer at t = 0 and SPA has uniform accuracy with an error of the order of ϵ^2 as shown in Eq.(3.1.25). Figure 3.1.2 also shows that EP is less than expected in some time interval, and outside it, SP well Approximates EP. Our study by numerical calculations shows that, as far as step insertion of reactivity is concerned, SPA has a good accuracy when $\rho < 0.5$ \$. In Fig.3.1.2 we used constants $\beta = 0.0064$, $\lambda = 0.078$.



The method developed here can treat arbitrary reactivity inputs so far as the asymptotic stability of the system for the neutron density n is preserved, by a simple modification and has close relation to the matched asymptotic expansion method proposed by Kaplun et al. independently of the method by Vasil'eva. Namely Eq.(3.1.9) corresponds to the "outer expansion" of the matched asymptotic expansions and Eq.(3.1.14) to the "inner expansion".

This method permits us to deal with more complex systems such as expressed by kinetic equations with multi-group of delayed neutrons, and furthermore with nonlinear systems.

3.2 Regulator problem of a point reactor

This section deals with the state regulator problem of a point reactor, which occurs when deviation from the equilibrium occurs. The same formulation may treat the power level change control in the sense of tracking problems.

3.2.1 Formulation of the problem

We consider here the same reactor model discussed in Section 3.1. Linearizing the equations (3.1.1) and (3.1.2) by considering small perturbations about some equilibrium neutron and precursor concentration level n_0 and C_0 , we obtain

$$\frac{d}{dt} \delta n(t) = -\frac{\beta}{\ell} \delta n(t) + \lambda \delta C(t) + \frac{n_0}{\ell} \delta k(t), \qquad (3.2.1)$$

$$\frac{d}{dt} \delta C(t) = \frac{\beta}{\ell} \delta n(t) - \lambda \delta C(t), \qquad (3.2.2)$$

where δn , δC , and δk are small deviations in n, C, and k respectively.

Normalizing quantities $\delta n(t)$ and $\delta C(t)$ with respect to each steady state values n_0 , C_0 respectively, the reactivity being represented in dollar units, we obtain the state equation to begin with

$$\frac{\&}{\beta}\frac{d}{dt}\tilde{n}(t) = -\tilde{n}(t) + \tilde{C}(t) + \rho(t), \qquad (3.2.3)$$

$$\frac{\mathrm{d}}{\mathrm{d}t}\tilde{\mathrm{C}}(t) = \lambda \tilde{\mathrm{n}}(t) - \lambda \tilde{\mathrm{C}}(t), \qquad (3.2.4)$$

where

$$\tilde{n}(t) = \frac{\delta n}{n_0}(t), \quad \tilde{C}(t) = \frac{\delta C}{C_0}(t), \quad (3.2.5)$$

and $\rho(t)$ is the externally applied reactivity. When the system deviates from the equilibrium, we meet the problem of regulating the deviation of the power level, or neutron density in a fixed time so that the performance index is a minimum, which is given here as below:

$$J = \frac{1}{2} \int_{t_0}^{t_f} (\tilde{n}^2 + r\rho^2) dt, \qquad (3.2.6)$$

where r > 0 is a scalar weight. Therefore we assume that the value of the deviation can be measured at the initial time t_0 , and should be represented in the form of initial conditions, as follows:

$$\tilde{n}(t_0) = \tilde{n}_0, \quad \tilde{C}(t_0) = \tilde{C}_0.$$
 (3.2.7)

Further we assume for simplicity the end-terminal is free and the control may be obtained without constraints.

Letting $\tilde{C} = x_1$, $\tilde{n} = x_2$, $\rho = u$ and $\lambda = a$ for the sake of convenience, Eqs.(3.2.3) and (3.2.4) become

$$\frac{d}{dt} x_1 = -ax_1 + ax_2, \qquad (3.2.8)$$

$$\varepsilon \frac{d}{dt} x_2 = x_1 - x_2 + u,$$
 (3.2.9)

with initial conditions

$$x_1(t_0) = \xi_1, \quad x_2(t_0) = \xi_2.$$
 (3.2.10)

Defining the Hamiltonian function H by

$$H = -\frac{1}{2}(x_1^2 + ru^2) + p_1(-ax_1 + ax_2) + p_2(x_1 - x_2 + u), \quad (3.2.11)$$

where p_1 and p_2 are costate variables of x_1 and x_2 , we derive the canonical equation

$$x_1 = -ax_1 + ax_2,$$
 (3.2.12a)

$$\varepsilon \dot{x}_2 = x_1 - x_2 + r^{-1}p_2,$$
 (3.2.12b)

$$\dot{p}_1 = ap_1 + p_2,$$
 (3.2.12c)

$$\varepsilon \dot{p}_2 = x_2 - ap_1 + p_2.$$
 (3.2.12d)

The optimal control u* is given by

$$u^* = r^{-1}p_2.$$
 (3.2.13)

The terminal conditions at both ends of the interval $[t_0, t_f]$ can be derived by considering condition (3.2.10) and the transversality condition as follows:

$$x_1(t_0) = \xi_1, \quad x_2(t_0) = \xi_2,$$
 (3.2.14)

and

$$p_1(t_f) = 0, \quad p_2(t_f) = 0.$$
 (3.2.15)

3.2.2 Asymptotic expansions

Expanding each variable into Taylor series in ε , and substituting the resulting series into Eqs.(3.2.12) and (3.2.13), and comparing coefficients of like powers of ε , we obtain the recursive set of equations, i.e., the outer recursive system,

$$\dot{\mathbf{p}}_{1}^{0} = a\mathbf{p}_{1}^{0} - \mathbf{p}_{2}^{0}, \qquad (3.2.16)$$
$$0 = \mathbf{x}_{2}^{0} - a\mathbf{p}_{1}^{0} - \mathbf{p}_{2}^{0},$$

and

$$\dot{x}_{1}^{k} = -ax_{1}^{k} + ax_{2}^{k},$$

$$0 = x_{1}^{k} - x_{2}^{k} + r^{-1}p_{2}^{k} - \dot{x}_{2}^{k-1},$$

$$\dot{p}_{1}^{k} = ap_{1}^{k} - p_{2}^{k},$$

$$0 = x_{2}^{k} - ap_{1}^{k} - p_{2}^{k} - \dot{p}_{2}^{k-1},$$
(3.2.17)

for k > 0.

Following the results obtained in Section 2.2, we can treat the two point boundary value problem (3.2.12) with (3.2.14) and (3.2.15) with use made of the Riccati transformation. The resulting matrix Riccati differential equation is

$$\dot{\mathbf{k}}_{1} = 2\mathbf{a}\mathbf{k}_{1} - 2\mathbf{k}_{2} + \mathbf{r}^{-1}\mathbf{k}_{2}^{2},$$

$$\varepsilon \ \dot{\mathbf{k}}_{2} = -\mathbf{a}\mathbf{k}_{1} + \mathbf{k}_{2} - \mathbf{k}_{3} + \mathbf{r}^{-1}\mathbf{k}_{2}\mathbf{k}_{3} + \varepsilon \mathbf{a}\mathbf{k}_{2},$$
 (3.2.18)

$$\varepsilon \ \dot{\mathbf{k}}_{3} = 2\mathbf{k}_{3} + \mathbf{r}^{-1}\mathbf{k}_{3}^{2} - 1 - 2\varepsilon \mathbf{a}\mathbf{k}_{2},$$

with the final terminal conditions

$$k_i(t_f) = 0; i = 1, 2, 3.$$
 (3.2.19)

Then we have the reduced system as follows:

$$\dot{k}_{1}^{0} = 2ak_{1}^{0} - 2k_{2}^{0} + r^{-1}k_{2}^{02} ,$$

$$0 = -ak_{1}^{0} + k_{2}^{0} - k_{3}^{0} + r^{-1}k_{2}^{0}k_{3}^{0}, \qquad (3.2.20)$$

$$0 = 2k_{3}^{0} + r^{-1}k_{3}^{02} - 1,$$

with the terminal condition

$$k_1(t_f) = 0,$$
 (3.2.21)

and the first correction system in the outer region

$$\dot{k}_{1}^{1} = 2ak_{1}^{1} - 2k_{2}^{1} + 2r^{-1}k_{2}^{0}k_{2}^{1},$$

$$\dot{k}_{2}^{0} = -ak_{1}^{1} + k_{2}^{1} - k_{3}^{1} + r^{-1}(k_{2}^{0}k_{3}^{1} + k_{2}^{1}k_{3}^{0}) + ak_{2}^{0},$$

$$\dot{k}_{3}^{0} = 2k_{3}^{1} + r^{-1}2k_{3}^{0}k_{3}^{1} - 2ak_{2}^{0}.$$
(3.2.22)

The final terminal condition for the first correction system is given as follows:

$$k_1^{1}(t_f) = \int_0^\infty \left[(-2\overline{k}_2 + r^{-1}\overline{k}_2^2) - (-2k_2^0 + r^{-1}k_2^{02}) \right] d\tau,$$
 (3.2.23)

where \overline{k}_2 in the integrand is a solution of the boundary layer system

$$-\frac{\mathrm{d}}{\mathrm{d}\tau} \overline{k}_{2}(\tau) = -\mathrm{ak}_{1}^{0}(t_{\mathrm{f}}) + \overline{k}_{2}(\tau) - \overline{k}_{3}(\tau) + \mathrm{r}^{-1}\overline{k}_{2}(\tau)\overline{k}_{3}(\tau), \quad (3.2.24)$$

$$-\frac{d}{d\tau} \overline{k}_{3}(\tau) = 2\overline{k}_{3}(\tau) + r^{-1}\overline{k}_{3}^{2}(\tau) - 1, \qquad (3.2.25)$$

with

$$\overline{k}_2(0) = 0, \quad \overline{k}_3(0) = 0.$$
 (3.2.26)

After solving the Riccati equations (3.2.20) and (3.2.21), we have a suboptimum trajectory, as follows: for the reduced trajectory

with

$$x_1(t_0) = \xi_1.$$
 (3.2.28)

For the first correction trajectory

$$\dot{\mathbf{x}}_{1}^{1} = - a\mathbf{x}_{1}^{1} + a\mathbf{x}_{2}^{1},$$

$$\dot{\mathbf{x}}_{2}^{0} = \mathbf{x}_{1}^{1} - \mathbf{x}_{2}^{1} + \mathbf{r}^{-1}(\mathbf{k}_{2}^{0}\mathbf{x}_{1}^{1} + \mathbf{k}_{3}^{0}\mathbf{x}_{2}^{1} + \mathbf{k}_{2}^{1}\mathbf{x}_{1}^{0} + \mathbf{k}_{3}^{1}\mathbf{x}_{2}^{0}),$$
(3.2.29)

with the initial condition

$$x_1^1(t_0) = \int_0^\infty a[\overline{x}_2(\tau) - x_2^0(t_0)]d\tau,$$
 (3.2.30)

where $x_2(\tau)$ is a solution of the following boundary layer system associated with the trajectory equation (3.2.12):

$$\frac{d}{d\tau} \overline{x}_{2}(\tau) = x_{1}^{0}(t_{0}) - \overline{x}_{2}(\tau) + r^{-1}(k_{2}^{0}(t_{0})x_{1}(t_{0}) + k_{3}^{0}(t_{0})\overline{x}_{2}(\tau)),$$
(3.2.31)

with

$$\overline{x}_2(t_0) = \xi_2$$
 (3.2.32)

At last we obtain the suboptimum control as follows:

$$u_{sub} = r^{-1} [(k_2^0 + \varepsilon k_2^1)(x_1^0 + \varepsilon x_1^1) + (k_3^0 + \varepsilon k_3^1)(x_2^0 + \varepsilon x_2^1)].$$
(3.2.33)

3.2.3 Boundary layer corrections

In order to obtain the uniformly valid expansion throughout the interval $[t_0, t_f]$, we seek a solution of the form

$$\tilde{K}(t,\varepsilon) = K(t,\varepsilon) + H(\tau,\varepsilon),$$
 (3.2.34)

~ ~ / \

where $\tau = (t_f - t)/\epsilon$. Here the boundary layer correction $H(\tau, \epsilon)$ vanishes as τ tends to infinity, i.e.,

$$\lim_{\tau \to \infty} H(\tau, \varepsilon) = 0. \qquad (3.2.35)$$

Further $H(\tau, \varepsilon)$ has an asymptotic expansion in ε as $\varepsilon \rightarrow 0$ of the form

$$H(\tau,\varepsilon) = \sum_{r=0}^{\infty} H^{r}(\tau)\varepsilon^{r}(r!)^{-1}. \qquad (3.2.36)$$

To determine each coefficient $H^{r}(\tau)$, as in the usual way, we substitute the asymptotic power series

$$\tilde{K}(t,\varepsilon) = \sum_{r=0}^{\infty} [K^{r}(t) + H^{r}(\tau)]\varepsilon^{r}(r!)^{-1}, \qquad (3.2.37)$$

into the stretched system

$$-\frac{d}{d\tau}\tilde{k}_{1} = \varepsilon(2a\tilde{k}_{1} - 2\tilde{k}_{2} + r^{-1}\tilde{k}_{2}^{2}),$$

$$-\frac{d}{d\tau}\tilde{k}_{2} = -a\tilde{k}_{1} + \tilde{k}_{2} - \tilde{k}_{3} + r^{-1}\tilde{k}_{2}\tilde{k}_{3},$$
 (3.2.38)

$$-\frac{d}{d\tau}\tilde{k}_{3} = 2\tilde{k}_{3} + r^{-1}\tilde{k}_{3}^{2} - 1,$$

and compare coefficients of like powers of ε thereof, then the following recursive equations for the boundary layer coefficients are obtained:

$$\begin{aligned} &-\frac{d}{d\tau} h_1^0 = 0, \qquad -\frac{d}{d\tau} h_2^0 = -ah_1^0(\tau) + h_2^0(\tau) - h_3^0(\tau) \\ &+ r^{-1} [k_2^0(t_f) h_3^0(\tau) + k_3^0(t_f) h_2^0(\tau) + h_2^0(\tau) h_3^0(\tau)], \qquad (3.2.39)_0 \\ &-\frac{d}{d\tau} h_3^0 = 2h_3^0(\tau) + r^{-1} [2k_3^0(t_f) h_3^0(\tau) + (h_3^0(\tau))^2] - 1, \\ &-\frac{d}{d\tau} h_1^1 = -2ah_1^0(\tau) - 2h_2^0(\tau) + r^{-1} [k_2^0(t_f) h_2^0(\tau) + (h_2^0(\tau))^2], \\ &-\frac{d}{d\tau} h_2^1 = -ah_1^1(\tau) + h_2^1(\tau) - h_3^1(\tau) + ah_2^0(\tau) \qquad (3.2.39)_1 \\ &+ r^{-1} [k_1^0(t_f) h_2^0(\tau) + k_2^0(t_f) h_1^0(\tau) + h_1^0(\tau) h_2^0(\tau)], \\ &-\frac{d}{d\tau} h_3^1 = 2h_3^1(\tau) + r^{-1} [2k_3^0(t_f) h_3^1(\tau) + 2k_3^1(t_f) h_3^0(\tau) + \\ &+ h_3^0(\tau) h_3^1(\tau)] - 2ah_2^0(\tau). \end{aligned}$$

Each recursive equations can be solved under the terminal conditions derived from the requirement

$$K^{r}(t_{f}) + H^{r}(0) = 0.$$
 (3.2.40)

3.2.4 Numerical example

We shall present the results obtained by numerical calculations according to the method described in the preceding section. The data used in the example are the same as in Section 3.1.4, namely $\beta = 0.0064$, $a = \lambda = 0.078$; mean life time of the prompt neutron $\ell = 3 \times 10^{-5}$, and therefore $\varepsilon = 0.47 \times 10^{-2}$.

In Figs. 3.2.1 and 3.2.2, the time behaviours of the solution of the Riccati equation (3.2.18) and those of the approximate solution derived via the singular perturbation theory are illustrated for two values of the weight r. The approximation is taken of the form of truncated series solution at the second term. The boundary layer appears in the case of the reduced solution.

In general, it can be seen that the feedback coefficients determined by solving the Riccati equation change rapidly in the neighbourhood of the terminal time, and computational difficulties should be encountered such as overflow of computers or increased size of the memory needed. Moreover, in this case the existence of the boundary layer makes the rate of the change larger and the difficulties should become more serious. The method adopted can make us avoid the difficulties, by using the idea of separation of the time scale.

Figures 3.2.3 and 3.2.4 show the time behaviours of the optimum and suboptimum trajectories of neutrons for r = 10 and r = 0.4 respectively. Table 3.2.1 represents values of the performance index calculated by using the optimum and suboptimum control.

Our numerical study shows that the reduced solution with boundary layer corrections (0-th order approximation) gives a satisfactory result. The error of the 0-th order approximation is, except the terminal time, of the order of 0.1 percent for trajectories of neutrons and 1 percent for feedback coefficients. The first correction term improves the approximation uniformly in the time coordinate, especially does in the boundary layers. The memory can be saved about 60 percent when we adopt our 0-th order approximation. This saving is a great

advantage of the present method.

These consequences show that the method gives satisfactory results from view points of computaional labour, computing time, accuracy of the approximation.

It is to be noted that when the original state equation is stiff, then the generated matrix Riccati equation will be stiffer because of the nonlinearity of the Riccati equation. So the method of separation of the time scale or the method of multiple time scale may be effective to such a problem. Above all the singular perturbation theory plays a very important role as seen in this example.



Table 3.2.1 Performance index





Chaper 4 Near-Optimum Control of Distributed Parameter System

4.1 Introduction

It is true that every realistic physical system is intrinsically distributed in nature. From practical motivations, however, the system's spatial distributed parameters are usually assumed to be sufficiently concentrated or invariant in form during the course of motion so that an approximate lumped parameter model may provide an adequate description. Although such a treatment is feasible in many physical systems, it is obvious that the effects of spatial distribution of practical systems are lost. In the precise consideration of the effects it is required to investigate the system maintaining the spatial distribution of physical parameters, so that a study of the theory of partial differential equations becomes indispensable. Typical examples of this case are nuclear reactors, many metallurgical plants, chemical reactors, and so forth.

Optimal control theory of lumped parameter systems has been developed powerfully and successfully by many mathematicians and control engineers, and is almost complete in theory. A maximum principle (and minimum principle) was obtained by Pontryagin and his students, which is a generalization of the classical variational theory in a sense [Pn.62]. Dynamic programming was found by Bellman independently of the maximum principle [B.57]. Kalman showed that the maximum principle, dynamic programming, and Hamilton-Jacobi theory are equivalent to each other, and constructed general solutions of linear state regulator problem by introducing the Riccati transformation and established the basic results [K.60]. The treatment adopted by Kalman seems to be one of the main streams in control theory. Theories above mentioned are systematic and complete in theory, and provide us powerful gu des.

On the other hand, control theory of distributed parameter system is not so sufficiently developed, since it is by far complex and difficult to handle or to analyze partial differential equations. Modern treatments of partial differential equations have recently made remark

able progress with the development of functional analysis; see for instance Dunford and Schwartz [Df.58], Lions and Magenes [L.72], or Mizohata [M.65]. On the basis of this development, optimal control of distributed parameter systems has become one of the most productive areas.

One of the earliest investigators of the subject was Butkovskii (see references listed in [Bt.69]). Butkovskii extended Pontryagin's maximum principle to a certain class of distributed systems, and studied the approximation method with the aid of method of moments.

Another approach was made by Wang [W.64]. His main theme seems to lie in applying the results of the theory of lumped parameter systems to distributed ones in line with Kalman, i.e., he obtained Hamilton-Jacobi formalism, a maximum principle, and Riccati-like partial differential integral equations. He also devoted himself to the study of stability, controllability, observability, and approximation methods.

Some other studies on the subject were made by Axelband [Ax.69], Brogan [Br.68], Lions [L.71], etc. with use made of functional analysis. Since it is not our object to review the theory comprehensively, for general surveys, Brogan [Br.68] or Lions's detailed and extensive work [L.71] may be referred to.

It is to be noted here that the practical treatment of partial differential equations cannot be made without involving an approximation, spatial discretization, modal expansion, etc., since the equations can be regarded to be infinite dimensional. Considering such a situation, it is easy to see the importance of the reduction of the number of independent variables if possible.

As mentioned in Section 1.3, the singular perturbations of abstract evolution equations are not fully developed, this chapter presents materials to contribute to the theory concerned: the chapter consists of two parts, the former one is concerned with the singular perturbations of a certain class of a set of evolution equations. The latter deals with near-optimum control problems on the basis of the results derived

in the former one. So far as the author knows, the singular perturbations of a set of partial differential equations involving construction of asymptotic expansions have never been treated. The regular degeneration of such systems was studied recently by Lions [L.73].

4.2 Evolution equations containing a small parameter

In this section, an investigation is made of the singular perturbation theory of a certain class of abstract evolution equations of the form

$$\frac{\partial}{\partial t} z = Az + f,$$

$$z = \begin{bmatrix} x \\ y \end{bmatrix}, \quad A = \begin{bmatrix} A_1 & A_2 \\ \epsilon^{-1}A_3 & \epsilon^{-1}A_4 \end{bmatrix},$$

where z is a state vector, operators A_1 and A_4 are parabolic, A_2 , A_3 are bounded, and ϵ is a small positive parameter.

This kind of system does not seem to have been treated with the closed form.

4.2.1 Mathematical notations and preliminaries

We shall consider functions defined on the sets as follows:

 Ω : a simply connected, open set in R^r.

- Γ : the boundary of Ω .
- $\Sigma = \Gamma \times]0,T].$
- $Q = \Omega \times]0,T].$

Points of Ω are denoted by $\omega_r = (\omega_1, \omega_2, \ldots, \omega_r)$.

Further some definitions are needed regarding the equivalence class of functions considered.

$$\begin{split} H &= L^2(\Omega) : \text{space of functions square integrable on } \Omega. \\ L^2(S;E) : \text{space of functions defined on S with values in a Hilbert} \\ &\qquad \text{space E, and whose second powers are integrable with} \end{split}$$

respect to the Lebesgue measure of S, $d\mu(t)$, i.e.

$$\int_{S} \left\| f(t) \right\|_{E}^{2} d\mu(t) < \infty.$$

 $H^{1}(\Omega)$: Sobolev space of order 1, i.e., space of functions ϕ such that

$$\phi$$
, $\frac{\partial \phi}{\partial \omega_1}$, $\frac{\partial \phi}{\partial \omega_2}$, \ldots , $\frac{\partial \phi}{\partial \omega_r}$ $\varepsilon L^2(\Omega)$.

 $H_0^1(\Omega)$: subspace of $H^1(\Omega)$, i.e., space of functions ϕ such that

$$\phi \in H^{1}(\Omega), \quad \phi \big|_{\Gamma} = 0.$$

The inner products and norm of elements f, g ϵ H are defined:

Inner product: $(f,g)_{H} = \int_{\Omega} f'(\omega)g(\omega)d\omega$. Norm: $\| f \|_{H} = [\int_{\Omega} |f(\omega)|^{2}d\omega]^{1/2}$.

For elements f, g ϵ L²(0,T;H),

Inner product:
$$(f,g)_{L^{2}(0,T;H)} = \int_{0}^{T} (f,g)_{H} dt.$$

Norm: $\|f\|_{L^{2}(0,T;H)} = [\int_{0}^{T} \|f\|_{H}^{2} dt]^{1/2}.$

In order to distinguish functions as elements of a function space and values taken by them, the following notations are adopted: $f(\omega,t)$ is a point in \mathbb{R}^1 , where $(\omega,t) \in \mathbb{Q}$. $f(\cdot,t)$ is an element of the Hilbert space H. $f(\cdot,\cdot)$ is an element of the Hilbert space $L^2(0,T;H)$. $f(\cdot, t, \epsilon)$ is an element of the Hilbert space $L^2(0,T;H)$. parameterized by a small parameter ϵ .

Hence the function $f(\cdot)$ is considered to be an element of some Hilbert space H. If the function f considered is m-dimensional vector valued, the minute notation

$$f \in L^{2}(\Omega) \times L^{2}(\Omega) \times \ldots \times L^{2}(\Omega) = L^{2,m}(\Omega)$$

should be adopted. In order to avoid notational trouble, it is noted that

f
$$\epsilon L^2(\Omega)$$
,

when no misunderstanding may occur.

For practical reasons, we consider here only separable Hilbert spaces; such a treatment is rather general. In a separable Hilbert space H, there exists a set of known "elementary" functions, countable and everywhere dense in H. The complete orthonormal system $\{\phi_i\}$, i = 1, 2, ... is adopted as such a set, and we have the following classical theorems [Km.71]:

<u>Theorem 4-1</u> In a separable Hilbert space, every complete orthonormal system is closed and conversely.

For L^2 the above theorem becomes more useful.

<u>Theorem 4-2</u>(Riesz-Fischer) Given a numerical sequence $\{c_k\}$ for which the series $\sum_{k=1}^{\infty} |c_k|^2$ is convergent, there exists a unique element f ε H such that its Fourier coefficients a_k are equal to c_k (k = 1, 2,) and

$$\sum_{k=1}^{\infty} |c_k|^2 = (f, f)_H = ||f||_{\frac{2}{H}}.$$
 (4.2.1)

Thus for any f ϵ H,

$$\lim_{m \to \infty} \| f - \sum_{k=1}^{m} c_k \phi_k \|_{H}^2 \to 0, \qquad (4.2.2)$$

where

$$c_k = (f, \phi_k) = \int_{\Omega} f(\omega) \phi_k(\omega) d\omega.$$

We use in the sequel mainly a system of complete orthonormal functions generated by the Strum-Liouville equation:

$$A[\phi] = \lambda \phi, \quad \text{in } \Omega \tag{4.2.3}$$

with any one of the boundary conditions

i)
$$\phi(\Gamma) = 0,$$
 (4.2.4)

ii)
$$\frac{\partial \phi}{\partial v_A}(\Gamma) = 0,$$
 (4.2.4')

iii)
$$\frac{\partial \phi}{\partial v_A}(\Gamma) + \sigma(\Gamma)\phi(\Gamma) = 0; \quad \sigma(\Gamma)>0,$$
 (4.2.4")

where

$$\frac{\partial}{\partial v_A}(\cdot) = \sum_{j=1}^r a_{ij\partial x_j}(\cdot)\cos(n,x), \quad \text{on } \Gamma.$$

It is assumed that

$$A[\phi] = -\sum_{i,j=1}^{L} \frac{\partial}{\partial \omega_{i}} [a_{ij}(\omega) \frac{\partial \phi}{\partial \omega_{j}}] + a_{0}(\omega) \phi(\omega), \qquad (4.2.5)$$

$$\sum_{ij} a_{ij}(\omega) \xi_{i} \xi_{j} \ge \alpha (\xi_{1}^{2} + \xi_{2}^{2} + \cdots + \xi_{r}^{2})$$
(4.2.6)

holds almost everywhere in Ω for all $\xi = (\xi_1, \xi_2, \dots, \xi_r)$ in \mathbb{R}^r .

The theorem described below is also called the Riesz-Fischer's theorem, and shall play an important role in Section 4.2.3.

<u>Theorem 4-3</u>(Riesz-Fischer) A necessary and sufficient condition for the given sequence of functions $\{f_n\}$ to be convergent to an element $f \in H$ is that

 $\| \mathbf{f}_{\mathbf{m}} - \mathbf{f}_{\mathbf{n}} \|_{\mathbf{H}} \rightarrow 0 \text{ as } \mathbf{m}, \mathbf{n} \rightarrow \infty.$

4.2.2 Formal construction of asymptotic expansion

We describe the formal procedure of constructing the asymptotic expansion of the solution of the Cauchy problem of the form

$$\frac{\partial}{\partial t}\tilde{x}(\omega,t,\varepsilon) + A_{1}\tilde{x}(\omega,t,\varepsilon) + A_{2}\tilde{y}(\omega,t,\varepsilon) = f(\omega,t,\varepsilon), \quad (4.2.7)$$

$$\varepsilon \frac{\partial}{\partial t} \tilde{y}(\omega, t, \varepsilon) + A_3 \tilde{x}(\omega, t, \varepsilon) + A_4 \tilde{y}(\omega, t, \varepsilon) = g(\omega, t, \varepsilon), \quad (4.2.8)$$

with initial data

$$\widetilde{\mathbf{x}}(\omega,0,\varepsilon) = \alpha(\omega,\varepsilon),$$

$$\widetilde{\mathbf{y}}(\omega,0,\varepsilon) = \beta(\omega,\varepsilon),$$
(4.2.9)

and boundary condition

B(x, y) = 0, on
$$ω ε Ω$$
, (4.2.10)

where A_1 , A_2 , A_3 , and A_4 are linear operators whose properties are specified in the succeeding Section 4.2.3.

On the analogy of the lumped parameter case, we seek the solution of (4.2.7) and (4.2.8) of the form,

$$\begin{aligned} \mathbf{x}(\omega, \mathbf{t}, \varepsilon) &= \mathbf{x}(\omega, \mathbf{t}, \varepsilon) + \mathbf{X}(\omega, \mathbf{t}/\varepsilon, \varepsilon), \\ \tilde{\mathbf{y}}(\omega, \mathbf{t}, \varepsilon) &= \mathbf{y}(\omega, \mathbf{t}, \varepsilon) + \mathbf{Y}(\omega, \mathbf{t}/\varepsilon, \varepsilon), \end{aligned} \tag{4.2.11}$$

where $x(\omega, t, \varepsilon)$, $y(\omega, t, \varepsilon)$, $X(\omega, t/\varepsilon, \varepsilon)$ and $Y(\omega, t/\varepsilon, \varepsilon)$ are assumed to have asymptotic expansions with respect to ε , as follows:

$$x(\omega,t,\varepsilon) = \sum_{\substack{r=0\\r=0}}^{\infty} x^{r}(\omega,t)\varepsilon^{r}(r!)^{-1},$$

$$y(\omega,t,\varepsilon) = \sum_{\substack{r=0\\r=0}}^{\infty} y^{r}(\omega,t)\varepsilon^{r}(r!)^{-1},$$
(4.2.12)

$$X(\omega, t/\varepsilon, \varepsilon) = \sum_{r=0}^{\infty} X^{r}(\omega, t/\varepsilon) \varepsilon^{r}(r!)^{-1},$$

$$Y(\omega, t/\varepsilon, \varepsilon) = \sum_{r=0}^{\infty} Y^{r}(\omega, t/\varepsilon) \varepsilon^{r}(r!)^{-1}.$$
(4.2.13)

It is easily seen that $x(\omega, t, \varepsilon)$ and $y(\omega, t, \varepsilon)$ represent the outer solution, and $X(\omega, \tau, \varepsilon)$ and $Y(\omega, \tau, \varepsilon)$ the boundary layer correction. Since Eq.(4.2.11) satisfies the given initial data (4.2.9), the following relations hold:

$$x^{r}(\omega,0) + X^{r}(\omega,0) = \alpha^{r}(\omega),$$

$$y^{r}(\omega,0) + Y^{r}(\omega,0) = \beta^{r}(\omega),$$
(4.2.14)

where $\alpha^{\mathbf{r}}(\omega)$ and $\beta^{\mathbf{r}}(\omega)$ are coefficients of the Taylor series of the given initial data (4.2.9). Further the boundary layer corrections $X^{\mathbf{r}}(\omega, t/\varepsilon)$ and $Y^{\mathbf{r}}(\omega, t/\varepsilon)$ are required to vanish as the stretched coordinate $\tau = t/\varepsilon$ tends to infinity,

$$\lim_{\substack{\tau \to \infty}} X^{\mathbf{r}}(\omega, \tau) = 0, \qquad (4.2.15)$$

$$\lim_{\substack{\tau \to \infty}} Y^{\mathbf{r}}(\omega, \tau) = 0. \qquad (4.2.16)$$

In the outer region, Eqs.(4.2.7) and (4.2.8) are valid for the outer expansion. We substitute the expansion (4.2.12) into Eqs.(4.2. 7) and (4.2.8), and compare the coefficients of like powers of ε , obtaining the following recursive set of equations.

$$\frac{\partial}{\partial t} x^{0} + A_{1} x^{0} + A_{2} y^{0} = f, \qquad (4.2.17)_{0}$$

$$A_3 x^0 + A_4 y^0 = g,$$
 (4.2.18)

$$\frac{\partial}{\partial t} x^{r} + A_{1} x^{r} + A_{2} y^{r} = 0, \qquad (4.2.17)_{r}$$

$$\frac{\partial}{\partial t} y^{r-1} + A_{3} x^{r} + A_{4} y^{r} = 0. \qquad (4.2.18)_{r}$$

For simplicity, we assume here that A_1 , A_2 , A_3 , A_4 , f and g are independent of ε . This assumption is not essential.

In order to make the boundary layer correction, the boundary layer method is adopted. The stretching transformation

$$\tau = t/\varepsilon, \qquad (4.2.19)$$

makes Eqs.(4.2.7) and (4.2.8) to be of the stretched form

$$\frac{\partial}{\partial \tau} \tilde{x}(\omega, \tau, \varepsilon) + A_1 \tilde{x}(\omega, \tau, \varepsilon) + A_2 \tilde{y}(\omega, \tau, \varepsilon) = f(\omega, \tau), \quad (4.2.20)$$

$$\frac{\partial}{\partial \tau} \tilde{y}(\omega, \tau, \varepsilon) + A_{3} \tilde{x}(\omega, \tau, \varepsilon) + A_{4} \tilde{y}(\omega, \tau, \varepsilon) = g(\omega, \tau). \qquad (4.2.21)$$

Substituting the solution (4.2.11) into Eqs.(4.2.20) and (4.2.21), we have

$$\frac{\partial}{\partial \tau} X(\omega, \tau, \varepsilon) + \varepsilon A_1 X(\omega, \tau, \varepsilon) + \varepsilon A_2 Y(\omega, \tau, \varepsilon) = 0, \qquad (4.2.22)$$

$$\frac{\partial}{\partial \tau} Y(\omega, \tau, \varepsilon) + A_3 X(\omega, \tau, \varepsilon) + A_4 Y(\omega, \tau, \varepsilon) = 0, \qquad (4.2.23)$$

where it is considered that outer solution x(t) and y(t) satisfy Eq. (4.2.7) and (4.2.8) with $t = \varepsilon \tau$. Putting each expansion (4.2.12) and (4.2.13) in place of $x(\omega, t, \varepsilon)$, $y(\omega, t, \varepsilon)$, $X(\omega, \tau, \varepsilon)$ and $Y(\omega, \tau, \varepsilon)$ in Eqs.(4.2.22) and (4.2.23), we have the following recursive equations for the boundary layer corrections,

$$\frac{\partial}{\partial \tau} X^{0}(\omega, \tau) = 0, \qquad (4.2.24)_{0}$$

$$\frac{\partial}{\partial \tau} Y^{0}(\omega, \tau) + A_{3} X^{0}(\omega, \tau) + A_{4} Y^{0}(\omega, \tau) = 0, \qquad (4.2.25)_{0}$$

$$\frac{\partial}{\partial \tau} X^{r}(\omega, \tau) = P_{r}(\omega, \tau), \qquad (4.2.24)_{r}$$

$$\frac{\partial}{\partial \tau} Y^{\mathbf{r}}(\omega,\tau) + A_{3} X^{\mathbf{r}}(\omega,\tau) + A_{4} Y^{\mathbf{r}}(\omega,\tau) = 0. \qquad (4.2.25)_{\mathbf{r}}$$

The remainders $P_r(\omega,\tau)$ are known successively in the preceding steps.

The initial data for the reduced outer equation is given by using the cancellation law, as

$$\mathbf{x}^{0}(\omega,0) = \alpha^{0}(\omega), \quad \omega \in \Omega, \quad (4.2.26)$$

hence the initial data for the boundary layer correction of the 0-th order (reduced), obtained considering Eq.(4.2.13), is as follows:

$$X^{0}(\omega, 0) = 0, \quad \omega \in \Omega,$$
 (4.2.27)

$$Y^{0}(ω, 0) = β^{0}(ω) - y^{0}(ω, 0); ω ε Ω.$$
 (4.2.28)

For the higher order recursive equations, the initial data is determined as follows. From Eq.(4.2.17)₁, $X^{1}(\omega,0)$ can be determined as

$$X^{1}(\omega,\tau) = X^{1}(\omega,0) + \int_{0}^{\tau} P_{1}(\omega,\tau)d\tau$$

= $X^{1}(\omega,0) - \int_{0}^{\tau} A_{2}Y^{0}(\omega,\tau)d\tau.$ (4.2.29)

Considering the property (4.2.15), we choose $X^{1}(\omega, 0)$ as

$$X^{1}(\omega,0) = \int_{0}^{\infty} A_{2}Y^{0}(\omega,\tau)d\tau. \qquad (4.2.30)$$

This infinite integral may exist if the integrand is dominated by functions of the boundary layer type which should be ascertained if Eq.(4.2.23) is asymptotically stable with respect to the root Y = 0. The conditions needed can be found in the follwing section. Then the initial data for the outer second coefficient of the outer expansion is derived by using Eq.(4.2.14) as

$$x^{1}(\omega, 0) = \alpha^{1}(\omega) - \int_{0}^{\infty} A_{2}Y^{0}(\omega, \tau)d\tau.$$
 (4.2.31)

A similar procedure may treat successfully the higher order equations.

If we obtain the Green's function $G_4(\omega, \omega')$ for $(4.2.18)_0$, then we can solve Eq. $(4.2.18)_0$ for $y^0(\omega, t)$ as

$$y^{0}(\omega,t) = \int_{\Omega} G_{4}(\omega,\omega')[g(\omega',t) - A_{3}x^{0}(\omega',t)]d\omega', \qquad (4.2.32)$$

or in the notation of the operator theory

$$y^{0}(\omega,t) = A_{4}^{-1}[g(\omega,t) - A_{3}x^{0}(\omega,t)],$$
 (4.2.33)

where A_4^{-1} is defined by

$$A_4^{-1}[\cdot] = \int_{\Omega} G_4(\omega, \omega')[\cdot] d\omega'.$$
 (4.2.34)

The above procedure eliminates $y^{0}(\omega,t)$ in Eq.(4.2.7) formally, and we can solve Eq.(4.2.7) for $x^{0}(\omega,t)$ with the initial data (4.2.26).

The same procedure can be applied to the higher order equation.

If the right hand sides of Eqs.(4.2.7) and (4.2.8) depend upon \tilde{x} and \tilde{y} regularly in the sense that

f, g = o($|\tilde{x}|$, $|\tilde{y}|$) as $|\tilde{x}|$, $|\tilde{y}| \rightarrow 0$

holds, the resulting recursive set of equations can be solved stepwise similarly.

4.2.3 Main theorems

The regular degeneration of the system (4.2.1) and (4.2.2), and the asymptotic accuracy of the result obtained formally above shall be studied.

Some prerequisite assumptions are offered, first we state considering Theorems 4-1 and 4-2 in Section 4.2.1:

<u>Assumption 1</u> A_4 is a linear operator which generates an orthonormal set of eigenfunctions, i.e., there exist a sequence of eigenfunctions $\{\phi_i\}$ and a sequence of eigenvalues $\{\lambda_i\}$ satisfying

$$- A_{\lambda}[\phi] = \lambda \phi, \qquad (4.2.35)$$

with the boundary condition that is any one of the conditions given above (4.2.4) - (4.2.4") and denoted here by

$$B_{2}[\phi] = 0. \tag{4.2.36}$$

And the sequence of eigenfunctions is orthonormal in the space $L^2(\Omega)$, that is

$$\int_{\Omega} \phi_{n} \phi_{m} d\omega = \delta_{nm},$$
$$\delta_{nm} = \{ \begin{array}{cc} 1 & n = m \\ 0, & n \neq m \end{cases} \}$$

<u>Assumption 2</u> A_1 admits the same eigenfunction as A_4 , e.g., A_1 can be expressed as,

$$A_1 = -kA_4 + h$$
 (4.2.37)

where k and h are constant with respect to the spatial coordinates.

Assumption 3 A_2 and A_3 are spatially homogeneous.

If the Assumptions 1 and 2 hold, we can expand each variable into an eigenfunction expansion by using Theorems 4-1 and 4-2:

$$x(\omega, t, \varepsilon) = \sum_{i=1}^{\infty} \xi_{i}(t, \varepsilon) \phi_{i}(\omega),$$

$$y(\omega, t, \varepsilon) = \sum_{i=1}^{\infty} \eta_{i}(t, \varepsilon) \phi_{i}(\omega).$$
(4.2.38)

Substituting the expansion (4.2.38) into Eqs.(4.2.7) and (4.2.8), we derive the following set of equations by using the orthonormality of eigenfunctions ϕ_i ,

$$\frac{d}{dt}\xi_{i}(t,\varepsilon) + (k\lambda_{i} + h)\xi_{i}(t,\varepsilon) + A_{2}n_{i}(t,\varepsilon) = f_{i}(t), \quad (4.2.39)$$

$$\varepsilon \frac{d}{dt} n_{i}(t,\varepsilon) + A\xi_{i}(t,\varepsilon) - \lambda n_{i}(t,\varepsilon) = g_{i}(t), \quad (4.2.40)$$

$$\varepsilon \frac{1}{dt} \eta_{i}(t,\varepsilon) + A_{3}\xi_{i}(t,\varepsilon) - \lambda_{i}\eta_{i}(t,\varepsilon) = g_{i}(t), \quad (4.2.40)$$

where

$$f_{i}(t) = \int_{\Omega} \phi_{i}'(\omega) f(t, \omega) d\omega,$$

$$g_{i}(t) = \int_{\Omega} \phi_{i}'(\omega) g(t, \omega) d\omega.$$
(4.2.41)

For the boundary layer correction terms, the same expansion by the eigenfunctions is adopted. The resulting set of equations is as follows:

$$\frac{d}{d\tau} \Xi_{i}(\varepsilon\tau,\varepsilon) + \varepsilon(k\lambda_{i} + h)\Xi_{i}(\varepsilon\tau,\varepsilon) + \varepsilon A_{2}H_{i}(\varepsilon\tau,\varepsilon) = \varepsilon f_{i}(\varepsilon\tau), (4.2.42)$$
$$\frac{d}{d\tau} H_{i}(\varepsilon\tau,\varepsilon) + A_{3}\Xi_{i}(\varepsilon\tau,\varepsilon) - \lambda_{i}H_{i}(\varepsilon\tau,\varepsilon) = g_{i}(\varepsilon\tau). \quad (4.2.43)$$

Now we introduce an important assumption which ascertains the asymptotic stability of the boundary layer system, as mentioned on occasion in the lumped parameter case.

Assumption 4 The Cauchy problem

$$\frac{\partial u}{\partial t} + A_4 u = 0 \tag{4.2.44}$$

is uniformly well-posed in the sense of Hadamard, and the semigroup

 $U(t,\tau)$ of this problem satisfies

$$|| U(t,\tau) || \leq \operatorname{Cexp}(-\delta t), \text{ for } \delta > 0.$$
(4.2.45)

Now consider the eigenfunction expansion of the reduced system using the same eigenfunction as used for the full system (4.2.39), (4.2.40):

$$\frac{d}{dt} \xi_{i}(t,0) + (k\lambda_{i} + h)\xi_{i}(t,0) + A_{2}\eta_{i}(t,0) = f_{i}(t), \quad (4.2.46)$$
$$A_{3}\xi_{i}(t,0) - \lambda_{i}\eta_{i}(t,0) = g_{i}(t), \quad (4.2.47)$$

where $f_i(t)$, $g_i(t)$ are the same as those given in Eq.(4.2.41).

We have the following lemmas required in the sequel.

Lemma 4-1 If Assumptions 1 - 4 hold, then

$$\lim_{\epsilon \to 0} \xi_{i}(t) = \xi_{i}^{0}(t) \equiv \xi_{i}^{*}(t,0), \text{ for } t \in [0,T], \qquad (4.2.48)$$
$$\lim_{\epsilon \to 0} \eta_{i}(t) = \eta_{i}^{0}(t) \equiv \eta_{i}(t,0), \text{ for } t \in [0,T]. \qquad (4.2.49)$$

Proof Assumption 2 leads to

$$- \int_{\Omega} \phi'_{i}(\omega) A_{4} \phi_{i}(\omega) d\omega$$
$$= \int_{\Omega} \phi'_{i}(\omega) \lambda_{i} \phi_{i}(\omega) d\omega$$
$$= \lambda_{i} < 0.$$

Hence the boundary layer system for each mode n_i is asymptotically stable, which assures the regular degeneration of the full expanded i-th mode equations (4.2.39) and (4.2.40) from Theorem 1-1. This completes the proof. Q.E.D.

Lemma 4-2 If Assumptions 1 - 4 hold, then $\lim_{\epsilon \to 0} \lim_{n \to \infty} \lim_{j \to 0} ||\xi_j(t) - \xi_j(t)||^2 = 0, \qquad (4.2.50)$

$$\lim_{t \to 0} \lim_{n \to \infty} \sum_{j=1}^{n} || n_{j}(t) - n_{j}(t) ||^{2} = 0.$$
(4.2.51)

Proof Assumptions 1 - 3 assure that the eigenvalues of the problem

$$- A_4 \phi(\omega) = \lambda \phi(\omega), \quad B_2[\phi] = 0,$$

constitute a sequence of monotonously decreasing

$$0 > \lambda_1 \stackrel{>}{=} \lambda_2 \stackrel{>}{=} \lambda_3 \stackrel{>}{=} \cdots \quad \rightarrow -\infty,$$

furthermore the series $\sum_{i=1}^{\infty} (\frac{1}{\lambda_i})^2$ converges to a finite value, i.e.

$$\sum_{i=1}^{\infty} \left(\frac{1}{\lambda_i} \right)^2 < \infty .$$

Hence we have the following estimate for λ_i :

$$\lambda_i = 0(i^r), \quad r \geq 1 \text{ as } i \to \infty.$$

Transforming λ_i into μ_i by the relation

$$\lambda_i = \mu_i / i^r$$
,

Eqs.(4.2.39) and (4.2.40) become

$$\frac{d}{dt}\xi_{i}(t,\varepsilon) + (ki^{r}\mu_{i} + h)\xi_{i}(t,\varepsilon) + A_{2}\eta_{i}(t,\varepsilon) = f_{i}(t), \quad (4.2.52)$$

$$\frac{\varepsilon}{i^{r}}\frac{d}{dt}\eta_{i}(t,\varepsilon) + A_{3}\xi_{i}(t,\varepsilon) - \mu_{i}\eta_{i}(t,\varepsilon) = g_{i}(t). \quad (4.2.53)$$

Therefore Theorem 1-2 can be applied to this case. The result is as follows:

$$\|\xi_{i}(t) - \xi_{i}^{0}(t)\| = O(\frac{\varepsilon}{i^{r}}),$$
 (4.2.54)

$$\|\eta_{i}(t) - \eta_{i}^{0}(t)\| = O(\frac{\varepsilon}{i^{r}}).$$
 (4.2.55)

Then the limit relation

$$\lim_{\varepsilon \to 0} \lim_{n \to \infty} \int_{j=1}^{\infty} \left(\frac{\varepsilon}{ir} \right)^2 = 0$$

completes the proof. Q.E.D.

For example, the most typical eigenfunctions we often meet are those given by

$$\frac{\partial^2}{\partial \omega^2} \phi + \lambda \phi = 0 \quad \text{on }]0,1[, \qquad (4.2.56)$$

$$\phi(0) = 0, \quad \phi(1) = 0, \qquad (4.2.57)$$

namely

$$\{\sqrt{2} \sin \sqrt{\lambda_{i}} \omega\}, i = 1, 2, \dots, \lambda_{i} = (i\pi)^{2}.$$

The sequence of the eigenvalues $\{\lambda_i\}$ satisfies

$$\sum_{i=1}^{\infty} \left(\frac{1}{\lambda_i} \right)^2 < \infty.$$

By using Lemmas 4-1 and 4-2, we obtain the desired theorem.

<u>Theorem 4-4</u> If Assumptions 1 - 4 hold, then the regular degeneration of the system (4.2.7), (4.2.8) is concluded, as follows:

$$\lim_{\epsilon \to 0} x(\omega, t, \epsilon) = x^{0}(\omega, t), \text{ for } t \epsilon [0, T], \qquad (4.2.58)$$
$$\lim_{\epsilon \to 0} y(\omega, t, \epsilon) = y^{0}(\omega, t), \text{ for } t \epsilon]0, T]. \qquad (4.2.59)$$

<u>Proof</u> Lemma 4-2 assures that, as ε tends to zero

$$\| \xi_{i}(t,\varepsilon) - \xi_{i}^{0}(t) \| \rightarrow 0; \text{ uniformly,} \\ \| \eta_{i}(t,\varepsilon) - \eta_{i}^{0}(t) \| \rightarrow 0; \text{ uniformly.}$$

Hence the series

$$\mathbf{x}(\omega, \mathbf{t}, \varepsilon) - \mathbf{x}^{0}(\omega, \mathbf{t}) = \sum_{i=1}^{\infty} [\xi_{i}(\mathbf{t}, \varepsilon) - \xi_{i}^{0}(\mathbf{t})]\phi_{i}(\omega), \qquad (4.2.60)$$

$$y(\omega,t,\varepsilon) - y^{0}(\omega,t) = \sum_{i=1}^{\infty} [n_{i}(t,\varepsilon) - n_{i}^{0}(t)]\phi_{i}(\omega), \qquad (4.2.61)$$

converge uniformly to zero respectively, which complete the proof. Q.E.D.

We expand each variable ξ_i and η_i into Taylor series in ϵ_i as

$$\xi_{i}(t,\varepsilon) = \sum_{r=0}^{\infty} \xi_{i}^{r}(t)\varepsilon^{r}(r!)^{-1},$$

$$\eta_{i}(t,\varepsilon) = \sum_{r=0}^{\infty} \eta_{i}^{r}(t)\varepsilon^{r}(r!)^{-1}.$$
(4.2.62)

Substituting the resulting series (4.2.62) into Eqs.(4.2.39) and (4.2.40), then comparing the coefficients of like powers of ε , we derive the following set of equations:

$$\frac{d}{dt} \xi_{i}^{r}(t) + (k\lambda_{i} + h)\xi_{i}^{r}(t) + A_{2}n_{i}^{r}(t) = p_{r}(t), \qquad (4.2.63)$$

$$\frac{d}{dt} \eta_{i}^{r-1}(t) + A_{3}\xi_{i}^{r}(t) - \lambda_{i}\eta_{i}^{r}(t) = q_{r}(t), \qquad (4.2.64)$$

where r > 0 and the remainders ${\rm P}_{\rm r}^{}, \, {\rm Q}_{\rm r}^{}$ are known stepwise in the preceding steps.

By the analogous arguments made in the proof of Theorem 4-4, the following theorem, giving the asymptotic accuracy, is derived.

<u>Theorem 4-5</u> If Assumptions 1 - 4 hold, then there exist bounded functions $R_n(\omega, t; \varepsilon)$ and $S_n(\omega, t; \varepsilon)$ such that

where $\delta = -C \epsilon \log \epsilon$; C is independent of ϵ .

The proof of this theorem can be made by using the following lemma.

Lemma 4-3 If Assumptions 1 - 4 hold, then

$$\mathbf{x}^{\mathbf{r}}(\omega, t) = \sum_{i=1}^{\infty} \xi_{i}^{\mathbf{r}}(t) \phi_{i}(\omega), \qquad (4.2.67)$$
$$\mathbf{y}^{\mathbf{r}}(\omega, t) = \sum_{i=1}^{\infty} \eta_{i}^{\mathbf{r}}(t) \phi_{i}(\omega), \qquad (4.2.68)$$

where x^{r} and y^{r} are coefficients of ε^{r} of their Taylor series. This lemma is a well-known idea of separation of variables.

As to the boundary layer corrections, we can proceed our argument in line with the lumped parameter case and derive the results similar to Theorem 1-2.

The proofs given in this section are constructive when the eigenfunctions are determined.

4.2.4 Further generalization

The results obtained in the preceding section are applicable to the restricted class of systems satisfying Assumptions 1 - 4described therein. Such systems may provide a broad class in view point of applications, considering the complexity of computational treatment of distributed parameter systems. For instance, very few numerical examples can be seen in literatures published, which treat exceptional systems violating Assumptions 1 - 4.

This section shows the outline of the approach to the more general systems, time variant systems, systems involving spatially inhomogeneous coefficients, and so forth.

Consider the Cauchy problem

$$\frac{d}{dt} x + A_1(t,\varepsilon)x + A_2(t,\varepsilon)y = f(x, y, t, \varepsilon), x(0,\varepsilon) = \xi(\varepsilon),$$
(4.2.69)

$$\varepsilon \frac{d}{dt} y + A_3(t,\varepsilon)x + A_4(t,\varepsilon)y = g(x, y, t, \varepsilon), y(0,\varepsilon) = n(\varepsilon).$$
(4.2.70)

Systems of this type are intractable, for showing the existence and uniqueness of the solution. So several hypotheses are made for it.

H1. The operators $A_4(t,\varepsilon)$ are closed (possibly unbounded) linear operators acting in H with the domain of definition $D = D(A_4(t,\varepsilon))$ everywhere dense in H and independent of (t,ε) for $t \in [0,T]$, and $\varepsilon \in [0,\varepsilon_0]$.

H2. The resolvent of $-A_4(t,\varepsilon)$, $R(A_4(t,\varepsilon); \lambda) = [A_4(t,\varepsilon) + \lambda I]^{-1}$, exists as a bounded operator in H for each λ , $Re\lambda \geq 0$, and

$$\| R(A_4(t,\varepsilon);\lambda) \| \leq C' (1 + |\lambda|)^{-1}, \qquad (4.2.71)$$

for $\operatorname{Re}\lambda \geq 0$ and some positive constant c' independent of (t,ε) . The identity operator in H is denoted by I, and the operator norm over H by $||\cdot||$.

Thus we have:

For each $\sigma \in [0,T]$, $-A_4(t,\varepsilon)$ generates an analytic semigroup $\{\exp[-tA_4(\sigma,\varepsilon)]\}, t \ge 0.$

There exist positive numbers δ and C independent of t, $\sigma,$ and ϵ such that for each $\sigma\,\epsilon$ [0,T],

$$\frac{d}{dt} \{ \exp[-tA_4(\sigma, \varepsilon)]x \} = -A_4(\sigma, \varepsilon) \exp[-tA_4(\sigma, \varepsilon)]x, \quad x \in \Omega \\ t > 0, \\ \| \exp[-tA_4(\sigma, \varepsilon)] \| \leq Cexp(-t), \quad t > 0, \\ \| A_4(\sigma, \varepsilon) \exp[-tA_4(\sigma, \varepsilon)] \| \leq Ct^{-1}exp(-t), \quad t > 0. \end{cases}$$

Hence Assumption 4 in Section 4.2.3 holds. Also it is assumed that

H3. There is a positive constant c_0 such that

$$\| [A_4(t,\varepsilon) - A_4(\tau,\varepsilon)] A_4^{-1}(s,\varepsilon) \| \leq C_0 |t - \tau|, \quad (4.2.72)$$

for all $0 \leq t$, τ , $s \leq T$ and $0 \leq \varepsilon \leq \varepsilon_0$.

In order to specify the dependence of A_{i} on (t,ε) we assume

H4. The operator function $A_4(t,\varepsilon)A_4^{-1}(0,0)$ has strongly continuous derivatives of all orders with respect to (t,ε) , for $0 \le t \le T$, $0 \le \varepsilon \le \varepsilon_0$.

H5. The functions f and g have continuous derivatives of all orders with respect to (t, x, y, ε) ε [0,T] × S × [0, ε_0] where S is some sphere about the origin in HxH (see Dieudonne [D.60, P186]). With hypothesis H5, f and g can be expanded into the formal Taylor series

$$f = \sum_{i,j,k,l=0}^{\infty} f_{ijkl} \varepsilon^{i} t^{j} x^{k} y^{l}, \qquad (4.2.73)$$

$$g = \sum_{i,j,k,l=0}^{\infty} g_{ijkl} \varepsilon^{i} t^{j} x^{k} y^{l}, \qquad (4.2.74)$$

where f_{ijkl} and g_{ijkl} are kl-linear bounded operators in H, while the operator A_4 is linear but unbounded. The above power series converges uniformly in the Fréchet sense, in the sphere $||x_0||^2 + ||y_0||^2 + \varepsilon^2 \le \rho$.

The following conditions give a specification of the certain solution of the reduced problem.

H6. There are functions $x_0(t)$ and $y_0(t)$ which satisfy the reduced Cauchy problem

$$\frac{d}{dt}x^{0} + A_{1}(t,0)x^{0} + A_{2}(t,0)y^{0} = f(x^{0}, y^{0}, t, 0), (4.2.75)$$
$$A_{3}(t,0)x^{0} + A_{4}(t,0)y^{0} = g(x^{0}, y^{0}, t, 0), (4.2.76)$$

with the initial data

$$\mathbf{x}^{0}(0) = \xi(0), \qquad (4.2.77)$$

and which has continuous derivatives of all orders.

H7. There is a function $y^{*}(t)$ which satisfies

$$A_{3}(t,0)x + A_{4}(t,0)y^{*} = g(x, y^{*}, t, 0),$$
 (4.2.78)

and which has continous derivatives of all orders. Further y*(t) is isolated in the sense that for each t ε [0,T], there is a sphere about y*(t) which contains no other solutions of this equation.

H8.
$$f_x(x^0, y^0, t, 0) = 0, \quad f_y(x^0, y^0, t, 0) = 0,$$

 $g_x(x^0, y^0, t, 0) = 0, \quad g_y(x^0, y^0, t, 0) = 0,$ (4.2.79)

for t ε [0,T]. Here f denotes the Fréchet derivative of f with respect to x and similar for f_v, g_x, g_v.

The boundary layer system

$$\frac{d}{d\tau} Y + A_3(0,0)X + A_4(0,0)Y = g(X, Y, 0, 0), Y(0) = \eta(0), \quad (4.2.80)$$

is derived through stretching transformation $\tau = t/\epsilon$ and setting $\epsilon = 0$.

H9. The boundary layer Cauchy problem (4.2.80) has a unique solution, $Y = Y_0(\tau)$, which exists for $\tau \in [0,\infty]$.

H10. The full problem has a unique solution, (x(t), y(t)) which exists for t ε [0,T].

H11. The linear operator $A_4^{-1}(t,0)$ takes infinitely differentiable functions into infinitely differentiable functions.

Under these hypotheses, the following theorem similar to Theorem 1-2 holds:

<u>Theorem 4-6</u> If hypotheses H1 - H11 hold, there exist a constant $\varepsilon_0 > 0$ and bounded functions temporally uniform, $R_n(\omega, t, \varepsilon)$ and $S_n(\omega, t, \varepsilon)$, such that

$$x(\omega, t, \varepsilon) = \sum_{\substack{r=0\\n}}^{n} [x^{r}(\omega, t) + X^{r}(\omega, \tau)] \varepsilon^{r}(r!)^{-1} + R_{n}(\omega, t, \varepsilon) \varepsilon^{n+1}, \quad (4.2.81)$$
$$y(\omega, t, \varepsilon) = \sum_{\substack{r=0\\r=0}}^{n} [y^{r}(\omega, t) + Y^{r}(\omega, \tau)] \varepsilon^{r}(r!)^{-1} + S_{n}(\omega, t, \varepsilon) \varepsilon^{n+1}, \quad (4.2.82)$$
for $\varepsilon \in [0, \varepsilon_{0}], \quad t \in [0, T].$

The proof of this theorem will be possible by using the lemmas similar to those of lumped parameter cases, and by rephrasing the proof given in the lumped parameter cases, e.g., in Wasow [Ws.65] or in Hoppensteadt [Hp.71.1].

Lemma 4-4 If H1 and H2 hold, then there exist positive constants ε_0 , δ and C independent of t, ε , and such that

$$\| \exp[-tA_4(s,\varepsilon)/\varepsilon] \| \leq \operatorname{Cexp}[-\delta(t-s)/\varepsilon], \qquad (4.2.83)$$

for $0 \leq s \leq t \leq T$ and $\varepsilon \in [0, \varepsilon_0]$.

Lemma 4-5 If H1 - H11 hold, then there exists a set of positive constants C_r and δ_r such that

$$\left\| \begin{array}{l} X^{\mathbf{r}}(\tau) \right\| + \left\| \begin{array}{l} Y^{\mathbf{r}}(\tau) \right\| \leq C_{\mathbf{r}} \exp(-\delta_{\mathbf{r}}\tau), \\ \left\| \begin{array}{l} \frac{d}{d\tau} X^{\mathbf{r}}(\tau) \right\| + \left\| \begin{array}{l} \frac{d}{d\tau} Y^{\mathbf{r}}(\tau) \right\| \leq C_{\mathbf{r}} \exp(-\delta_{\mathbf{r}}\tau), \end{array} \right\|$$
(4.2.84)

where $X^{r}(\tau)$ and $Y^{r}(\tau)$ are solutions of the boundary layer equations corresponding to Eqs.(4.2.24)_r and (4.2.25)_r.

These lemmas can be shown in the similar way to those given in Hoppensteadt [Hp.69.1]. But our present object being the investigation of the singular perturbations of optimal control problems, with regard to proofs of these lemmas, Hoppensteadt [Hp.69.1] may be referred to.

4.3 Tracking problem

4.3.1 Problem statement

We are given Hilbert spaces V and H. The dual of V is denoted by V' and the dual of H is here assumed to be identified with H, and hence

$V \subset H \subset V'$.

For simplicity we are concerned with the system described by the following set of equations:

$$\frac{\partial}{\partial t} x - (a_1 \Delta - a_2) x + a_3 y = b_1 u_1, \qquad (4.3.1)$$

$$\varepsilon \frac{\partial}{\partial t} y - (a_4 \Delta - a_5) y + a_6 x = b_2 u_2, \qquad (4.3.2)$$

where $x \in H_1$ and $y \in H_2$ are scalars and a_i 's are spatially homogeneous and time invariant parameters, Adopting the matrix representation, we express the system, in place of Eqs.(4.3.1) and (4.3.2), as

$$I_{\varepsilon \partial t}^{\partial} z + Az = Bu, \qquad (4.3.3)$$

where

$$z = \begin{pmatrix} x \\ y \end{pmatrix}, A = \begin{pmatrix} -a_1 \Delta + a_2 & a_3 \\ a_6 & -a_4 \Delta + a_5 \end{pmatrix},$$

$$B = \begin{pmatrix} b_1 & 0 \\ 0 & b_2 \end{pmatrix}, u = \begin{pmatrix} u_1 \\ u_2 \end{pmatrix}.$$
 (4.3.4)

Let us denote the Hilbert space of control by U, and we suppose that

$$u = \{u_1, u_2\} \in \mathcal{U} = L^2(Q) \times L^2(Q).$$
 (4.3.5)

Then the operator B is given by

$$B \in \mathcal{L}(\mathcal{U}; V'). \tag{4.3.6}$$

We define $\hat{\mathcal{U}}$ by

$$\hat{\boldsymbol{u}} = \boldsymbol{L}^{2}(\boldsymbol{\Omega}) \times \boldsymbol{L}^{2}(\boldsymbol{\Omega}).$$
(4.3.7)

The initial and boundary conditions are given as below:

$$z(\omega,0) = z_0(\omega) \in \mathbb{H}, \quad \text{in } \Omega, \qquad (4.3.8)$$

$$z = 0$$
, on Σ . (4.3.9)

We also assume that

$$z \in L^2(0,T;V), \quad \frac{\partial}{\partial t} z \in L^2(0,T;V').$$
 (4.3.10)

The notations $L^2(0, T; V), L^2(0, T; V')$ are defined in Section 4.2.1.

The problem concerned is the tracking problem to transport the given initial state $z_0(\omega)$ to the desired state $z_d(\omega)$ given as

$$z_{d}(\omega) = \begin{bmatrix} x_{d}(\omega) \\ y_{d}(\omega) \end{bmatrix}; \quad z_{d} \in L^{2}(\Omega), \qquad (4.3.11)$$

with minimizing the following performance index:

$$J = \int_{0}^{1} ||z(t;u) - z_{d}(t)||_{\hat{H}} dt + (u, Ru)_{\mathcal{U}}, \qquad (4.3.12)$$

where

$$\mathcal{H} = L^2(0,T;\hat{\mathcal{H}}),$$
 (4.3.13)

and H is defined similarly as \dot{U} in Eq.(4.3.7).

In this case, we take

$$V = H_0^1(\Omega) \times H_0^1(\Omega), \qquad (4.3.14)$$

$$H = L^{2}(\Omega) \times L^{2}(\Omega). \qquad (4.3.15)$$

The adjoint equations satisfied by the adjoint variables P and q corresponding to the state variables x and y respectively, are introduced as follows:

$$- \varepsilon \frac{\partial}{\partial t} p - (a_1 \Delta - a_2) p + a_6 q = w_1 x - x_d, \qquad (4.3.16)$$

$$-\frac{\partial}{\partial t}q - (a_4 \Delta - a_5)q + a_3 p = w_2 y - y_d, \qquad (4.3.17)$$

or

$$-I_{\varepsilon \partial t} s + A^* s = W_7 - z_d.$$
(4.3.18)

In the case where there are no constraints, that is the optimal control u* minimizing J(u) when the control U ranges over the whole space U, is contained in the space of admissible control U_{ad} , we can derive the optimal control u* as
$$u^* = R^{-1}B's,$$
 (4.3.19)

or

$$u_1^* = r_1^{-1}b_1^p, \quad u_2^* = r_2^{-1}b_2^q.$$
 (4.3.19)

Eliminating u(t) in Eqs.(4.3.1) and (4.3.2) by using Eq.(4.3.18), the following canonical equation is obtained:

$$I_{\varepsilon \partial t}^{\partial} z + Az = BR^{-1}B's, \qquad (4.3.20)$$

$$- I_{\varepsilon} \frac{\partial}{\partial t} s + A^* s = W_z - z_d, \qquad (4.3.21)$$

with initial and final conditions

$$z(\omega,0) = z , \quad \omega \in \Omega, \quad (4.3.22)$$

$$s(\omega, T) = 0, \quad \omega \in \Omega.$$
 (4.3.23)

We decouple the canonical system (4.3.20), (4.3.21) by using the Riccati-like transformation: there exist an operator P(t) and a function r(t) such that

$$s(t) = P(t)z(t) + r(t),$$
 (4.3.24)

where P(t) is given by

$$- I_{\varepsilon \partial t} P(t) + P(t)A + A*P(t) + P(t)BR^{-1}B'P(t) = W, \quad (4.3.25)$$

$$P(t) \in \mathcal{L}(H; H).$$
 (4.3.26)

An operator P(t) satisfies Eq.(4.3.25) in the sense that

$$[-I_{\varepsilon \partial t}^{\partial} P(t)]\phi + P(t)A\phi + A*P(t)\phi + P(t)BR^{-1}B'P(t)\phi = W\phi, \quad (4.3.27)$$

for any $\phi \in V$.

The final terminal data are given by

$$P(T) = 0.$$
 (4.3.28)

which holds in the same sense as in Eq.(4.3.27).

The vector function r(t) satisfies the following linear parabolic equation

$$- I_{\varepsilon} \frac{\partial}{\partial t} r(t) + A^{*}r(t) + P(t)BR^{-1}B'r(t) = z_{d}(t), \quad (4.3.29)$$

$$r(t) \in H$$
, (4.3.30)

with the terminal condition

$$r(T) = 0.$$
 (4.3.31)

The validity of the above derivation of P(t) and r(t) was ascertained in Lions [L.71], whose approach is a straight extension of the results in lumped parameter case obtained by Kalman [K.60]. Meyer [My.73] may also be referred to for Riccati equations in abstract space.

By using the Kernel Theorem due to Schwartz (cf [L.71]), P(t) has the representation

$$P(t)\mathcal{G} = \int_{\mathcal{Q}} P(\omega, \omega', t) \mathcal{G}(\omega') d\omega' \qquad (4.3.32)$$

where $P(\omega, \omega', t)$ is a kernel of P(t), which is a distribution on $\Omega_{\omega} \times \Omega_{\omega'}$, defined uniquely by P(t). The generated kernel $P(\omega, \omega', t)$ can be expressed as

$$P(\omega, \omega', t) = \begin{bmatrix} \varepsilon P_1(\omega, \omega', t) & \varepsilon P_2(\omega, \omega', t) \\ \varepsilon P_2(\omega, \omega', t) & P_3(\omega, \omega', t) \end{bmatrix}, \quad (4.3.33)$$

with

$$P_{1}(\omega, \omega', t) = P_{1}(\omega', \omega, t),$$
 (4.3.34)

$$P_{3}(\omega, \omega', t) = P_{3}(\omega', \omega, t).$$
 (4.3.35)

Then the kernel corresponding to the operator P(t) characterized by Eq.(4.3.25) satisfies the following integro-differential equation of Riccati type

$$-\varepsilon \frac{\partial}{\partial t} P_{1} = (A_{1\omega} + A_{1\omega}^{*})P_{1}(\omega, \omega', t) + r^{-1}b_{1}^{2}P_{1}\circ P_{1} - w_{1}\delta(\omega - \omega')$$

$$- 2\varepsilon a_{6}P_{2} + \varepsilon^{2}r^{-1}b_{2}^{2}P_{2}\circ P_{2}, \qquad (4.3.36)$$

$$-\varepsilon \frac{\partial}{\partial t} P_{2} = A_{1\omega}^{*}P_{2} - a_{6}P_{3} + r^{-1}b_{1}^{2}P_{2}\circ P_{1} - \varepsilon A_{4\omega}P_{2} + \varepsilon r^{-1}b_{2}^{2}P_{3}\circ P_{2}, \qquad (4.3.37)$$

$$-\frac{\partial}{\partial t} P_{3} = (A_{4\omega} + A_{4\omega}^{*})P_{3} - 2a_{3}P_{2} + r^{-1}b_{1}^{2}P_{2}\circ P_{2} - w_{2}\delta(\omega - \omega'), \qquad (4.3.38)$$

where

$$A_{1\omega} = a_1 \Delta_{\omega} - a_2, \qquad A_{4\omega} = a_4 \Delta_{\omega} - a_5$$

PoQ(ω , ω' , t) = $\int_{\Omega} P(\omega, \sigma, t)Q(\sigma, \omega', t)d\sigma$, (4.3.39)

with the terminal and boundary conditions

$$P_{i} = 0, \quad \text{on } \partial(\Omega \times \Omega), \qquad (4.3.40)$$

$$P_{i}(\omega, \omega', T) = 0.$$
 (4.3.41)

It is to be noted that we obtained Eqs.(4.3.36) - (4.3.38) by using the relation

$$PA\phi(\omega) = \int_{\Omega} P(\omega, \omega^{\dagger}, t) A_{\omega} \phi(\omega) d\omega$$
$$= \int_{\Omega} A_{\omega}^{\star} P(\omega, \omega^{\dagger}, t) \phi(\omega) d\omega,$$

hence the kernel of PA is $A^*_{\omega}, P(\omega, \omega', t)$. It is added that the kernel of the identity is $\delta(\omega - \omega')$.

4.3.2 Eigenfunction expansions

By using Theorem 4-2, there exists an orthonormal system of eigenfunctions satisfying

$$- A_{1\omega} \phi_{j}(\omega) = \lambda_{j} \phi_{j}(\omega), \qquad (4.3.42)$$

with

$$\phi_{j}|_{\Gamma} = 0,$$
 (4.3.43)

where

$$A_{1\omega} = a_1 \Delta_{\omega} - a_2.$$

Then it can be shown that approximations to the state vector z and the costate vector s may be represented to be of the form,

$${}_{m}^{z} = \sum_{j=1}^{m} z_{j}(t)\phi_{j}, \qquad {}_{m}^{s} = \sum_{j=1}^{m} s_{j}(t)\phi_{j}, \qquad (4.3.44)$$

$$z_{j} \in L^{2}(0,T), \qquad s_{j} \in L^{2}(0,T), \qquad (4.3.45)$$

$$\sum_{j=1}^{\infty} |\lambda_{j}| \int_{0}^{1} ||z_{j}(t)||^{2} dt < \infty, \qquad (4.3.46)$$

$$\sum_{j=1}^{\infty} |\lambda_{j}| \int_{0}^{T} ||s_{j}(t)||^{2} dt < \infty.$$
(4.3.47)

These approximations $_{m}^{z}$ and $_{m}^{s}$ have the properties

$$\lim_{m} z = z, \quad \ln L^{2}(\Omega), \quad (4.3.48)$$

$$\lim_{m \to \infty} s = s, \text{ in } L^2(\Omega). \qquad (4.3.49)$$

An approximation to the kernels $P_i(\omega, \omega', t)$ may also be represented as

$$mP_{i}(\omega,\omega', t) = \sum_{k,l=1}^{m} P_{i}^{kl} \phi_{k}(\omega) \phi_{l}(\omega'), \qquad (4.3.50)$$

$$P_i^{k1} \in L^2(0,T).$$
 (4.3.51)

It is to be noted that the sequence $\{\phi_k(\cdot)\phi_1(\cdot)\}_{k,1=1,2,\ldots}$ constitutes an orthonormal basis in $L^2(\Omega \times \Omega)$ and $\underset{m}{P_i}$ has also the property

$$\lim_{m \to \infty} {}_{m} P_{i} = P_{i}, \quad \text{in } L^{2}(\Omega \times \Omega). \quad (4.3.52)$$

If we set

$$u_{j} = (u, \phi_{j}),$$
 (4.3.53)

$$z_{0j} = (z_0, \phi_j),$$
 (4.3.54)

and

$$z_{dj} = (z_d, \phi_j),$$
 (4.3.55)

then, we obtain, substituting Eq.(4.3.44) into Eqs.(4.3.3) and (4.3.18):

$$I_{\varepsilon dt} z_{j} + A_{j} z_{j} = Bu_{j}, \qquad (4.3.56)$$

$$- I_{\varepsilon dt} s_{j} + A_{j} s_{j} = Wz_{j} - z_{dj}, \qquad (4.3.57)$$

where

$$A_{j} = \begin{bmatrix} \lambda_{j} & a_{3} \\ a_{6} & \alpha_{1}\lambda_{j} + \alpha_{2} \end{bmatrix}, \quad \alpha_{1} = \frac{a_{2}a_{4}}{a_{1}}, \quad \alpha_{2} = -a_{5} + \frac{a_{2}a_{4}}{a_{1}}.$$

The performance index (4.3.12) can also be decoupled into each mode by using the orthonormality of the basis $\{\phi_i(\omega)\}$, such that

$$J = \sum_{j=1}^{\infty} J_{j}, \quad J_{j} = \int_{0}^{T} ||z_{j} - z_{dj}||^{2} dt + \int_{0}^{T} (u_{j}, Ru_{j}) dt. \quad (4.3.58)$$

It is to be noted that J_j includes only variables with suffix j. Then we can synthesize the decoupled modal control. The property of the decoupled modal control ascertains that the time coefficients $P_i^{kl}(t)$ in Eq.(4.3.50) are represented by the form

$$P_{i}^{kl}(t) = \delta_{kl} P_{i}^{kl}(t), \qquad (4.3.59)$$

where δ_{k1} denotes Kronecker's delta, i.e., we derive the following expression in place of Eq.(4.3.50):

$${}_{m}P_{i}(\omega, \omega', t) = \sum_{k=1}^{m} P_{i}^{kk} \phi_{k}(\omega) \phi_{k}(\omega'). \qquad (4.3.60)$$

It is clear that $P_i^{kl}(t)$ is equivalent to the solution $P_i^{k}(t)$, i = 1, 2, 3, of the Riccati equation derived for the k-th mode canonical equation with the k-th mode performance index J_k . Considering the situations described above, we give the matrix representation for $\{P_i^{kl}(t)\}$ i = 1, 2, 3, as

$$-\varepsilon \frac{d}{dt} P_{1} = 2A_{1}P_{1} + r^{-1}b_{1}^{2}P_{1}P_{1} - w_{1} - 2\varepsilon a_{6}P_{2} + \varepsilon^{2}r^{-1}b_{2}^{2}P_{2}P_{2},$$

$$-\varepsilon \frac{d}{dt} P_{2} = A_{1}P_{2} - a_{6}P_{3} + r^{-1}b_{1}^{2}P_{2}P_{1} - \varepsilon A_{4}P_{2} + \varepsilon r^{-1}b_{2}^{2}P_{3}P_{2},$$

$$-\frac{d}{dt} P_{3} = 2A_{4}P_{3} - 2a_{3}P_{2} + r^{-1}b_{2}^{2}P_{3}P_{3} + r^{-1}b_{1}^{2}P_{2}P_{2} - w_{2},$$

$$(4.3.62)$$

$$(4.3.63)$$

where A_i , i = 1, 4 and P_i , i = 1, 2, 3, are of the diagonal form

$$A_{1} = \begin{bmatrix} \lambda_{1} & 0 \\ 2 & \ddots \\ 0 & \lambda_{m} \end{bmatrix}, \quad A_{4} = \begin{bmatrix} \alpha_{1}\lambda_{1}^{+\alpha}2_{\alpha}\lambda_{2}^{+\alpha}2 & 0 \\ 0 & \ddots & \alpha_{1}\lambda_{m}^{+\alpha}2 \end{bmatrix},$$
$$P_{i} = \begin{bmatrix} P_{i}^{11} & 0 \\ P_{i} & \ddots & P_{i}^{mm} \\ 0 & \ddots & P_{i}^{mm} \end{bmatrix}$$

Thus the Riccati equation for the mode time coefficients $P_i^{kl}(t)$ can be separated into each mode Riccati equation and the generated Riccati equation of the individual mode can be treated independently of those of other modes. The Riccati equation of the k-th mode is represented as

$$-\varepsilon \frac{d}{dt} P_1^{kk} = 2\lambda_k P_1^{kk} + r^{-1} b_1^2 (P_1^{kk})^2 - w_1 - 2\varepsilon a_6 P_2^{kk}$$

+
$$\varepsilon^2 r^{-1} b_2^2 (P_2^{kk})^2$$
, (4.3.64)

$$-\varepsilon \frac{d}{dt} P_2^{kk} = \lambda_k P_2^{kk} - a_6 P_3^{kk} + r^{-1} b_1^2 P_2^{kk} P_1^{kk} - \varepsilon (\alpha_1^{\lambda} \lambda_k + \alpha_2) P_2^{kk} + \varepsilon r^{-1} b_2^2 P_3^{kk} P_2^{kk}, \qquad (4.3.65)$$

$$-\frac{d}{dt}P_{3}^{kk} = 2(\alpha_{1}\lambda_{k} + \alpha_{2})P_{3}^{kk} - 2a_{3}P_{2}^{kk} + r^{-1}b_{2}^{2}(P_{3}^{kk})^{2} + r^{-1}b_{1}^{2}(P_{2}^{kk})^{2} - w_{2}, \qquad (4.3.66)$$

with the terminal conditions

$$P_i^{KK}(T) = 0, \quad i = 1, 2, 3.$$
 (4.3.67)

4.3.3 Asymptotic expansions via boundary layer method

As in the lumped parameter case, we are naturally led to constructing solutions of Eqs.(4.3.36) - (4.3.38) of the form

$$P(\omega, \omega', t) = K(\omega, \omega', t) + H(\omega, \omega', \tau), \qquad (4.3.68)$$

where $\tau = (T-t)/\varepsilon_{j}$ and $K(\omega, \omega', t)$ and $H(\omega, \omega', \tau)$ admit the following asymptotic expansions in ε as ε tends to zero:

$$K(\omega, \omega', t) = \sum_{r=0}^{\infty} K^{r}(\omega, \omega', t) \varepsilon^{r}(r!)^{-1}, \qquad (4.3.69)$$

$$H(\omega, \omega', \tau) = \sum_{r=0}^{\infty} H^{r}(\omega, \omega', \tau) \varepsilon^{r}(r!)^{-1}, \qquad (4.3.70)$$

and

$$K^{r}(\omega, \omega', t) = K^{r}(\omega', \omega, t),$$
 (4.3.71)

$$H^{r}(\omega, \omega', \tau) = H^{r}(\omega', \omega, \tau).$$
 (4.3.72)

The results of the preceding section, i.e., availability of the decoupled modal control, the kernel corresponding to the "slow mode" (see Section 1.4), $K(\omega, \omega', t)$ and the "fast mode" kernel $H(\omega, \omega', \tau)$, can be expanded into the following eigenfunction expansions:

$$K(\omega, \omega', t) = \sum_{k,l=1}^{\infty} \delta_{kl} K^{kl}(t) \phi_{k}(\omega) \phi_{l}(\omega'), \qquad (4.3.73)$$

$$H(\omega, \omega', \tau) = \sum_{k,1=1}^{\infty} \delta_{k1} H^{k1}(\tau) \phi_{k}(\omega) \phi_{1}(\omega'). \qquad (4.3.74)$$

As to the coefficient kernel of ϵ^{r} in Eqs.(4.3.69) and (4.3.70), the results analogous to Eqs.(4.3.73) and (4.3.74) hold. Thus the coefficient kernels of the outer expansion $K^{r}(\omega, \omega', t)$ and those of the boundary layer corrections $H^{r}(\omega, \omega', \tau)$ can be approximated by

$$\mathbf{m}^{\mathbf{K}^{\mathbf{r}}}(\omega, \omega', t) = \sum_{k=1}^{m} \mathbf{K}^{\mathbf{r}, kk}(t) \phi_{k}(\omega) \phi_{k}(\omega'), \qquad (4.3.75)$$

$${}_{m}H^{r}(\omega, \omega', \tau) = \sum_{k=1}^{m} H^{r,kk}(\tau)\phi_{k}(\omega)\phi_{k}(\omega'), \qquad (4.3.76)$$

where ${}_{m}K^{r}(\omega,\;\omega^{\prime},\;t)$ and ${}_{m}H^{r}(\omega,\;\omega^{\prime},\;\tau)$ have the property

$$\lim_{m\to\infty} K^{\mathbf{r}}(\cdot, \cdot, t) = K^{\mathbf{r}}(\cdot, \cdot, t), \text{ in } L^{2}(\Omega \times \Omega), \quad (4.3.77)$$

$$\lim_{m\to\infty} H^{\mathbf{r}}(\cdot, \cdot, \tau) = H^{\mathbf{r}}(\cdot, \cdot, \tau), \text{ in } L^{2}(\Omega \times \Omega). \quad (4.3.78)$$

The above derivations are founded on the construction of the solutions of Eqs.(4.3.64) - (4.3.66) with use made of the boundary layer method developed in the lumped parameter case. We seek the solutions of Eqs.(4.3.64) - (4.3.66) with the terminal conditions (4.3.67) of the form

$$P_{i}^{kk}(t) = K_{i}^{kk}(t) + H_{i}^{kk}(\tau), \qquad (4.3.79)$$

where $K_i^{kk}(t)$ and $H_i^{kk}(\tau)$ admit the asymptotic expansions in ε as ε tends to zero,

$$K_{i}^{kk}(t) = \sum_{r=0}^{\infty} K_{i}^{r,kk}(t) \varepsilon^{r}(r!)^{-1}, \qquad (4.3.80)$$

$$H_{i}^{kk}(\tau) = \sum_{r=0}^{\infty} H_{i}^{r,kk}(\tau) \varepsilon^{r}(r!)^{-1}, \qquad (4.3.81)$$

the suffix i indicates the position of the element in the original

matrices such as

$$K = \begin{bmatrix} \varepsilon K_1 & \varepsilon K_2 \\ & & \\ \varepsilon K_2 & & K_3 \end{bmatrix}, \quad H = \begin{bmatrix} \varepsilon H_1 & \varepsilon H_2 \\ & & \\ \varepsilon H_2 & & H_3 \end{bmatrix}. \quad (4.3.82)$$

In order to determine the coefficients $K_i^{r,kk}(t)$ and $H_i^{r,kk}(\tau)$, the same procedure can be adopted as in the lumped parameter case. The Riccati equation (4.3.64) - (4.3.66) gives the recursive set of equations of the outer expansion, when we substitute the outer expansion (4.3.80) into Eqs.(4.3.64) - (4.3.66). The construction of the boundary layer corrections is carried out by stretching the Riccati equation (4.3.64) - (4.3.66) and by substituting Eq.(4.3.79) into them with consideration given to Eqs.(4.3.80) and (4.3.81). And the generated recursive set of equations determines each coefficient of the boundary layer corrections $H_i^{r,kk}(\tau)$. We have shown here the complete applicability of the method derived in Chaper 2.

4.3.4 Main theorem

The object of this section is to establish the theorem giving the asymptotic accuracy of the resulting series solutions derived in the preceding section. First we state the following important condition:

C1. The Cauchy problem

$$\frac{\partial \mathbf{u}}{\partial t} - \mathbf{A}_{1\omega} \mathbf{u} = 0, \quad \mathbf{A}_{1\omega} = \mathbf{a}_{1\omega}^{\Delta} - \mathbf{a}_{2}$$
(4.3.83)

is uniformly well-posed in the sense of Hadamard, and the semigroup $exp [tA(\sigma)]$ of this problem satisfies

$$\|\exp[tA_1(\sigma)]\| \leq Cexp(-\delta t), \text{ for } \delta > 0.$$
(4.3.84)

In this case the inequality(4.3.84) means that

 $a_1 > 0, a_2 - a_1 > 0$ (4.3.85)

holds for $t \in [0,T]$.

We are led to the next condition, which considers the boundary layer controllability in the lumped parameter case.

C2. The boundary layer system associated with Eqs.(4.3.1) and (4.3.2)

$$\frac{\partial}{\partial \tau} x - (a_1 \Delta - a_2) x + a_3 y = b_1 u_1$$
(4.3.86)

is controllable, i.e.

$$b_1 \neq 0.$$
 (4.3.87)

This definition of the boundary layer controllability can be extended easily to the case where x is a state vector of multidimensional. Then we can offer the following main theorem.

<u>Theorem 4-6</u> If conditions C1 and C2 and other specified conditions in Sections 4.3.2. and 4.3.3 hold, then there exist $\varepsilon_0 > 0$ and functions Rⁿ_i(t, ε) bounded uniformly in the t-interval [0,T] such that

$$\mathbf{R}_{i}^{\mathbf{P}}(\omega,\omega',t) = \sum_{\mathbf{r}=0}^{n} [\mathbf{K}_{i}^{\mathbf{r}}(\omega,\omega',t) + \mathbf{H}_{i}^{\mathbf{r}}(\omega,\omega',\tau)] \boldsymbol{\varepsilon}^{\mathbf{r}}(\mathbf{r}!)^{-1} + \mathbf{R}_{i}^{\mathbf{n}}(t,\boldsymbol{\varepsilon})\boldsymbol{\varepsilon}^{\mathbf{n}+1},$$

$$(4.3.88)$$

$$\mathbf{K}_{i}^{\mathbf{r}}(\omega,\omega',t) = \sum_{k=1}^{\infty} \mathbf{K}_{i}^{\mathbf{r}}, \mathbf{k}^{\mathbf{k}} \boldsymbol{\phi}_{k}(\omega) \boldsymbol{\phi}_{k}(\omega'), \quad \mathbf{H}_{i}^{\mathbf{r}}(\omega,\omega',\tau) = \sum_{k=1}^{\infty} \mathbf{H}_{i}^{\mathbf{r}}, \mathbf{k}^{\mathbf{k}} \boldsymbol{\phi}_{k}(\omega) \boldsymbol{\phi}_{k}(\omega'),$$

for $\varepsilon \in [0,\varepsilon_0]$ where $K_i^{r,kk}(t)$ and $H_i^{r,kk}(\tau)$ are determined stepwise by solving the recursive set of equations derived by using the procedure described in Section 4.3.3.

The proof of this theorem can be carried out in line with the proofs made in Theorem 4-5. We state the following lemmas verified in the course of the derivation of the eigenfunction expansion of the kernels $K(\omega, \omega', t)$ and $H(\omega, \omega', \tau)$, as shown in Eqs.(4.3.73) and (4.3.74).

Lemma 4-6 Under all the conditions specified in the preceding

Sections 4.3.2 and 4.3.3, the kernels $K(\omega, \, \omega', \, t)$ and $H(\omega, \, \omega', \, \tau)$ can be expanded into

$$K(\omega, \omega', t) = \sum_{k=1}^{\infty} \kappa^{kk}(t) \phi_{k}(\omega) \phi_{k}(\omega'), \qquad (4.3.89)$$

$$H(\omega, \omega', \tau) = \sum_{k=1}^{\infty} H^{kk}(\tau) \phi_k(\omega) \phi_k(\omega'), \qquad (4.3.90)$$

where $K^{kk}(t)$ and $H^{kk}(\tau)$ are derived by

$$K^{kk}(t) = \int_{\Omega \times \Omega} \phi_{k}(\omega) K(\omega, \omega', t) \phi_{K}(\omega') d\omega d\omega' \qquad (4.3.91)$$

$$H^{kk}(\tau) = \int_{\Omega \times \Omega} \phi_{k}(\omega) H(\omega, \omega', \tau) \phi_{k}(\omega') d\omega d\omega' \qquad (4.3.92)$$
$$K^{kk} \in L^{2}(0,T), \quad H^{kk} \in L^{2}(0,\infty).$$

<u>Proof</u> Theorem 4-2 also holds in the case of $L^2(\Omega \times \Omega)$ instead of $L^2(\Omega)$ with adequate modifications. (see Phillipson [Ph.71]) Q.E.D.

The truncation series of (4.3.75) and (4.3.76) up to the finite terms gives the so-called Galerkin approximation scheme.

<u>Lemma 4-7</u> If the conditions C1 and C2 and other specified conditions in Section 4.3.2 and 4.3.3 hold, then there exist $\varepsilon_0 > 0$ and functions Sⁿ(t, ε) bounded uniform! in t-interval considered such that

$${}^{n}P^{kk}(t) = {}^{n}K^{kk}(t) + {}^{n}H^{kk}(\tau) + S^{n}(t,\varepsilon)\varepsilon_{k}^{n+1}$$

$$= \sum_{\substack{r=0\\r=0}}^{n}K^{r,kk}(t)\varepsilon_{k}^{r}(r!)^{-1} + \sum_{r=0}^{n}H^{r,kk}(\tau)\varepsilon_{k}^{r}(r!)^{-1}$$

$$= \sum_{\substack{r=0\\r=0}}^{n}[K^{r,kk}(t) + H^{r,kk}(\tau)]\varepsilon_{k}^{r}(r!)^{-1} + S^{n}(t,\varepsilon)\varepsilon_{k}^{n+1},$$
(4.3.93)

where

$$\varepsilon_{\mathbf{k}} = \varepsilon/\mathbf{k}^2, \ \varepsilon \in [0, \varepsilon_0].$$

<u>Proof</u> The singular perturbation theory of the lumped parameter case can be applied to the ordinary differential equations of the

Riccati type (4.3.64) - (4.3.66). As in Chapter 2, we can construct solutions of Eqs.(4.3.64) - (4.3.66) of the form

$$P_{i}^{kk}(t) = \tilde{K}_{i}^{kk}(t) + \tilde{H}_{i}^{kk}(\tau), \qquad (4.3.94)$$

where $\tilde{K}_{i}^{kk}(t)$ and $\tilde{H}_{i}^{kk}(\tau)$ admit asymptotic expansions in ε as ε tends to zero (see the proof of Lemma 4.2)

$$\tilde{K}_{i}^{kk}(t) = \sum_{r=0}^{\infty} \tilde{K}_{i}^{r,kk}(t) \varepsilon^{r}(r!)^{-1}, \qquad (4.3.95)$$

$$\widetilde{H}_{i}^{kk}(\tau) = \sum_{r=0}^{\infty} \widetilde{H}_{i}^{r,kk}(\tau) \varepsilon^{r}(r!)^{-1}.$$
(4.3.96)

The coefficients $\tilde{K}_{i}^{kk}(t)$ and $\tilde{H}_{i}^{kk}(\tau)$ are identical with those derived in Section 4.3.3, which completes the proof. Q.E.D.

The connection of Lemma 4-6 with Lemma 4-7 completes the proof of Theorem 4-6.

We restrict ourself to the investigation of the Riccati equation (4.3.36) - (4.3.38), and the same procedure adopted in the proof of the main theorem can be applied to the associated parabolic equation (4.3.29) by using the eigenfunction expansions as

$$\mathbf{r}(\omega, \mathbf{t}) = \sum_{j=1}^{\infty} \mathbf{r}_{j}(\mathbf{t})\phi_{j}(\omega). \qquad (4.3.97)$$

With regard to the construction of coefficients $r_j(t)$, the result of Section 4.2 may be referred to.

Chapter 5 Nuclear Reactor Control with Distributed Parameters

5.1 Introduction

This chapter treats the nuclear reactor control involving spatially distributed parameters. The importance of this treatment has been increasing, since the size of reactors is becoming larger and the spatial effects cannot be neglected. For instance, control of the neutron flux distribution, burnup control with consideration of hotspot-factor, Xenon oscillation, etc. can be treated by considering distributed parameters.

Many studies have been made on the subject via various approaches, i.e., modal expansion method with use made of various types of mode; clean reactor mode, Kaplan mode(natural mode), lambda mode, etc., nodal expansion method, function space method, dynamic programming, and so forth. For a more precise survey, Iwazumi and Koga [I.73], or references listed in it, may be referred to.

The method proposed in this chapter is the singular perturbation theory connected with one of the modal expansion method, by using the Helmholtz mode. The Helmholtz mode gives the complete base in the space L^2 . The theoretical foundation, upon which the validity of the method presented relies, is given in the preceding chapter.

It is to be noted that the criticality of the reactor considered violates the asymptotic stability of the equation for the prompt neutron. Hence we can not neglect time derivative term of the prompt neutron in the original partial differential equation. But the resulting equation using the modal expansion can be reduced in dimensionality except for the fundamental mode. Such a situation shows that the dimension of the dynamics governing the spatial control can be reduced and that of the power control cannot, since the power control is dominated by the fundamental mode and the spatial control is by the higher mode (see Wiberg [Wi.67]). Thus the singular perturbation theory is effectively applied to the spatial control system.

The system described by multi-group diffusion equations with multigroup of delayed neutrons has a different aspect. In this case, it is possible to reduce the dimensionality of this system in both power control and spatial control. Considerations of the situation are found in Sections 5.2.3 and 5.3.2.

5.2 Formulation of the problem

5.2.1 Derivation of the state equation

We consider here an infinite slab reactor described by the following set of equations, the one-group diffusion equation with one delayed neutron group:

$$\frac{1}{v} \frac{\partial}{\partial t} \psi(\omega, t) = \frac{\partial}{\partial \omega} [D(\omega) \frac{\partial}{\partial \omega} \psi(\omega, t)] - \Sigma_{a}(\omega) \psi(\omega, t) + (1 - \beta) v \Sigma_{f}(\omega) \psi(\omega, t) + \lambda C - \Sigma_{c}(\omega, t) \psi(\omega, t),$$

$$\frac{\partial}{\partial t} C(\omega, t) = \beta v \Sigma_{f}(\omega) \psi(\omega, t) - \lambda C(\omega, t),$$
(5.2.2)

where ω is the spatial coordinate, $\psi(\omega,t)$ is neutron flux, $C(\omega,t)$ is precursor density, and $\sum_{c} (\omega,t)$ is the absorption cross section which is used for control variable, and assumed to be distributed in this case for simplicity. This system is bilinear form as in the lumped parameter model dealt with in Chapter 3. Hence linearization is made as in Chapter 3 by expanding each variable about the initial steady state.

The spatially dependent coefficient $D(\omega)$ prevents usfrom constructing analytical eigenfunctions. The approximation made to avoid this trouble, is the separation of the coefficient $D(\omega)$ into two parts, spatially variant and invariant parts. To ensure the finality of the expansion series (see Kaplan [Ka.61]), the other coefficients are also separated in the same way. The spatially independent parts should be chosen so that the fundamental mode may sustain the initial

steady state and the effects of the spatially variant parts are taken into the control as the initial distribution of the control. This approximation was proposed by Iwazumi and Koga [I.73]. If the system behaves well in the course of motion, the approximation gives satisfactory results. The validity depends upon the property of the problem considered. In this chapter, we consider the tracking problem which shall give the validity, i.e., the desired state is given not so far from the initial state.

The linearized state equation can be obtained by considering small deviations from the steady fundamental mode, which is determined by the system at the initial time involving only spatially invariant parameters, and by normalizing with respect to the fundamental mode of neutron flux ψ_{10} , as

$$I_{\varepsilon} \frac{\partial}{\partial t} \Phi(\omega, t) = S_{\omega} \Phi(\omega, t) + Bu(\omega, t), \qquad (5.2.3)$$

where

$$\Phi(\omega, t) = \begin{bmatrix} \delta \psi(\omega, t)/\psi_{10} \\ \delta C(\omega, t)/\psi_{10} \end{bmatrix}, \quad u = -\psi(\omega, t)\Sigma_{c}(\omega, t)/\psi_{10},$$

$$S_{\omega} = \begin{bmatrix} a_{1}\partial^{2}/\partial\omega^{2} + a_{2} & a_{3} \\ a_{4} & a_{5} \end{bmatrix}, \quad I_{\varepsilon} = \begin{bmatrix} \varepsilon & 0 \\ 0 & 1 \end{bmatrix}, \quad B = \begin{bmatrix} 1 \\ 0 \end{bmatrix},$$

$$(5.2.4)$$

$$a_{1} = D, \quad a_{2} = (1 - \beta)\Sigma_{f} - \Sigma_{a}, \quad a_{3} = \lambda,$$

$$a_{4} = \beta \Sigma_{f}, \quad a_{5} = -\lambda]\varepsilon = v^{-1}.$$

The resulting equation (5.2.3) is of the same form that Kuroda and Makino [Kd.69] treated. The difference lies in that Kuroda and Makino first assumed the spatial homogenity of the system parameters, but the treatment presented takes the effects of spatially varying parameters into consideration.

The problem to be considered is the terminal cost problem to minimize the following performance index:

$$J = \int_{0}^{\ell} [\Phi(\omega, T) - \Phi_{d}(\omega)]' Q[\Phi(\omega, T) - \Phi_{d}(\omega)]d\omega + \int_{0}^{T} \int_{0}^{\ell} r[u(\omega, t) - u_{0}(\omega)]^{2} d\omega dt, \qquad (5.2.5)$$

where

$$Q = \begin{bmatrix} q_1 & 0 \\ 0 & q_3 \end{bmatrix}; q_1, q_3, r > 0,$$

 $\Phi_{d}(\omega)$ is the desired state, and $u_{0}(\omega)$ is the initial value of the control determined beforehand so that the spatial homogenity of the system parameters holds. As to the construction of this initial value, Iwazumi and Koga [I.73] may be referred to.

5.2.2 Modal expansion

In the case of one dimensional space, i.e., an infinite slab reactor, the Helmholtz mode is defined by

$$\frac{\mathrm{d}^2}{\mathrm{d}\omega^2}\phi + \lambda\phi = 0, \qquad (5.2.6)$$

with

$$\phi(0) = 0, \quad \phi(\ell) = 0, \quad (5.2.7)$$

where ${\tt l}$ is an extrapolated thickness, ${\tt \lambda}_{i}$ eigenvalue, and ${\tt \phi}_{i}$ eigenfunction;

$$\phi_{i} = \sqrt{2/\ell} \sin \sqrt{\lambda_{i}} \omega, \qquad (5.2.8)$$

$$\lambda_{i} = (\frac{i\pi}{\ell})^{2}.$$
 (5.2.9)

The sequence $\{\phi_i\}$ constitutes an orthonormal set in L^2 , i.e.,

$$\int_{0}^{\ell} \phi_{i}(\omega)\phi_{j}(\omega)d\omega = \delta_{ij}, \qquad (5.2.10)$$

where δ_{ij} is the Kronecker's delta.

If the existence of the solution has been proved, it can be shown that the system $\{\sqrt{2/\ell} \sin\sqrt{\lambda i} \omega\}$ can be replaced by other suitable orthonormal systems chosen in infinite ways (see Green [Gr.53]).

The state vector Φ can be expanded into infinite series with respect to the eigenfunctions generated above,

$$\Phi(\omega,t) = \sum_{i=1}^{\infty} \Phi_i(t) \sqrt{2/\ell} \sin \frac{i\pi}{\ell} \omega, \qquad (5.2.11)$$

or

The control rods are physically concentrated, which should be represented mathematically by the Dirac measure δ . Then the control term is given by

$$u(\omega,t) = \sum_{k=1}^{M} u_k(t) \delta(\omega - \omega_k), \qquad (5.2.12)$$

where M is the number of control rods and ω_k is the position of the k-th control rod. In the case where there are lots of control rods, however, the spatially concentrated control can be represented approximately by the spatially distributed control, which is simpler to analyze. The spatially concentrated case will be considered later. Here we restrict ourself to the spatially distributed case for simplicity.

The control term $u(\omega,t)$ can be expanded with respect to the eigenfunction of the Helmholtz mode, if the control $u(\cdot,t)$ is given in $L^{2}(\Omega)$.

$$u(\omega,t) = \sum_{i=1}^{\infty} u_i(t) \sqrt{2/\ell} \sin \frac{i\pi}{\ell} \omega . \qquad (5.2.13)$$

In order to obtain a set of equations for the modal coefficients, $\xi_i(t)$, $n_i(t)$ and $u_i(t)$ which are functions of time, the expansions (5.2.11) and (5.2.13) are substituted into Eq.(5.2.3) and the

orthonormality (5.2.10) is used, yielding

$$\varepsilon \frac{d}{dt} \xi_{i}(t) = [-a_{1}(\frac{i\pi}{\lambda})^{2} + a_{2}]\xi_{i}(t) + a_{3}\eta_{i}(t) + u_{i}(t), \quad (5.2.14)$$

$$\frac{d}{dt} \eta_{i}(t) = a_{4}\xi_{i}(t) + a_{5}\eta_{i}(t). \qquad (5.2.15)$$

It is to be noted that the modal equations (5.2.14) and (5.2.15) are decoupled with each of the other modes, which results from the approximation made beforehand about the spatial homogenity of parameters.

To synthesize the optimal feedback control in the modal space, the performance index should be expanded into modal form. Thus the desired terminal distribution $\Phi_{d}(\omega)$ and the initial control $u_{0}(\omega)$ are expanded by using the same spatial eigenfunctions, as follows:

$$\Phi_{d}(\omega) = \sum_{i=1}^{\infty} \Phi_{di} \sqrt{2/\ell} \sin \frac{i\pi}{\ell} \omega = \sum_{i=1}^{\infty} \begin{bmatrix} \xi_{di} \\ \eta_{di} \end{bmatrix} \sqrt{2/\ell} \sin \frac{i\pi}{\ell} \omega ,$$

$$u_{0}(\omega) = \sum_{i=1}^{\infty} u_{0i} \sqrt{2/\ell} \sin \frac{i\pi}{\ell} \omega .$$
(5.2.16)

Then we obtain the modal expansion of the performance index, as

$$J = \sum_{i=1}^{\infty} \Psi'_{i}(T) Q \Psi_{i}(T) + r \int_{0}^{T} \sum_{i=1}^{\infty} \tilde{u}_{i}^{2}(t) dt$$
 (5.2.17)

where

$$\Psi_{i}(t) = \Phi_{i}(t) - \Phi_{di}(t),$$
$$\tilde{u}_{i}(t) = u_{i}(t) - u_{0i}(t).$$

Hence the performance index for the i-th mode J, becomes

$$J_{i} = \frac{1}{2} \Psi_{i}^{\prime}(T) Q \Psi_{i}^{\prime}(T) + \frac{1}{2} r \int_{0}^{T} \tilde{u}_{i}^{2}(t) dt. \qquad (5.2.18)$$

It is easily seen that optimal control theory for lumped parameter systems can be applied to each mode, since coupling between the other modes does not occur in total system including the performance index.

The state equation of the i-th mode and the corresponding i-th performance index are given in Eqs.(5.2.14), (5.2.15) and (5.2.18) respectively. Applying the well known results in optimal control theory of lumped parameter systems, the following control law is derived in the form of feedback type:

$$u_{i}^{*}(t) = -r^{-1}B'[K_{i}(t)\Phi_{i}(t) + g_{i}(t)] + u_{0i},$$
 (5.2.19)

where $K_{i}(t)$ is a solution of the matrix Riccati differential equation of the i-th mode

$$\dot{K}_{i}(t) = -K_{i}(t)A_{i} - A_{i}K_{i}(t) + r^{-1}K_{i}(t)BBK_{i}(t),$$
 (5.2.20)

and $g_i(t)$ is a solution of the following linear differential equation associated with the Riccati equation derived above:

$$g_{i}(t) = -(A_{i} - r^{-1}BB'K_{i}(t))'g_{i}(t) - K_{i}'(t)Bu_{0i},$$
 (5.2.21)

where

$$A_{i} = \begin{bmatrix} a_{1}(i\pi/\ell)^{2} + a_{2} & a_{3} \\ a_{4} & a_{5} \end{bmatrix}.$$

As the partitioning similar to Eqs.(2.2.16) and (2.2.17) holds also, then we obtain the following Riccati equations for each element,

$$\varepsilon \dot{k}_{1} = -2[-a_{1}(i\pi/\ell)^{2} + a_{2}]k_{1} + r^{-1}k_{1}^{2} - 2\varepsilon a_{4}k_{2},$$

$$\varepsilon \dot{k}_{2} = -a_{3}k_{1} - [-a_{1}(i\pi/\ell)^{2} + a_{2}]k_{2} - a_{4}k_{3} + r^{-1}k_{1}k_{2}$$

$$(5.2.22)$$

$$\varepsilon a_{5}k_{2},$$

$$\dot{k}_{3} = -2a_{3}k_{2} - 2a_{5}k_{3} + r^{-1}k_{2}^{2},$$
(t)

and for $g_i(t)$

$$\varepsilon \dot{g}_{1} = - [a_{1}(i\pi/\ell)^{2} + a_{2} - r^{-1}k_{1}]g_{1} - a_{4}g_{2} - k_{1}u_{0},$$

$$\dot{g}_{2} = - a_{3}g_{1} - a_{5}g_{2} + r^{-1}k_{2}g_{1} - k_{2}u_{0}.$$
(5.2.23)

These equations should be solved under the terminal conditions

$$K(T) = Q,$$
 (5.2.24)

$$g(T) = -Q\Phi_{d}$$
 (5.2.25)

In the above expressions of Eqs.(5.2.22) - (5.2.25) the subscript of each variable representing the modal order is omitted, and the same expressions are adopted in what follows unless confusions may occur.

5.2.3 Consideration on the criticality condition

From the fact that the decoupled modal control is applicable to the system (5.2.3), the resulting infinite set of differential equations can be treated separately. Hence the technique of constructing asymptotic expansions of solutions developed in the lumped parameter systems can be applied to the i-th mode system of ordinary differential equations respectively, within the i-th mode system.

However, the distributed parameter reactor described by the kinetic equation of one-group diffusion with one delayed neutron or multiple-delayed neutron, has a distinct situation from the one-point reactor. The fundamental mode (the first mode) should be considered separately from the other higher modes, since the criticality of the reactor requires that the eigenvalue corresponding to the first mode of the clean reactor mode should be set to zero, which sustains the steady state of the initial state under consideration. The critical condition is thus obtained in the Helmholtz mode as

$$a_{2}a_{5} - a_{3}a_{4} - a_{1}a_{5}(\pi/\ell)^{2} = 0$$
 (5.2.26)

The criticality condition is assumed to hold at the initial time, so the asymptotic stability of the boundary layer system of the first

time coefficient of the neutron flux is not concluded [Hn. 69]. So the technique of the singular perturbations cannot be applied to the first mode system. The separation of the control system into two systems, the power control system and the spatial control system, gives a physical meaning of the above situation. The contributions of the higher modes to the power control are minor, and the reactor can be controlled by one feedback system represented by the fundamental mode. And the other feedback systems detect and control the other higher modes which determine the spatial shape of the neutron flux (see Wiberg [Wi.67, P322]). Hence we can say that the power control system should be treated in the original form, and the spatial control system can be dealt with by using the singular perturbation theory. In other words, singular perturbation theory is effectively applied to the spatial control system with reduction of the dimensionality.

The stability theory of the reactor with distributed parameters are found in literatures [S.68], [Ks.69], etc. The situations considered above is consistent with the result described in Kastenberg [Ks.69], which gives an extensive survey on the asymptotic stability of distributed systems adopting several approaches, Lyapunov method, semigroup theory, etc. The theory is a flourishing field.

5.2.4 Construction of asymptotic expansions

Applying the boundary layer method developed in the lumped parameter systems, we seek solutions of the form

$$\tilde{k}_{j}(t,\varepsilon) = k_{j}(t,\varepsilon) + h_{j}(\tau,\varepsilon); \quad j = 1, 2, 3,$$
 (5.2.27)

where $\tau = (T - t)/\epsilon$, and $k_j(t,\epsilon)$, $h_j(\tau,\epsilon)$ admit asymptotic expansions in ϵ , as ϵ tends to zero of the form

$$k_{j}(t,\varepsilon) = \sum_{r=0}^{\infty} k_{j}^{r}(t)\varepsilon^{r}(r!)^{-1},$$

$$h_{j}(\tau,\varepsilon) = \sum_{r=0}^{\infty} h_{j}^{r}(\tau)\varepsilon^{r}(r!)^{-1}.$$
(5.2.28)

The subscript j represents the position of the element such that

$$K = \begin{bmatrix} \varepsilon k_1 & \varepsilon k_2 \\ \varepsilon k_2 & k_3 \end{bmatrix}, \quad H = \begin{bmatrix} \varepsilon h_1 & \varepsilon h_2 \\ \varepsilon h_2 & h_3 \end{bmatrix}. \quad (5.2.29)$$

It can be easily seen that the sequences $\sum_{ir i,j} r(t)$ and $\sum_{ir i,j} r(\tau)$ converge to the solution of the Riccati integro-differential equation and that of the stretched Riccati equation associated with it, respectively, which are obtained directly by using the Riccati-like decoupling shown in the preceding chapter.

Substituting the outer expansion $\Sigma k_j^r \varepsilon^r (r!)^{-1}$ into (5.2.22), and comparing the coefficients of like powers of ε , we have the following set of equations:

for the higher mode i > 1

$$0 = -2[-a_{1}(i\pi/\ell)^{2} + a_{2}]k_{1}^{0} + r^{-1}k_{1}^{02},$$

$$0 = -a_{4}k_{3}^{0} - [-a_{1}(i\pi/\ell)^{2} + a_{2}]k_{2}^{0} - a_{3}k_{1}^{0} + r^{-1}k_{1}^{0}k_{2}^{0}, (5.2.30)_{0}$$

$$\dot{k}_{3}^{0} = -2a_{5}k_{3}^{0} - 2a_{3}k_{2}^{0} + r^{-1}(k_{2}^{0})^{2},$$

and

$$\dot{k}_{1}^{r-1} = -2[-a_{1}(i\pi/\ell)^{2} + a_{2}]k_{1}^{r} + 2r^{-1}k_{1}^{0}k_{1}^{r} + p_{1}^{r}(t),$$

$$\dot{k}_{2}^{r-1} = -a_{3}k_{1}^{r} - [-a_{1}(i\pi/\ell)^{2} + a_{2}]k_{2}^{r} - a_{4}k_{3}^{r} + r^{-1}k_{1}^{0}k_{2}^{r} + r^{-1}k_{1}^{0}k$$

where the remainders $p_i^r(t)$, $i = 1 \sim 3$ are polynomials involving the known terms in the preceding steps.

In order to construct a uniformly valid expansion, the boundary layer corrections are considered. The recursive set of equations satisfied by the boundary layer correctors is obtained by using the stretched system as in the lumped parameter case in Section 3.2.3. The resulting recursive set of equations thus becomes

$$-\frac{d}{d\tau}h_{1}^{0} = -2[-a_{1}(i\pi/\ell)^{2} + a_{2}]h_{1}^{0} + r^{-1}h_{1}^{0}[h_{1}^{0} + k_{1}^{0}(T)],$$

$$-\frac{d}{d\tau}h_{2}^{0} = -a_{3}h_{1}^{0} - [-a_{1}(i\pi/\ell)^{2} + a_{2}]h_{2}^{0} - a_{4}h_{3}^{0}$$

$$+ r^{-1}[h_{1}^{0}h_{2}^{0} + h_{1}^{0}k_{2}^{0}(T) + k_{1}^{0}(T)h_{2}^{0}],$$

$$-\frac{d}{d\tau}h_{3}^{0} = 0.$$
(5.2.31)

For the higher order

$$-\frac{d}{d\tau}h_{1}^{r} = -2[-a_{1}(i\pi/\ell)^{2} + a_{2}]h_{1}^{r} + r^{-1}h_{1}^{0}[h_{1}^{r} + k_{1}^{r}(T)] + p_{1}^{r}(\tau),$$

$$-\frac{d}{d\tau}h_{2}^{r} = -a_{3}h_{1}^{r} - [-a_{1}(i\pi/\ell)^{2} + a_{2}]h_{2}^{r} - a_{4}h_{3}^{r}$$

$$+ r^{-1}h_{1}^{0}[h_{2}^{r} + k_{2}^{r}(T)] + r^{-1}h_{2}^{0}[h_{1}^{r} + k_{1}^{r}(T)] + p_{2}^{r}(\tau),$$

$$-\frac{d}{d\tau}h_{3}^{r} = p_{3}^{r}(\tau),$$
(5.2.31)

where the remainders p_i^r consist only of functions decaying exponentially. Hence the terminal condition of the first boundary layer correction $h_3^1(0)$ can be determined by the following semi-infinite integral formula:

$$h_3^{1}(0) = \int_0^\infty (-2a_3h_2 - 2a_5h_3)d\tau.$$
 (5.2.32)

The integrand involves only the terms determined in the O-order boundary layer equation $(5.2.31)_0$ with the terminal condition derived by using the given data by setting ε to zero. Thus the terminal condition of the first coefficient of the outer expansion $k_3^r(t_f)$ can be derived by using the followng relation:

$$k_{i}(T) + h_{i}(0) = q_{i}; i = 1, 2, 3,$$
 (5.2.33)

where q_i are given in Eq.(5.2.24). Hence we obtain

$$k_{3}^{1}(T) = l_{3}^{-1} \int_{0}^{\infty} (-2a_{3}h_{2} - 2a_{5}h_{3})d\tau,$$
 (5.2.34)

under which the first correction system of the outer expansion can be solved thoroughly. Then we calculate the value of each coefficient of the first outer expansion at the terminal time $t = t_f$. The values of the boundary layer correctors at the terminal time $\tau = 0$ directly follow by using the relation (5.2.33),

$$h_1^{1}(0) = q_1 - k_1^{1}(T), \quad h_2^{1}(0) = -k_2^{1}(T).$$
 (5.2.35)

Thus we can carry out the computation of the first boundary layer correctors given in Eq. $(5.2.31)_1$.

The above procedure determines the total system of the reduced and first order completely. A similar result can be derived by using the same algorithm, for the higher order systems.

As to the linear differential equation for g(t), the same procedure can be applied as in Section 2.2.

The practical computation can not be carried out with regard to the infinite terms of the modal expansion, hence the truncated series is adopted at finite N-th mode. The suboptimum distributed control is given by

$$u_{sub}(\omega,t) = \sum_{i=1}^{N} u_i^{*}(t) \sqrt{2/\ell} \sin \frac{i\pi}{\ell} \omega, \qquad (5.2.36)$$

and the suboptimum trajectory, spatially dependent, is obtained by

$$\Phi_{sub}(\omega,t) = \sum_{i=1}^{N} \Phi_{i}(t) \sqrt{2/\ell} \sin \frac{i\pi}{\ell} \omega, \qquad (5.2.37)$$

where the optimal trajectory Φ_i of the i-th mode can be derived as a solution of the i-th mode trajectory equation

$$\varepsilon \dot{\xi}_{i} = [-a_{1}(i\pi/\ell)^{2} + a_{2} - r^{-1}k_{i1}]\xi_{i} + a_{3}n_{i} - r^{-1}g_{i1} + u_{0i},$$
(5.2.38)

$$n_{i} = a_{4}\xi_{i} + a_{5}n_{i},$$
 (5.2.39)

with the initial conditions

$$\xi_{i}(0) = [-a_{1}(i\pi/\ell)^{2} + a_{2} - a_{4}/a_{5}]^{-1} u_{0_{i}},$$

$$\eta_{i}(0) = -a_{4}/a_{5} \cdot \xi_{i}(0),$$
(5.2.40)

which can be obtained by equating the differential terms in Eqs.(5.2.14) and (5.2.15) to zero. The above procedure to compute the initial condition (5.2.40) is reasonable from the requirement of the steady state sustenance at the initial time. Equations (5.2.38) and (5.2.39) can be treated approximately by the singular perturbation theory as shown in Section 2.2.3.

5.2.5 Numerical example

The results obtained by numerical computations are presented. The data used in this example are listed in Table 5.2.1.

λ	0.078	sec	l	100	cm
Σ_{f}	0.0202	-1 cm	Т	10	sec
$\Sigma_{\mathbf{a}}$	0.05	cm ⁻¹	r	1000	
D.	0.5071	cm	9 ₁	1	
A	2.2 × 10 ⁵	cm/sec	^q 3	78	

Table 5.2.1 Data used in the example.

Figure 5.2.1 shows various time behaviours of the first mode: the elements of the feedback coefficient matrix, g_i , viz. the solution of the equation associated with the Riccati equation, and the amplitude of the neutron flux. This figure presents the exact solutions derived by using Runge-Kutta-Gill method.

Figure 5.2.2 presents the time behaviours of the elements of the



Fig. 5.2.1 First mode.









feedback coefficients matrix of the second to fourth mode; each figure shows both the exact and approximate solutions. The approximate solutions are derived of the form of truncated series at the second term.

As mentioned in the lumped parameter case, the feedback coefficients of each mode change rapidly in the neighbourhood of the terminal time, and computational difficulties should arise. The difficulties are much more serious in distributed parameter cases, because many modes equations must be treated and the existence of the boundary layers appearing in every mode makes the change more rapid. The present method can make us avoid the difficulties in the same way as in Section 3.2 except the fundamental mode.

The approximation to each feedback coefficient has as good accuracy as expected except k_3 of the second mode. But the error of k_3 of the second mode is of the order of 10^{-14} which is consistent with the theoretical results in the preceding sections. The order of k_3 itself is small relatively. Such situations show that the approximations except k_3 of the second mode is particularly good in this example.

Figure 5.2.3 shows the time behaviours of g_i of the second to fourth mode. These results for g_i do not give so good approximation as those for k_i . The cause of this situation is that the order of g_i itself is not sufficiently large compared with the order of the perturbing parameter ε . The truncated series approximation at the second term seems to be insufficient in appearance; but it is good enough from the practical point of view, considering the small effects of the higher modes to the motion considered.

In Fig. 5.2.4, the time behaviours of the amplitudes of the neutron flux of the higher modes are illustrated. It can be seen in this figure that the error of g_i of each higher mode is compensated.

Figure 5.2.5 shows the exact and approximate transients of the neutron flux shape.

Our numerical study shows that the computing time can be reduced to 21 percent of that in the case of the exact solution (the original modal solution). These savings prove that the method presented is

very efficient in various points.

In this example, we have treated the terminal cost problem in order to compare with the results obtained by Iwazumi and Koga[I.73]. The method can be extended to the regulator problem, in which the rapid change of the feedback coefficients in the neighbourhood of the terminal time is relaxed more than this example. But as mentioned in Section 3.2, the existence of the boundary layer and the nonlinearity of the Riccati equation of each mode also cause the rapid change which leads computational difficulties. Of course, the singular perturbation theory is effective in the regulator problem.

It is to be noted that Iwazumi and Koga[I.73] and Asatani, Iwazumi and Hattori[A.73.1] used the Kaplan mode. The results of [A.73.1] are incorrect in that the modal expansion via the Kaplan mode is not of the singular perturbation type, and the method presented is not applicable. The modal expansion equation via the Kaplan mode approaches asymptotically to that via the Helmholtz mode.





After the completion of the theses, it has been found that the fundamental mode can also be solved approximately by utilozing the singular pertrubation theory, since the parameters adopted make the boundary layer system asymptotically stable even under the criticality condition. But in the case of reactor kinetic equation of one-group diffusion without delayed neutrons, the description of the text is valid.

5.3 Additional remarks

5.3.1 Spatially concentrated control

In the case of the spatially concentrated control the representation (5.2.12) by the Dirac measure gives the modal expansion form of the original system coupled with each other modein place of Eqs.(5.2. 14) and (5.2.15), as



or in the matrix notation

where

 $I_{\varepsilon} \frac{d}{dt} \Phi_{i} = A_{i} \Phi_{i} + B \sum_{r=0}^{M} \sin(\frac{i\pi}{\ell} \omega_{r}) u_{r}(t); i = 1, 2, \dots, \infty,$ $I_{\varepsilon} = \begin{bmatrix} \varepsilon & 0 \\ 0 & 1 \end{bmatrix}.$

f we truncate at the N-th finite mode, the matrix T becomes

$$T = \begin{bmatrix} \sin(\pi/\ell)\omega_1, \dots, \sin(\pi/\ell)\omega_M \\ \vdots \\ \sin(N\pi/\ell)\omega_1, \dots, \sin(N\pi/\ell)\omega_M \end{bmatrix}.$$
 (5.3.2)

The matrix T violates the modal decoupling in the case of spatially distributed control, hence the individual mode cannot be treated separately. The transformation

$$v(t) = T \cdot u(t),$$
 (5.3.3)

gives the decoupled modal control formally, where v(t) is a new control vector [1.73]. The controllability in the modal space requires that

$$N \ge M$$
 (5.3.4)

where N is the number of the modes considered, and M is the dimension of the original control rods. The existence of the original M-dimensional control vector u(t) is ascertained through Eq.(5.3.3) by the following requirement :

$$rank(T) = M,$$
 (5.3.5)

which imposes a restriction upon the positioning of the control rods. The decoupled modal control can be carried out if the following performance index is given:

$$J = \int_{0}^{\ell} [\Phi(\omega, T) - \Phi_{d}(\omega)]' Q[\Phi(\omega, T) - \Phi_{d}(\omega)] d\omega + \int_{0}^{T} \int_{0}^{\ell} r[v(\omega, t) - v_{0}(\omega)]^{2} d\omega dt, \qquad (5.3.6)$$

where $v(\omega,t)$ is a distributed new control vector determined from the vector v(t) of Eq.(5.3.3). Then the technique of the preceding section is applicable to the spatially concentrated control case.

5.3.2 Multi-group diffusion equation

The results obtained in Section 5.2 can be extended to the more general systems than that described by Eqs.(5.2.1) and (5.2.2), such as multi-group diffusion equations, which approximate better the accurate Boltzmann equation describing the balance of the neutrons in the reactor.

The form of the equation for one fast and one slow groups of neutrons and one group of delayed neutrons is given as follows:

$$\frac{1}{v_{f}} \frac{\partial}{\partial t} \psi_{F} = [\nabla D_{F} \nabla - (\Sigma_{aF} + \Sigma_{R}) + (1 - \beta) \nabla \Sigma_{fF}] \psi_{F} + (1 - \beta) \nabla \Sigma_{fS} \psi_{S} + \lambda_{c} C,$$

$$\frac{1}{v_{s}} \frac{\partial}{\partial t} \psi_{s} = \Sigma_{R} \psi_{F} + (\nabla D_{s} \nabla - \Sigma_{as}) \psi_{s},$$

$$\frac{\partial}{\partial t} C = \beta \nabla \Sigma_{fF} \psi_{F} + \beta \nabla \Sigma_{fs} \psi_{s} - \lambda_{c} C.$$
(5.3.7)

In this case the criticality condition can be sustained with consideration to both the fast group and slow group neutrons. This situation should affect the result of the criticality condition in the case of one group diffusion equation (5.2.1), derived in Section 5.2.3.

The fundamental modes of the fast and slow neutrons cannot sustain the steady state without the other, which ascertains that the eigenvalue corresponding to the fundamental mode should be negative. Hence the asymptotic stability of the boundary layer system belonging to Eq. (5.3.7) is established. We conclude that the singular perturbation theory can be satisfactorily applied to the multi-group diffusion equation with multi-group delayed neutrons, in both the power control systems and the spatial control systems with reduction of the dimensionalities.

Chapter 6 Conclusions

Near-optimum control of the singularly perturbed systems has been investigated. The results obtained are summarized in this chapter. Additional remarks involving suggestions to the future works remaining to be done, are also made.

Chapter 1 is concerned with the singular perturbation theory with the view point of historical survey. Further, a supplementary relation between the Vasil'eva's method(the method of inner and outer expansions) and the method of matched asymptotic expansions has been shown.

Chapter 2 deals with the near-optimum synthesis of controls arising in lumped parameter systems. The results obtained in each section are similar to each other in a formal sense. Indeed, the results of Section 2.2 and those of Section 2.4 are related closely in that, the results of Section 2.4(two small parameters case) is a generalization of those of Section 2.2(single parameter case). The extension to more parameters case can be made easily in the same way as in Section 2.4.

The situation in Section 2.3 is rather different from that in Sections 2.2 and 2.4. It is to be noted that the full system expressed by Eqs.(2.3.6) and (2.3.7) can be solved consistently with the initial conditions (2.3.8) and (2.3.9) or with the final conditions (2.3.8') and (2.3.9'). The recursive system, however, can be solved not in the forward direction but only in the backward direction and vice versa, according to the properties of the boundary layer systems (2.3.28) - (2.3.30).

Without loss of generality, we can fix the end-value to be zero, and then we have the advantage that we need not solve the equations for g_i 's of the backward system because these have only trivial solutions.

Recently, O'Malley and others(e.g. [0.72.1.]) developed another

technique to treat the singularly perturbed linear regulator problem from a differnt point of view via Turrittin's works [Tu.52] and Harris's [Hr.60, Hr.62] on two-point boundary value problems of the singular perturbation type. Their method may be applied to the fixed-terminal optimization problem and allows a more general property of the boundary layer system, viz. that its matrix of canonical system

$$G = \begin{bmatrix} A_4 & -B_2 R^{-1} B_2' \\ -Q_3 & -A_4' \end{bmatrix}$$

should have m-eigenvalues with negative real part and m-eigenvalues with positive real part, which includes obviously our condition C5 of Section 2.3 as a special case. However, the generated recursive system must be solved under two-point boundary value conditions, and since the boundary layer occurs at both ends of the interval considered, it must treat both of them in order to make the "boundary layer corrections".

The results obtained in Section 2.3, fixed terminal optimization problems, can be extended to the several parameters case in the same manner as made in Section 2.4.

Chapter 3 is concerned with the application of the results derived in the preceding chapter to nuclear reactor control. The numerical examples show that the approximation by the singular perturbation theory is uniformly satisfactory in the interval considered. We restrict ourselves to the problem of the state regulation.

The method presented can be applied to the more general control problems such as power level change with minimum control energy, regulator problem of reactors involving temperature feedback, etc. (Weaver [Wv.68]).

Chapter 4 deals with near-optimum control of systems described by a set of partial differential equations of parabolic type. To the author's knowledge, the theory of singular perturbations of the systems described by evolution equations has not been developed sufficiently.

In Section 4.2 the singular perturbation theory for the restricted class of evolution equation is considered in respect of the regular degeneration and the construction of asymptotic expansions. The conditions assumed here are rather more rigid than necessary. But from the practical point of view, the derived results offer us with powerful tools to attack a broad class of realistic distributed systems. Further extensions to the more general systems can be made in line with the method of Section 4.2.4.

In Section 4.3, asymptotic expansions of controls by extending the technique for lumped parameter systems was obtained. The advantage of this technique lies in that the result obtained is directly construc tive. As suggested in Section 4.2.4, the results derived in Section 4.3 can be generalized to the more general systems. Lions [L.73] studied also the regular degeneration of control systems of singular perturbation type with construction of the boundary layer corrections only of the 0-th order. He also studied the Riccati equation of singular perturbation type, but his result is not consistent with the author's, and there remains some doubt in the partitioning of the Riccati equation. While the author adopted Eq.(4.3.33), Lions's partitioning is as follows:

(Eq.(3.36) P563 of Lions [L.73])

$$P(\omega, \omega', t) = \begin{bmatrix} P_{11}(\omega, \omega', t) & P_{12}(\omega, \omega', t) \\ P_{21}(\omega, \omega', t) & P_{22}(\omega, \omega', t) \end{bmatrix}$$

which is not logical seeing that the reduction of dimensionality of the original state equation causes the reduction of that of the Riccati equation generated.

The theory presented is seen to be applicable to the problems considered in Chapter 2, such as regulator problems, fixed-end-point problems, and optimization problems of systems involving several parameters.

Chapter 5 is concerned with the regulator problems with terminal

cost of the reactor with distributed parameters. The state equations (5.2.1) and (5.2.2) are not the same as treated in Chapter 4. Equation (5.2.2) does not involve any spatially differential operator, but this situation does not violate the applicability of the results obtained in Chapter 4, looking at it from the point of functional analysis (cf. Riesz-Fischer's Theorem, Theorem 4-2). Some numerical examples show the efficiency of the theory presented. The error estimation (Theorem 4-6) should be made with the consideration of the truncation is estimated by Phillipson [Ph.71], or referred by Green [Gr.53].

Throughout this thesis, the case where the terminal time is fixed is treated, but the extension to the free-terminal time problem can be made in line with Kao and Bankoff [Kao. 73]. The system with small delay can be treated by the singular perturbation theory, constructing asymptotic expansions. Other interesting subjects are differential games of singular perturbation type, filtering theory of singularly pertubed system, etc. The theory is very widely applicable since the real physical system is often described by stiff differential equations, if the system is large in dimensionality.

It should be noted that the singular perturbation theory can be adopted in modeling, in order to select state variables essential to the motion considered, and this may offer us tools to avoid the knotty problem of exceeding dimensionality.

We add that the interesting field remains to be studied in respect of the deduction of the "canonical system of singular perturbation type" from the system of arbitrarily given form.
Acknowledgments

The author wishes to express his sincere gratitude to Professor Yoshio Hattori for his constant guidance and encouragement during the course of this research. He also wishes to express his thanks to Professor Tetsuo Iwazumi, Nagoya Institute of Technology, formerly an associate professor of Institute of Atomic Energy, Kyoto University, for his valuable discussions. Especially the content of Chapter 5 was first motivated by his work.

The author acknowledges the assistances given by Mr. R. Jayakumar, Scientific Officer, Bhabha Atomic Research Center, India, Messrs. R. Koga, Institute of Atomic Energy, Kyoto University, and H. Kokame, Department of Electrical Engineering, Kyoto University, in writing this thesis. He also expresses his gratitude to Mr. M. Shiotani for his contribution to the part of numerical calculations, and to Misses Y. Asatani, K. Suzuki and K. Tsuchida for their helps in typewriting this thesis. Appendix A Theorems Related to Matrix Riccati Equations

The fundamental theorems due to Reid [R.63, R.65] are offered here.

The linear vector differential systems to be treated are of the form

$$\frac{d}{dt} u = Au + Bv, \quad \frac{d}{dt} v = Cu + Dv, \quad (A.1)$$

where u(t) and v(t) are n-dimensional vector functions, and A, B, C, and D are $n_{x}n$ matrices with time-invariant elements which are Lebesgue integrable on the interval considered. The theorems due to Reid are stated for the corresponding matrix differential equations

$$\frac{d}{dt} U = AU + BV, \quad \frac{d}{dt} V = CU + DV, \quad (A.2)$$

where in general U(t) and V(t) are matrices of $r \times n$, and here we consider the case where they are $n \times n$.

We adopt the following matrix notations in the sequel; the symbols E and O are used for the identity and zero matrices respectively of any dimensions; the conjugate transpose of a matrix M is denoted by M*. The notation $M \ge N$ or $N \le M$, (M > N or N < M), is used to signify that M and N are Hermitian matrices of the same dimensions and M - N is non-negative, (positive), definite.

By definition, a system (A.1) is identically normal if the only solution (u(t), v(t)) of this system with $u(t) \equiv 0$ on a nondegenerate interval is the identically vanishing solution $u(t) \equiv 0$, $v(t) \equiv 0$.

<u>Theorem A-1</u> ([R.63]) A system (A.2), identically normal and nonoscillatory on]- ∞ , ∞ [, has a principal solution at ∞ (at - ∞) if and only if the solution (U₀(t), V₀(t)) of (A.2) for which U₀(0) = 0, V₀(0) = E is such that W₀(t) = V₀(t)U₀(t)⁻¹ converges to a limit W_{∞} (W_{- ∞}) as t $\rightarrow -\infty$ (t $\rightarrow \infty$); the corresponding distinguished solution of the associated Riccati equation

$$-\frac{d}{dt}W = WA + DW + WBW - C, \qquad (A.3)$$
$$= -\infty) \text{ is } W_{\infty}(t) = W_{\infty}(W_{\infty}(t) = W_{\infty}).$$

<u>Theorem A-2</u> ([R.63]) A system (A.2) with coefficient matrices satisfying $A^* = D$, $C^* = C$, $B^* = B \ge 0$, and which is identically normal, is non-oscillatory on $]-\infty,\infty[$ if and only if there exists an Hermitian constant matrix W satisfying the algebraic matrix equation

at t = ∞ (t

$$WA + A*W + WBW - C = 0;$$
 (A.4)

moreover, if such a system is non-oscillatory on $]-\infty,\infty[$ then there exist Hermitian matrices W_{∞} and $W_{-\infty}$ which are individually solutions of Eq.(A.4), and are extreme solutions for Eq.(A.3) in the sense that if W(t) is any Hermitian solution of Eq.(A.3) on $]-\infty,\infty[$ then $W_{\infty} \leq W(t)$ $\leq W_{-\infty}$; in particular, if W is any Hermitian solution of Eq.(A.4) then $W_{\infty} \leq W \leq W_{-\infty}$.

<u>Theorem A-3</u> ([R.65]) If A and B are constant n×n matrices with B Hermitian and $B \ge 0$, while the n×n² matrix

$$[B, AB, A^{2}B, \ldots, A^{n-1}B]$$
(A.5)

has rank n, then there exist Hermitian matrices $W_{\infty} \leq 0$ and $W_{\infty} \geq 0$ that are extreme solutions of the matrix equation

$$WA + A*W + WBW = 0 \tag{A.6}$$

in the sense that $W = W_{\infty}$ and $W = W_{\infty}$ are individually solutions of Eq.(A.6), while if W is any Hermitian matrix satisfying Eq.(A.6) then $W_{\infty} \leq W \leq W_{\infty}$. Moreover, $W_{\infty} > 0$, $(W_{\infty} < 0)$, if and only if all eigenvalues λ of A, (-A), have $\operatorname{Re}(\lambda) < 0$.

Theorem A-3 are related to the system (A.1) through the following lemma.

Lemma A-1 ([R.65]) The system (A.1) with constant coefficient matrices is identically normal if and only if the matrix (A.5) is of rank n.

Appendix B A Matrix Inversion Formula

The well-known inversion formula for a partitioned matrix leads

$$\begin{bmatrix} M_1 & M_2 \\ M_2 & \varepsilon^{-1}M_3 \end{bmatrix}^{-1} \begin{bmatrix} A & B \\ - \\ C & D \end{bmatrix}$$
(B.1)

where

to

$$A = [M_{1} - \varepsilon M_{2}M_{3}^{-1}M_{2}']^{-1} = M_{1}^{-1} + \varepsilon M_{1}^{-2}M_{2}M_{3}^{-1}M_{2}' + O(\varepsilon^{2}),$$

$$B = - [M_{1} - \varepsilon M_{2}M_{3}^{-1}M_{2}']^{-1}[\varepsilon M_{2}M_{3}^{-1}] = -\varepsilon M_{1}^{-1}M_{2}M_{3}^{-1} + O(\varepsilon^{2}),$$

$$C = -\varepsilon M_{3}^{-1}M_{2}'[M_{1} - \varepsilon M_{2}M_{3}^{-1}M_{2}']^{-1} = -\varepsilon M_{3}^{-1}M_{2}'M_{1}^{-1} + O(\varepsilon^{2}),$$

$$D = [\varepsilon^{-1}M_{3} - M_{2}'M_{1}^{-1}M_{2}]^{-1} = \varepsilon M_{3}^{-1} + O(\varepsilon^{2}).$$

Then we have

$$M^{-1} = \begin{bmatrix} M_1^{-1} + \varepsilon M_1^{-2} M_2 M_3^{-1} M_2' - \varepsilon M_1^{-1} M_2 M_3^{-1} \\ - \varepsilon M_3^{-1} M_2' M_1^{-1} & \varepsilon M_3^{-1} \end{bmatrix} + O(\varepsilon^2). \quad (B.2)$$

Substituting Taylor expansions for M_i 's (see Section 2.3.2) into Eq.(B.2), we derive Eq.(2.3.29).

Appendix C The Detailed Expression of Eq.(2.4.7)

$$\frac{d}{dt} K_{11} = -K_{11}A_{11} - A_{11}K_{11} - K_{12}A_{21} - A_{21}K_{12} - K_{13}A_{31} - A_{31}K_{13}$$

$$+ K_{11}E_{11}K_{11} + K_{12}E_{12}K_{11} + K_{13}E_{13}K_{11} + K_{11}E_{12}K_{12}'$$

$$+ K_{12}E_{22}K_{12}' + K_{13}E_{23}K_{12}' + K_{11}E_{13}K_{13}' + K_{12}E_{23}K_{13}'$$

$$+ K_{13}E_{33}K_{13}' - Q_{1},$$

$$\mu_{1} \frac{d}{dt} \kappa_{12} = -\kappa_{11} \kappa_{12} - \kappa_{12} \kappa_{22} - \kappa_{13} \kappa_{32} - \kappa_{21} \kappa_{22} - \kappa_{31} \kappa_{23}$$

$$+ \kappa_{11} \kappa_{12} \kappa_{22} + \kappa_{12} \kappa_{22} \kappa_{22} + \kappa_{13} \kappa_{23} \kappa_{22} + \kappa_{11} \kappa_{13} \kappa_{23}$$

$$+ \kappa_{12} \kappa_{23} \kappa_{23} + \kappa_{13} \kappa_{33} \kappa_{23} + \mu_{1} (\kappa_{11} \kappa_{12} + \kappa_{11} \kappa_{11} \kappa_{12} + \kappa_{11} \kappa_{12} \kappa_{12} + \kappa_{12} \kappa_{12} \kappa_{12} + \kappa_{13} \kappa_{13} \kappa_{12}),$$

$$\mu_{1} \frac{d}{dt} \kappa_{22} = -\kappa_{22} \kappa_{22} - \kappa_{22} \kappa_{22} - \kappa_{23} \kappa_{23} - \kappa_{32} \kappa_{32} + \kappa_{22} \kappa_{22} \kappa_{22} + \kappa_{23} \kappa_{23} \kappa_{23} - \kappa_{32} \kappa_{32} + \kappa_{22} \kappa_{22} \kappa_{22} + \kappa_{23} \kappa_{23} \kappa_{23} \kappa_{23} - \kappa_{23} \kappa_{23} - \kappa_{23} \kappa_{23} - \kappa_{23} \kappa_{23} \kappa_{23} \kappa_{23} \kappa_{23} - \kappa_{23} \kappa$$

$$\mu_{1}\mu_{2} \frac{d}{dt} \kappa_{23} = -\kappa_{22}A_{23} - \kappa_{23}A_{33} - A_{32}K_{33} + \kappa_{22}E_{23}K_{33} + \kappa_{23}E_{33}K_{33} + \mu_{1}(-\kappa_{12}A_{13} + \kappa_{12}E_{13}K_{33}) + \mu_{2}(-A_{22}K_{23} + \kappa_{22}E_{22}K_{23} + \kappa_{23}E_{23}K_{23}) + \mu_{1}\mu_{2}(-A_{12}K_{13} + \kappa_{22}E_{12}K_{13} + \kappa_{23}E_{13}K_{13} + \kappa_{12}E_{12}K_{23}) + \mu_{1}^{2}\mu_{2}K_{12}E_{11}K_{13},$$

 $\mu_1 \mu_2 \frac{d}{dt} \kappa_{33} = -\kappa_{33} \Lambda_{33} - \Lambda_{33} \kappa_{33} + \kappa_{33} \kappa_{33} - \rho_3 + \mu_2 (-\kappa_{23} \Lambda_{23})$

 $- A_{23}K_{23} + K_{23}E_{23}K_{33} + K_{33}E_{23}K_{23}) + \mu_{1}\mu_{2}(-K_{13}A_{13}) \\ - A_{13}K_{13} + K_{33}E_{13}K_{13} + K_{13}E_{13}K_{33}) + \mu_{2}^{2}(K_{23}E_{22}K_{23}) \\ + \mu_{1}\mu_{2}^{2}(K_{23}E_{12}K_{13} + K_{13}E_{12}K_{23}) + \mu_{1}^{2}\mu_{2}^{2}K_{13}E_{11}K_{13}.$

Appendix D Time-Invariant Regulator Problem [Y.73]

Consider the regulator problem when $t_f\to\infty$ and all the system and weighting matrices are time-invariant. The solution of this problem is

$$u = -R^{-1}[(B_1H_1 + B_2'H_2')x_1 + (\epsilon B_1'H_2 + B_2'H_3)x_2].$$
(D.1)

The matrix

$$H = \begin{bmatrix} H_1 & \varepsilon H_2 \\ \varepsilon H_2' & \varepsilon H_3 \end{bmatrix}$$

is the positive definite root of the following algebraic equation (cf. Eq.(2.2.18):

$$0 = -K_{1}A_{1} - A_{1}K_{1} - K_{2}A_{3} - A_{3}K_{2} + K_{1}E_{1}K_{1} + K_{2}E_{2}K_{1} + K_{1}E_{2}K_{2}'$$

$$+ K_{2}E_{3}K_{2}' - C_{1}Q_{1}C_{1},$$

$$0 = -K_{1}A_{2} - K_{2}A_{4} - A_{3}K_{3} + K_{1}E_{2}K_{3} + K_{2}E_{3}K_{3} - C_{1}Q_{2}C_{2}$$

$$- \epsilon(A_{1}K_{2} - K_{1}E_{1}K_{2} - K_{2}E_{2}K_{2}),$$

$$0 = -K_{3}A_{4} - A_{4}K_{3} + K_{3}E_{3}K_{3} - C_{2}Q_{3}C_{2} - \epsilon(K_{2}A_{2} + A_{2}K_{2})$$

$$- K_{3}E_{2}K_{2} + K_{2}E_{2}K_{3} + \epsilon^{2}K_{2}E_{1}K_{2}.$$
(D.2)

This algebraic equation can be written compactly in the vector form as

$$F(H;\varepsilon) = 0. \tag{D.3}$$

It defines the vector H as an implicit function of ε . If the Jacobian matrix $\partial F/\partial H$ at $\varepsilon = 0$ is nonsingular, the differentiability of the coefficient matrices in Eq.(D.2) is concluded. It was shown that the nonsingularity of $\partial F/\partial H$ at $\varepsilon = 0$ is guaranteed by the controllability of \hat{A} , \hat{B} and the observability of \hat{A} , \hat{C} , where

$$S_{1} = [E_{2}K_{3} - A_{2}][A_{4} - E_{3}K_{3}]^{-1},$$

$$S_{2} = [A_{3}'K_{3} + Q_{2}][A_{4} - E_{3}K_{3}]^{-1},$$

$$\hat{A} = A_{1} + S_{1}A_{3} + E_{2}S_{2}' + S_{1}E_{1}S_{2},$$

$$\hat{B} = B_{1} + S_{1}B_{2},$$

$$\hat{C} = a \text{ solution of } \hat{C}'\hat{C} = \hat{Q},$$

$$\hat{Q} = -S_{2}A_{3} - A_{3}'S_{2}' - S_{2}E_{3}S_{2}' + Q_{1},$$

(cf. Section 2.3.4). Hence the following theorem.

<u>Theorem D-1</u> In addition to letting conditions C1 - C5 and C7 of Section 2.2.3 be satisfied for the time-invariant system considered, assume the following:

CD1. rank[
$$\hat{C}', \hat{A}'\hat{C}', \dots, (\hat{A}')^{n-1}\hat{C}'$$
] = n, (D.4)

CD2.
$$rank[\hat{B}, \hat{A}\hat{B}, ..., (\hat{A})^{n-1}\hat{B}] = n.$$
 (D.5)

Then there exists an $\varepsilon_0 > 0$ such that, for $\varepsilon \in [0, \varepsilon_0]$,

$$H_{i}(\varepsilon) = \sum_{j=0}^{n} H_{i}^{j}(0) (j!)^{-1} \varepsilon^{j} + H_{i}^{n+1}(\theta \varepsilon) (n+1)^{-1} \varepsilon^{n+1}, i=1-3, (D.6)$$

where $0 \leq \theta \leq 1$. The coefficients H_1^{0} are the solutions of Eq.(D.2) and H_3^{0} is the positive definite root of Eq.(D.2c) at $\varepsilon = 0$. Higher order coefficients satisfy the linear equations resulting from the successive differentiation of Eq.(D.2) with respect to ε .

References

- [A.70] K. Asatani. T. Iwazumi and Y. Hattori: Error estimations of the prompt jump approximation by singular perturbation theory, Preprint of 1970 Fall Meeting on Reactor Physics and Reactor Engineering AESJ., pp. 181 - 182 (in Japanese).
- [A.71.1] K. Asatani: A study on near-optimum control of reactor systems by singular perturbation theory, Master thesis, Dep. Electical Eng., Kyoto Univ.
- [A.71.2] K. Asatani, T. Iwazumi and Y. Hattori: Error estimation of prompt jump approximation by singular perturbation theory, J. Nucl. Sci. Technol. vol. 8, pp. 653 - 656.
- [A.71.3] K. Asatani, Y. Hattori and T. Iwazumi: Applications of singular perturbation theory to suboptimal control of point reactor kinetics, Proc. 1971 Ann. Meeting AESJ.
- [A.71.4] K. Asatani: Suboptimal control of tracking problem for nuclear reactors, Preprint of 1971 Fall Meeting on Reactor Physics and Reactor Eng. AESJ., pp. 230 - 231.
- [A.72.1] K. Asatani: Suboptimal control of fixed-end-point, minimum energy problems, 15-th Joint Convention of JAACE and others, No. 1009.
- [A.72.2] K. Asatani and Y. Hattori: A note on the initial conditions of singular perturbations, Bull. IAE Kyoto Univ. vol. 42. p. 19, 1972.
- [A.73.1] K. Asatani, T. Iwazumi and Y. Hattori: Suboptimal control of reactor involving distributed parameters, Preprint of 1973 Fall Meeting on Reactor Physics and Reactor Eng. AESJ., pp. 204 - 205.
- [A.73.2] K. Asatani and T. Iwazumi: Singular perturbations of systems involving two small parameters, Bull. IAE Kyoto Univ. vol. 44, p. 38, 1973.
- [A.73.3] K. Asatani, T. Iwazumi and Y. Hattori: Suboptimal control

of systems involving two small parameters, Bull. IAE Kyoto Univ. vol. 44, p. 37, 1973.

- [A.74.1] K. Asatani: Suboptimal control of fixed-endpoint minimum energy problem via singular perturbation theory, J. Math. Anal. Appl. vol. 45, pp. 684-697.
- [A.74.2] K. Asatani, T. Iwazumi and Y. Hattori: Singular perturbations of abstract evolution equations, Bull. IAE., Kyoto Univ. vol. 45, p.13, 1974.
- [Ac.71] Z. Ackasu, S. Gerald and S. Lellouche: Mathematical Mehtods in Nuclear Reactor Dynamics, Academic Press, N.Y.
- [Am.69] F. Amano: Approximate solution of one-point reactor kinetic equations for arbitrary reactivities, J. Nucl. Sci. Technol., vol. 6, pp. 646 - 656.
- [Am.71] F. Amano: An error analysis of the prompt jump approximation, ibid., vol. 8, pp. 184 - 191.
- [At.66] M. Athan and P. Falb: Optimal Control, McGraw-Hill, N.Y.
- [Ax.69] E. I. Axelband: Optimal Control of linear distributed parameter systems, Advances in Control Systems. vol. 7, Academic Press, N. Y.
- [B.57] R. Bellman: Dynamic Programming, Princeton Univ. Press, Princeton, N.J.
- [B.65] R. Bellman and R. Kalaba: Quasilinearization and Non-Linear Boundary Value Problems, American Elsevier, N.Y.
- [B1.65] P. B. Bail^ey and G. M. Wing: Some recent developments in invariant imbedding with applications, J. Math. and Phys. vol. 6, pp. 453 - 462.
- [Br.68] W. L. Brogan: Optimal control theory applied to systems described by partial differential equations, Advances in Control Systems, vol. 6, Academic Press, N.Y.
- [Bt.69] A. G. But kovskii: Distributed Control Systems, American Elsevier, N.Y.
- [C.67] J. Canosa: A new method for nonlinear reactor dynamics problems, Nucleonik, vol. 9, pp. 289 - 295.

- [C1.68] K.W. Chang: Almost periodic solutions of singularly perturbed systems of differential equations, J. Diff. Eqs., vol. 4, pp. 300 - 307.
- [Cn.69.1] K. W. Chang and W. A. Coppel: Singular perturbations of initial value problems over finite intervals, Arch. Ratl. Mech. Anal., vol. 32, pp. 268 - 280.
- [Cn.69.2] K.W. Chang: Remarks on certain hypotheses in singular perturbations, Proc. Amer. Math. Soc., vol. 23, pp. 41 - 45.
- [Co.67] W. A. Coppel: Dichotomies and reducibility, J. Diff. Eqs., vol. 3, pp. 500 - 521.
- [D.60] J. Dieudonne: Foundations of Modern Analysis, Academic Press, N.Y.
- [Df.58] N. Dunford and J. T. Scwartz: Linear Operators, Part I, Interscience, N. Y.
- [Dm.69] N. S. Demuth and W. L. Hendry: Numerical studies of an initial-value neutron transport problem, Trans. Amer. Nucl. Soc., vol. 12, pp. 729 - 730.
- [F.55] K. O. Friedrichs: Asymptotic phenomena in mathematical physics, Bull. Amer. Math. Soc., vol. 61, pp. 485 - 504.
- [Fk.69] L. E. Frankel: On the method of the matched asymptotic expansions, Part I: A matching priciple, Proc. Camb. Phil. Soc., vol. 65, pp. 209 - 231.
- [F1.55] L. Flatto and N. Levinson: Periodic solutions of singularly perturbed systems, J. Ratl. Mech. Anal., vol. 4, pp. 943 -950.
- [G1.63] I. M. Gelfand and S. V. Fomin: Calculus of Variations, Prentice-Hall, Englewood Cliffs, N. Y.
- [Gr.53] J. W. Green: An expansion method for parabolic partial differential equations, J. Res. Natl. Bur. Standards, vol. 51, pp. 127 - 132.
- [Gs.69] R. Goldstein and L. M. Shotkin: Use of the prompt jump approximation in fast-reactor kinetics, Nucl. Sci. Eng., vol. 35, pp. 94 - 103.

- [Ha.71] A. H. Haddad and P. V. Kokotović: On a singular perturbation problem in linear filtering theory, Proc. 5-th Princeton Conf. Inform. Sci. Sys., Princeton, NJ.
- [Hd.70.1] C. Hadlock and P. V. Kokotović: Near optimum design of three time scale systems, Proc. 4-th Princeton Conf. Inform. Sci. Sys., Princeton, NJ.
- [Hd.70.2] C. Hadlock: On a class of singularly perturbed two point boundary value problems, Proc. 8-th Allerton Conf. on Circuit and System Theory, pp. 331 - 339.
- [Hn.69] W. L. Hendry and G. I. Bell: An analysis of the timedependent neutron transport equation with delayed neutrons by the method of mathced asymptic expansions, Nucl. Sci. Eng., vol. 35, pp. 240 - 248.
- [Hn.70] W. L. Hendry: Application of the method of matched asymptotic expansions to a problem in linear transport theory, J. Mathematical Physics, vol. 11, pp. 1743 - 1749.
- [Hn.71] W. L. Hendry: Solution to the linear time-dependent neutron transport equation with time dependent source and cross sections, Nucl. Sci. Eng., vol. 45, pp. 1 - 6.
- [Hp.66] F. C. Hoppensteadt: Singular perturbations on the infinite interval, Trans. Amer. Math. Soc. vol. 123, pp. 521 - 535.
- [Hp.67] F. C. Hoppensteadt: Stability in systems with parameters, J. Math. Anal. Appl., vol. 18, pp. 129 - 134.
- [Hp.68] F. C. Hoppensteadt: Asymptotic stability in singular perturbation problems, J.Diff. Eqs., vol. 4, pp. 350 - 358.
- [Hp.69.1] F. C. Hoppensteadt: Asymptotic series solutions of some nonlinear parabolic equations with a small parameter, Arch. Ratl. Mech, Anal., vol. 35, pp. 284 - 298.
- [Hp.69.2] F. C. Hoppensteadt: Cauchy problems involving a small parameter, Bull. AMS, vol. 76, pp. 142 - 146.
- [Hp.69.3] F. C. Hoppensteadt: On systems of ordinary differential equations with several parameters multiplying the derivatives, J. Diff. Eqs. vol. 5, pp. 106 - 116.

- [Hp.70] F. C. Hoppensteadt: A geometric approach to boundary value problems for nonlinear ordinary differential equations with a small parameter: Proc. Conf. on Analytic Theory of Ordinary Differential Equations, Western Mich. Univ.
- [Hp.71.1] F. C. Hoppensteadt: Properties of solutions of ordinary differential equations with small parameter, Commun. Pure Appl. Math., vol. 24, pp. 807 - 840.
- [Hp.71.2] F. C. Hoppensteadt: On quasilinear parabolic equations with a small parameter, ibid., vol. 24, pp. 17 - 38.
- [Hr.60] W. A. Harris, Jr.: Singular perturbations of two-point boundary problems for systems of ordinary differential equations, Arch. for Ratl. Mech. Math., vol. 5, pp. 212 - 225.
- [Hr.62] W. A. Harris, Jr.: Singular perturbations of two point boundary problems, J. Math. Mech., vol. 11, pp. 371 - 382.
- [Hy.64] C. Hayashi: Nonlinear Oscillations in Physical Systems, McGraw-Hill, N. Y.
- [I.73] T. Iwazumi and R. Koga: Optimal feedback control of a nuclear reactor as a distributed parameter system, J. Nucl. Sci. Technol. vol. 10, pp. 674 - 684.
- [K.60] R. E. Kalman: Contribution to the theory of optimal control, Bol. Soc. Mat. Mex., vol. 5, pp. 102 - 119.
- [Ka.61] S. Kaplan: The property of finality and the analysis of problems in reactor space-time kinetics by various modal expansions, Nucl. Sci. Eng., vol. 9, pp. 357 - 361.
- [Kao.73] Y. K. Kao and S. G. Bankoff: Singular perturbation analysis of free-time optimal control problems, 1973 Joint Automatic Cnt. Conf. of Amer. Automatic Cnt. Council, Ohio Univ., Columbia, Ohio, pp. 176 - 183.
- [Kd.69] Y. Kuroda and A. Makino: Optimal control of a nuclear reactor system with distributed parameters (I), Proc. Faculty Eng., Tokai Univ., No. 1, p. 13, (in Japanese).
- [K1.68] H. B. Keller: Numerical Methods for Two Point Boundary Value Problems, Blaisdell, Boston, MA.

- [Km.71] A. Kolmogolov and S. Fomin: Elementary Theory of Functions and Functional Analysis, Iwanami, Tokyo (Japanese ed.).
- [Ko.68] P. V. Kokotović and P. Sannuti: Singular perturbation method for reducing the model order in optimal control design, IEEE Trans. Automatic Control, vol. AC-13, pp. 374 - 384.
- [Ko.72] P. V. Kokotović and R. A. Yackel: Singular perturbation of linear regulators: Basic theorems, ibid., vol. Ac-16, pp. 29 - 37.
- [Kr.67] S. G. Krein: Linear Differential Equations in a Banach Space, Yoshioka Publ. Inc., Kyoto, (Japanese ed.)
- [Ks.69] W. E. Kastenberg: Stability analysis of nonlinear space dependent reactor kinetics, Advances in Nuclear Sci. and Technol., vol. 5, Academic Press, N. Y.
- [Ku.67] S. Kaplun: Fluid Mechanics and Singular perturbations, Academic Press, N. Y.
- [L.71] J. L. Lions: Optimal Control of Systems Governed by Partial Differential Equations, Springer-Verlag.
- [L.72] J. L. Lions and E. Magenes: Nonhomogeneous Boundary Value Problems and Applications, vol. 1, Springer-Verlag.
- [L.73] J. L. Lions: Perturbations Singuliers dans les Preblems aux Limites et en Control Optimal, Lecture Notes in Math. vol. 323, Springer-Verlag.
- [Lv.54] J. J. Levin and N. Levinson: Singular perturbations of nonlinear system of differential equations and an associated boundary layer equation, J. Ratl. Mech. Anal., vol 3, pp. 247 - 270.
- [Lv.56] J. J. Levin: Singular perturbations of nonlinear systems of differential equations related to conditional stability, Duke Math, J., vol. 23, pp. 609 - 620.
- [Lv.59] J. J. Levin: The asymptotic behavior of the stable initial manifolds of a system of nonlinear differential equations, Trans. AMS., vol. 85, pp357-368.
- [M.65] S. Mizohata: Theory of Partial Differential Equations,

Iwanami, Tokyo (in Japanese).

- [Mc.67] J. W. Macki: Singular perturbations of a boundary value problem for a system of nonlinear differential equations, Arch. Ratl. Mech. Anal., vol. 24, pp. 219 - 232.
- [Mf.69] I. H. Mufti, C. K. Chow and F. T. Stock: Solution of illconditioned linear two-point boundary value problems by the Riccati trnsformation, SIAM Review, vol. 11, pp. 616 - 619.
- [Mh.70] R. R. Mohler and C. N. Shen: Optimal Control of Nuclear Reactors, Academic Press, N. Y.
- [My.73] G. H. Meyer: Initial Value Mehtods for Boundary Value Problems, Academic Press, N. Y.
- [N.64] Y. Nishikawa: A Contribution to the Theory of Nonlinear Oscillations, Nippon Print. Pub. Co., Osaka.
- [0.67] R. E. O'Malley, Jr.: Singular perturbations of boundary value problems for linear ordinary differential equitions involving two small parameters, J. Math. Anal. Appl., vol. 19, pp. 291 - 308.
- [0.68.1] R. E. O'Malley, Jr. and J. B. Keller: Loss of boundary conditions in the asymptotic solution of linear ordinary differential equations II, Boundary value problems, Commun. Pure Appl. Math., vol. 21, pp. 263 - 270.
- [0.68.2] R. E. O'Malley, Jr.: Topics in singular perturbations, Advances in Mathematics, vol. 2, pp. 365 - 470.
- [0.69] R. E. O'Malley, Jr.: Boundary value problems for linear systems of ordinary differential equations involving many parameters, J. Math. Mech., vol. 18, pp. 835 - 856.
- [0.70.1] R. E. O'Malley, Jr.: Singular perturbation of a nonlinear boundary value problem, Proc. 8-th Allerton Conf. Circuit Sys. Theory, pp. 322 - 330.
- [0.70.2] R. E. O'Malley, Jr.: Singular perturbations of a boundary value problem for a system of nonlinear differential equations, J. Diff. Eqs., vol. 8, pp. 431 - 447.
- [0.71.1] R. E. O'Malley, Jr.: Boundary layer methods for nonlinear

initial value problems, SIAM Review, vol. 13, pp. 425 - 434.

- [0.71.2] R. E. O'Malley, Jr.: On intial value problems for nonlinear systems of differential equations with two parameters, Arch. Ratl. Mech. Anal., vol 40, pp. 209 - 222.
- [0.72.1] R. E. O'Malley, Jr.: The singularly perturbed linear state regulator problem, SIAM J. Control, vol. 10, pp. 399 413.
- [0.72.2] R. E. O'Malley, Jr.: Singular perturbation of time-invariant linear state regulator problem, J. Diff. Eqs., vol. 12, pp. 117 - 128.
- [P.68] C. E. Pearson: On a differential equation of boundary layer type, J. Math. and Phys., vol. 47, pp. 134 - 154.
- [Ph.71] G. A. Phillipson: Identification of Distributed Systems, American Elsevier, N. Y.
- [Pn.62] L. S. Pontryagin et al: The Mathematical Theory of Optimal Process, Interscience, N. Y.
- [R.63] W. T. Reid: Riccati matrix differential equations and nonosillation criteria for associated linear differential systems, Pacific J. of Math. vol. 13, pp. 665 - 685.
- [R.65] W. T. Reid: A matrix equation related to a non-oscillation criterion and Liapunov stability, Quar. Appl. Math. vol. 23, pp. 83 - 87.
- [S.68] D. H. Sattinger: Stability of nonlinear parabolic systems, J. Math. Anal. Appl., vol. 24, pp. 241 - 245.
- [Sa.69.1] P. Sannuti and P. V. Kokotović: Near-optimum design of linear systems by a singular perturbation method, IEEE Trans. Automatic Control, vol. AC-14, pp. 15 - 21.
- [Sa.69.2] P. Sannuti and P. V. Kokotović: Singular perturbation method for near optimum design of higher-order nonlinear systems, Automatica, vol. 5, pp. 773 - 779.

- **Ŧ**

- [Sa.69.3] P. Sannuti: Continuity and differentiability properties of optimal control with respect to singular perturbations, IEEE., vol. AC-14, pp. 762 - 763.
- [Sa.71] P. Sannuti: Asymptotic expansions of singularly perturbed

quasilinear optimal systems, Proc. 9-th Ann.Allerton Conf. on Circuit and System theory, pp. 182 - 191.

- [Sa.73] P. Sannuti and P. Reddy: Asymptotic series solution of optimal systems with small time delay, IEEE Trans. Automatic Control, vol. AC-18, pp. 250 - 259.
- [Tk.52] A. N. Tikhonov: Systems of differential equations containing a small parameter multiplying the highest derivatives, Mat. Sb. N. S., vol. 73, pp. 575 - 585 (in Russian).
- [Tp.62.1] V. A. Tupčiev: The existence, uniqueness and asymptotic behavior of the solution of differential equations with a small parameter in the term containing the highest derivative, Soviet Math. Dokl., pp. 302 - 305.
- [Tp.62.2] V. A. Tupčiev: Asymptotic behavior of the solution of a boundary problem for systems of differential equations of first order with a small parameter in the derivative, Soviet Math. Dokl., vol. 3, pp. 612 - 615.

1

-

- [Tr.63] V. A. Trenogin: Asymptotic behavior and existence of a solution of the Cauchy problem for a first order differential equation with a small parameter in a Banach space, Soviet Math., vol. 4, pp. 1261 - 1265.
- [Tu.52] H. L. Turrittin: Asymptotic expansions of solutions of systems of ordinary linear differential equations containing a parameter, Contributions to Non linear Oscillations, vol. 2, pp. 81 - 116.
- [vd.64] M. Van Dyke: Perturbation method in Fluid Mechanics, Acadimic Press, N. Y.
- [Vi.57] M. I. Vishià and L. A. Lyusternik: Regular degeneration and boundary layer for linear differential equations with small parameter, Amer. Math. Soc. Translation ser. 2, pp. 239 - 364.
- [Vi.58] M. I. Vishik and L. A. Lyusternik: On the asymptotic behavior of the solution of the boundary problems for quasilinear differential equations, Dokl. Akad. N. S., vol.121,

pp. 778 - 781.

- [Vs.63.1] A. B. Vasil'eva: Asymptotic behaviour of solutions to certain problems involving nonlinear differential equations containing a small parameter multiplying the highest derivatives, Russian Math. Surv. vol. 18, pp. 13 - 81.
- [Vs.63.2] A. B. Vasil'eva: Asymptotic methods in the theory of ordinary differential equations containing small parameters in front of the higher derivatives, USSR Compt. and Math. Phys., vol. 3, pp. 823 - 863.
- [W.64] P. K. C. Wang: Control of distributed parameter systems, Advances in Control Systems, vol. 1, Academic Press, N. Y.
- [Wi.67] D. M. Wiberg: Optimal control of nuclear reactor systems, Advances in Control Systems, vol. 5, Academic Press, N. Y.

) ii

- [W1.73] R. R. Wilde and P. V. Kokotovic: Optimal open- and colsedloop control of singularly perturbed linear systems, IEEEL, vol. AC-18, pp. 616 - 626.
- [Wr.68] R. A. Werner and J. B. Cruz: Feedback control which preserves optimality for systems with unknown parameters, IEEE Trans. Automatic Control, vol. AC-13, pp. 621 - 629.
- [Ws.65] W. R. Wasow: Asymptotic Expansions for Ordinary Differential Equations, John Wiley and Sons, N. Y.
- [Wv.68] L. E. Weaver: Reactor Dynamics and Control, American Elsevier, N. Y.
- [Y.72] R. A. Yackel and R. R. Wilde: Boundary layer methods for the optimal control of singularly perturbed systems, J.ACC of AACC, Stanford Univ., Stanford, California, pp. 676 - 677.
- [Y.73] R. A. Yackel and P. V. Kokotović: A boundary layer method for the matrix Riccati equation, IEEE Trans. Automatic Control, vol. AC-18, pp. 17 - 24.